# Southern Illinois University Carbondale OpenSIUC

#### Dissertations

Theses and Dissertations

12-1-2018

## Labor and Social Identity in Ancient Peru: A Bioarchaeological Perspective

Sarah Katherine Muno Southern Illinois University Carbondale, smuno@pasadena.edu

Follow this and additional works at: https://opensiuc.lib.siu.edu/dissertations

#### **Recommended** Citation

Muno, Sarah Katherine, "Labor and Social Identity in Ancient Peru: A Bioarchaeological Perspective" (2018). *Dissertations*. 1608. https://opensiuc.lib.siu.edu/dissertations/1608

This Open Access Dissertation is brought to you for free and open access by the Theses and Dissertations at OpenSIUC. It has been accepted for inclusion in Dissertations by an authorized administrator of OpenSIUC. For more information, please contact opensiuc@lib.siu.edu.

## LABOR AND SOCIAL IDENTITY IN ANCIENT PERU:

## A BIOARCHAEOLOGICAL PERSPECTIVE

by

Sarah Katherine Muno

B.A., San Francisco State University, 2003 M.A., Western Michigan University 2006

A Dissertation Submitted in Partial Fulfillment of the Requirements for the Doctor of Philosophy degree

> Department of Anthropology in the Graduate School Southern Illinois University Carbondale December 2018

Copyright by Sarah Katherine Muno, 2018 All Rights Reserved

## DISSERTATION APPROVAL

## LABOR AND SOCIAL IDENTITY IN ANCIENT PERU: A BIOARCHAEOLOGICAL PERSPECTIVE

By

Sarah Katherine Muno

A Dissertation Submitted in Partial

Fulfillment of the Requirements

for the Degree of

Doctor of Philosophy

in the field of Anthropology

Approved by:

Dr. Izumi Shimada, Chair

Dr. Roberto Barrios

Dr. Susan Ford

Dr. Michael Olson

Dr. John Verano

Graduate School Southern Illinois University Carbondale September 26, 2018

## AN ABSTRACT OF THE DISSERTATION OF

Sarah K. Muno for the Doctor of Philosophy degree in Anthropology, presented on September 26, 2018 at Southern Illinois University Carbondale.

TITLE: LABOR AND SOCIAL IDENTITY IN ANCIENT PERU: A BIOARCHAEOLOGICAL PERSPECTIVE

## MAJOR PROFESSOR: Dr. Izumi Shimada

This dissertation presents a bioarchaeological study of labor and social identity in coastal Peru during the Late Intermediate Period (900 – 1470 CE), using data from contemporaneous Middle Sicán (Sicán Precinct and El Brujo, north coast) and Ychsma (Pachacamac, central coast) mortuary contexts. I combine information about funerary treatment with skeletal evidence of trauma, degenerative joint disease, and muscle attachment site morphology (enthesial changes or EC) to test whether inferred commoners were "over-worked" relative to their elite counterparts, as often assumed based on western, Marxist notions of social class.

Much of what has been inferred about socio-economic organization in coastal Peru during the Late Intermediate Period is modeled after the *parcialidades* described in early Spanish chronicles and colonial documents. In this system, occupation, social status, and ethnicity were intimately intertwined, with common fishers and farmers serving as the "productive base" for privileged members of society, including full-time artisans and their elite patrons. Archaeological evidence of elite sponsored large-scale labor projects, including specialized craft production, in pre-Hispanic coastal Peru accords well with the *parcialidad* model, but assumptions about the social identities of laborers often go untested. Human skeletal data offer a unique opportunity to redress this situation, providing information about life experience –

i

including patterns of physical activity – that are not typically accessible with other kinds of archaeological data.

Bioarchaeological studies of physical activity hold great promise for testing hypotheses about social identity and life experience in ancient societies, but they are not without some limitation. People who engage in strenuous physical activity tend to have more degenerative joint disease and enthesial changes than those who do not, but the precise mechanisms behind this are not well understood. Age and body size are known to influence these skeletal markers, although some researchers have suggested certain entheses may be less sensitive to size and thus more informative about activity, than others.

In my sample, there were no discernible differences in skeletal trauma, degenerative joint disease, or ECs between elites and non-elites, or between males and females, when statistically controlling for the influence of age and/or size. These results do not support the hypotheses that non-elites were over-burdened by arduous labor tasks or that exemption from such tasks was part of the social privileges afforded to elites. Therefore, conventional perspectives that tend to conflate elite and non-elite identities with oppressor/oppressed or manager/laborer roles appear to have little relevance for characterizing the social dynamics of labor organization in Middle Sicán and Ychsma societies.

My study supports, at least in part, previous research that argues some entheses are less prone to the influence of size than others, and may therefore be more reliable indicators of activity. In this sample, strong statistical correlations between EC scores, age, and size as determined from three humeral measurements were found for fibrous entheses, but humeral size did not correlate to scores for the fibrocartilaginous type. However, current uncertainties about the precise etiology of enthesial changes makes it difficult to interpret variation in EC scores

ii

with a high degree of certainty, and thus my study also highlights some of the drawbacks associated with using EC scores to infer patterns of activity. Experimental research to better understand how the timing, duration, and severity of muscle stress and strain influence enthesial development and technological innovations to quantify enthesial size and shape will be key to resolving these issues in the future.

## ACKNOWLEDGMENTS

First and foremost, I would like to thank Dr. Izumi Shimada for giving me the opportunity to be a part of the Sicán and Pachacamac Archaeological Projects; without his guidance and support this research would not have been possible. Special thanks are extended to Dr.'s Shimada, Ford, Corruccini, Verano, Olson, and Barrios for their encouragement and willingness to remain on my dissertation committee despite a long hiatus as family and teaching responsibilities took precedence. Their expertise and suggestions have been instrumental not only for improving this dissertation, but also for my growth as a person. At the times when I needed it most, they offered direction, fresh perspectives, and most importantly, inspiration.

There are a number of people who graciously welcomed me into their research facilities and homes while I was in Peru collecting data, for which I am extremely grateful. Dr. Carlos Elera, Juan Carlos Santoyo, and the staff at Museo Nacional Sicán provided access to the human skeletal remains excavated from the Sicán Precinct, as well as an ideal space to work in. My research at El Brujo was possible thanks to Régulo Franco, Denis Vargas, Carmen Gamarra, and César Gálvez. Daniel Jesús provided laboratory assistance at El Brujo and was ever patient with my limited command of the Spanish language. Felipe Calderón Jiménez and Jesús Esperanza García Cauricci made me feel at home and helped me a great deal during my stay in Magdalena de Caõ. Dr. Denise Pozzi-Escot provided access to the Ychsma materials and laboratory space at Pachacamac, while Patricia Landa Cragg and Paloma Carcedo housed me in Lima and were also extremely generous with their time and resources. Some of my fondest memories were made in Peru, thanks to the kindness, patience, and friendship of the people I encountered while there.

This work was completed thanks to the unwavering love and support of my family, who have gone above and beyond to help me reach my goals. My parents cultivated in me a passion

iv

for science and teaching, and because of them I gained the courage to embark on this journey in the first place. I wish my dad were here to see me complete this work, and I think he would have been proud of what I have achieved thus far. And finally, words cannot begin to express how grateful I am to my dear partner, Allan Boggess. His selfless love and commitment to both me and our son allowed me to focus on my writing, and this work is in many ways as much a product of his efforts as it is of my own.

<u>CHAPTER</u>	<u>PAGE</u>
ABSTRACT	i
ACKNOWLEDGMENTS	iv
LIST OF TABLES	vii
LIST OF FIGURES	ix
CHAPTERS	
CHAPTER 1 – Introduction	1
CHAPTER 2 – Theoretical Orientation	17
CHAPTER 3 – Literature Review	
CHAPTER 4 – Methods and Materials	61
CHAPTER 5 – Results	
CHAPTER 6 – Discussion	
REFERENCES	
APPENDICES	
APPENDIX A – Non-elite Skeletal Sample Inventory	
APPENDIX B – Elite Skeletal Sample Inventory	
APPENDIX C – Non-elite Sample Fibrous Entheses Data	
APPENDIX D – Non-elite Sample Fibrocartilaginous Entheses Data	
APPENDIX E – Elite Sample Fibrous Entheses Data	
APPENDIX F – Elite Sample Fibrocartilaginous Entheses Data	
APPENDIX G – Letter of Permission to Use Copyrighted Photograph	174
VITA	175

## TABLE OF CONTENTS

## LIST OF TABLES

TABLE	PAGE
Table 1 – Mortuary characteristics used to infer social status	65
Table 2 – Entheses evaluated by bone and type	
Table 3 – Sample size by site, status, and sex	77
Table 4 – Descriptive statistics for age by site, status, and sex	78
Table 5 – Frequencies of upper limb trauma by bone and status	
Table 6 – Frequencies of torso trauma by bone and status	
Table 7 – Frequencies of lower limb (tibia) trauma by status	90
Table 8 – Frequencies of upper limb trauma by bone and sex	90
Table 9 – Frequencies of torso trauma by bone and sex	91
Table 10 – Frequencies of lower limb (tibia) trauma by sex	91
Table 11 – Test of association between status and upper limb trauma	95
Table 12 – Test of association between status and torso trauma	95
Table 13 – Test of association between status and lower limb trauma	95
Table 14 – Test of association between sex and upper limb trauma	95
Table 15 – Test of association between sex and torso trauma	95
Table 16 – Test of association between sex and lower limb trauma	96
Table 17 – Frequencies of upper limb DJD by joint, status, and age group	96
Table 18 – Frequencies of spinal DJD by vertebral segment, status, and age group	97
Table 19 – Frequencies of lower limb (knee) DJD by status and age group	98
Table 20 – Frequencies of upper limb DJD by joint, sex, and age group	98
Table 21 – Frequencies of spinal DJD by vertebral segment, sex, and age group	99
Table 22 – Frequencies of lower limb (knee) DJD by sex and age group	

TABLE	<u>PAGE</u>
Table 23 – Test of association between upper limb DJD and status by age group	106
Table 24 – Test of association between spinal DJD and status by age group	106
Table 25 – Test of association between lower limb DJD and status by age group	106
Table 26 – Test of association between upper limb DJD and sex by age group	107
Table 27 – Test of association between spinal DJD and sex by age group	107
Table 28 –Test of association between lower limb DJD and sex by age group	107
Table 29 – Descriptive statistics for aggregate EC scores by enthesial type and site	110
Table 30 – Descriptive statistics for aggregate EC scores by enthesial type and sex	110
Table 31 – Descriptive statistics for aggregate EC scores by enthesial type and status	111
Table 32 – One-way ANOVA for status and residual fibrous and fibrocartilaginous EC scores	111
Table 33 – Correlation of aggregate fibrous EC scores and average age	112
Table 34 – Correlation of aggregate fibrous EC scores and humeral size	112
Table 35 – Correlation of aggregate fibrocartilaginous EC scores and average age	112
Table 36 – Correlation of aggregate fibrocartilaginous EC scores and humeral size	112

## LIST OF FIGURES

<u>FIGURE</u> <u>PA</u>	<u>GE</u>
Figure 1 – Map of Peru, yellow stars indicate approximate locations of the Sicán Precinct	
(top), El Brujo (center), and Pachacamac (bottom)	3
Figure 2 – Normal right clavicle (left); fractured right clavicle (right)	30
Figure 3 – Thoracic compression fracture (left); Schmorl's node (right)	30
Figure 4 – Eburnation of distal femur	30
Figure 5 – Example of non-elite burial at the Sicán Precinct; drawing by Izumi	
Shimada and César Samillán	.48
Figure 6 – Example of possible sacrificial burial at the Sicán Precinct; drawing by	
Izumi Shimada and César Samillán	49
Figure 7 – Example of an elite burial at the Sicán Precinct; drawing by Izumi Shimada	
and César Samillán	. 50
Figure 8 – "Strong" fibrous EC on humerus (left) and "strong" fibrocartilaginous EC on radius	S
(right)	73
Figure 9 – Decorated textile associated with inferred elite Ychsma Bundle N	86
Figure 10 – Inferred elite Ychsma Bundle B; photo by Izumi Shimada. Original source:	
Takigami et al. 2014. Assessing the chronology and rewrapping of funerary	
Bundles at the prehispanic religious center of Pachacamac, Peru. Latin American	
Antiquity 25(3): 322-343	86
Figure 11 – Graphic comparison of skeletal trauma frequencies by site	92
Figure 12 – Graphic comparison of skeletal trauma frequencies by sex	93
Figure 13 – Graphic comparison of skeletal trauma frequencies by status	94

FIGURE	PAGE
Figure 14 – Graphic comparison of DJD by site for < 30 years age group	
Figure 15 – Graphic comparison of DJD by site for 30+ years age group	
Figure 16 – Graphic comparison of DJD by sex for < 30 years age group	
Figure 17 – Graphic comparison of DJD by sex for 30+ years age group	
Figure 18 – Graphic comparison of DJD by status for < 30 years age group	
Figure 19 – Graphic comparison of DJD by status for 30+ years age group 90	

### CHAPTER 1

### INTRODUCTION

## Research Objectives & Hypotheses

In this study, I infer patterns of physical activity from human skeletal remains to explore how labor demands varied in relation to social status in two pre-Hispanic Peruvian societies. I offer new insights regarding how social identities were experienced as part of daily life, focusing on the Sicán (also known as Lambayeque) and Ychsma archaeological cultures that flourished on the northern and central coasts (respectively) during the Late Intermediate Period (900 – 1470 CE). This was a time characterized by significant socio-political reorganization and the rise of regional elites and polities of varied complexity and scale, just prior to Inka and then Spanish conquest. The power and authority of Sicán and Ychsma elites were thought to have been rooted in religious ideologies that naturalized social inequality, and contingent upon the ability to mobilize laborers on a large scale (Eeckhout 2008b; 2013; Klaus et al. 2017; Shimada 2014), ideal conditions for exploring the issue at hand. I also evaluate recent revisions to how patterns of activity are inferred from human skeletal remains, highlighting some of the strengths and weaknesses of these approaches.

Using contemporaneous (c. 1000 – 1400 CE) samples from the Sicán Precinct and Pachacamac, major ceremonial centers occupied by Middle Sicán and Ychsma societies (respectively), as well as Middle Sicán burials at the site of El Brujo some 300 km south of the Precinct (see Figure 1 for a map showing Peru and the approximate location of these archaeological sites), I question whether elites and non-elites differed significantly with respect to their engagement in manual labor. Was exemption from physically-demanding labor tasks part of the social privileges afforded to elite members of society? Were the bulk of subsistence activities, resource extraction and transportation, as well as construction activities being performed by non-elites? These assumptions are often implied in archaeological studies of labor based on contemporary, western notions of social class, but may have little relevance for understanding social dynamics in the ancient past, and have seldom been examined using human skeletal and mortuary data. I address these primary questions by focusing on patterns of skeletal trauma, degenerative joint disease (DJD) and enthesial changes [ECs] (i.e muscle attachment site morphology) in relation to information about social status inferred from mortuary contexts. These skeletal markers have long been associated with physical activity in the archaeological literature, offering a useful approach for my study (Derevenski 2000; Hawkey 1988; Hawkey and Merbs 1995; Klaus et al. 2009; Pearson and Buikstra 2006; Porčić and Stefanović 2009; Schrader 2014; Steen and Lane 1998; Villotte et al. 2010; Weiss 2007; Wesp 2014).

To assess whether conventional models of social class are relevant for understanding how social inequality was experienced as part of daily life in Middle Sicán and Ychsma societies, I test three related hypotheses: 1) Non-elites engaged in activities that placed them at greater risk of accidental injury compared to elites, so that evidence of skeletal trauma will be more prevalent in this group; 2) Degenerative joint disease (DJD)



Figure 1 - Map of Peru, yellow stars indicate approximate locations of the Sicán Precinct (top), El Brujo (center), and Pachacamac (bottom).

will be more common and more severe in non-elites, because they were more likely to engage in strenuous tasks on regular basis; and 3) the greatest physical demands were imposed upon nonelites as part of their labor obligations, resulting in more pronounced muscle attachments and thus higher combined EC scores relative to elites.

My secondary objective is to evaluate recent assertions regarding ECs, which have broader implications for how studies of activity should be approached in an archaeological context. Henderson and Cardoso (2013), Villotte et al. (2010), and Weiss (2015) have suggested that fibrocartilaginous entheses are less prone to systemic influences and thus more reliable as indicators of activity than fibrous entheses. I test this supposition by analyzing EC scores separately for the two enthesial types and comparing their statistical correlations with measures of age and size. Results of these analyses will contribute to current debates regarding the utility of ECs as markers of activity, and help to clarify future directions for this approach.

## Problem Context: Labor and Social Identity

Labor division and task specialization have long been considered hallmarks of social complexity and stratification (Hayden 1995; Henrich and Boyd 2008; Kerbo 2009; Service 1971). From the very inception of the social sciences and through the works of pioneers like Émile Durkheim, Karl Marx, and Max Weber, labor has emerged as one of the main focal points for discussing political power, ideology, and social relations. How labor is organized – and the social significance attributed to different kinds of work – says something of the prevailing cultural, political, and economic values and ideals held by a society at large, what Marx referred to as a society's "superstructure."

At the same time that labor reveals something of the structuring principles of society, it is also a very personal endeavor. What we do is a huge component of who we are, not only in terms of how we are valued and perceived by others or judged in relation to social values, ideals, and "norms", but also in how we understand and express our sense of being and engage in daily life. Identities are constructed, negotiated, and made meaningful only through social interactions, and labor activities provide an important medium for these interactions (Christiansen 1999; Costin 1998b; Meskell 2001; Robb 2010).

That labor activities reveal something of the organizing principles of a society and help shape peoples' social identities is not lost to archaeology, a discipline that relies primarily on material remains – the very products of human labor – for social analysis. Heavily influenced by Karl Marx's theory of class which argues that peoples' position in the social hierarchy is inextricably linked with their relationship to the means of production, archaeological investigations of labor have traditionally tended to focus on assessing political complexity and making inferences about elite strategies for control and legitimization (Brumfiel and Earle 1987; Childe 1950; Costin 2006; Peregrine 1991; Schortman and Urban 2004; Service 1971; Vaughn 2006). According to this perspective, often labeled as political economy, elites emerged and came to power by controlling all or some aspects of the production and/or distribution of goods, particularly corporate architecture and "luxury" or "exotic" items that served as symbols of status and power. Implicit in this perspective is the Marxian notion that social status is dependent upon the abilities to amass material goods/property (i.e. "wealth") and to control rather than participate in – the actual physical work of production (Babić 2005). Following this line of reasoning, archaeologists, particularly those who adhere to a Marxist perspective, have often taken as a given the notion of elites as consumers and non-elites as producers, thereby

failing to explicitly test the interplay of labor activities and social status (Inomata 2001; 2007).

The idea that strenuous labor and marginal social status go hand in hand seems almost "natural" in the context of Western society, where serfdom, slavery, and class distinctions between "blue collar" and "white collar" workers have a long history. But the degree to which we can project these perceptions of class onto complex societies of the ancient past is questionable, especially in light of ethnographic and archaeological examples that seem contrary to modern, western assumptions (Brumfiel 1998; Costin 1998a; Costin 1998b; Inomata 2001; 2007; Lass 1998; Meskell 2002; Reents-Budet 1998). In fact, there is good reason to believe that in traditional societies where craft specialization and social stratification were present, artisans involved in the production and/or acquisition of resources perceived to be of high value by one or more social groups may have in some cases achieved a special, privileged status because of, not despite, their labor activities (Helms 1993; Inomata 2001; Netherly 1977; Ramírez 2008; Shimada and Craig 2013; Uceda and Rengifo 2006). Furthermore, while there is little doubt that those of low social status tend to have a reduced quality of life compared to their "privileged" counterparts, the specifics of what this "privilege" actually entailed in ancient contexts need to be tested rather than assumed. Projecting modern day, eurocentric ideals that connect manual labor to marginal social status onto the past misses an important opportunity to explore what it actually meant to be "elite" or "plebeian" in terms of peoples' daily lives, an essential endeavor given the increasing emphasis on lived experience within archaeology over the last twenty years (Buikstra and Scott 2011; Fisher and Loren 2003; Hodder and Hutson 2003; Joyce 2005; Knapp and van Dommelen 2008; Knudson and Stojanowski 2008; Meskell 1996; Robb 2010).

Much of what has been written about the social division of labor in pre-Hispanic coastal Peru draws upon historical reference to lineage-based (real or fictive), ethnically-distinct, and in some cases highly-specialized work collectives, referred to as *parcialidades* by the Spanish (Costin 2016; Eeckhout 2008b; Isbell 1995; Netherly 1977; Ramírez 2007; Rostworowski 1975). Each parcialidad was headed by a lord or principal responsible for mobilizing labor to sustain higher-level regional leaders and infrastructure, in exchange for access to corporately-held resources (e.g. land, water), provisions, and other goods (Netherly 1977; Ramírez 2007; Ramírez-Horton 1981; Rostworowski 1975). Access to these resources was granted according to the respective rank of each *principal*, which was largely determined by the size of the labor force they commanded and the kinds of goods and services they provided (Netherly 1977; 1990). It has been postulated that the greatest labor demands were placed upon parcialidades engaged in seasonal activities such as fishing and farming, as they may have also been primarily responsible for extracting raw materials, transporting goods, and building and maintaining irrigation canals or other public works on a rotating basis (Netherly 1977; 1984; Ramírez 2007; 2008; Ramírez-Horton 1981). Full-time, specialized artisan groups – particularly potters, metalsmiths, and perhaps weavers as well – may have been largely exempt from this labor tax, known during colonial times as mita (Netherly 1977; Ramírez 2007). These groups fulfilled their obligations by providing goods that in some cases served as important symbols of elite ideology and power, and are thus assumed to have had a certain degree of social prestige (Helms 1993; Netherly 1977; Ramírez 2007).

If historical documentation of *parcialidades* reflect socio-economic systems in place prior to European contact, it may be reasonable to expect agreement between the degree and kind of physical tasks people engaged in and their place in the social hierarchy. This is especially true in the case of the north coast, where previous bioarchaeological studies highlight social status as the single best predictor of health in terms of nutrition and exposure to developmental insults (Farnum 2002; Shimada 2006). My research extends this discussion by investigating how – or even if – labor demands factored in to quality of life differentials already observed between elites and non-elites in Sicán society. Less is known about the nature and impact of social hierarchy in Ychsma society (Eeckhout 2013; Marsteller et al. 2017), and recent research has raised some interesting questions about the social identities of the Pachacamac burial sample included in this study (Shimada et al. 2006a; Shimada et al. 2010; Shimada et al. 2006b; Takigami et al. 2014). My work contributes to these discussions by looking for patterns in physical activity and mortuary treatment that might be informative with regards to the samples' social composition

## Research Orientation: The Bioarchaeological Approach

Bioarchaeology is a rapidly expanding branch of physical anthropology that utilizes human skeletal remains in concert with other kinds of archaeological, historical, and/or ethnographic data to test hypotheses and draw inferences about ancient societies and the human condition across space and time (Buikstra 2006b; Buikstra et al. 2011; Larsen 1997; Larsen 2018). Archaeologists have long appreciated the value of mortuary contexts for understanding hierarchy and social identity (Binford 1971; Brown 1995; Pearson 2000), and human skeletal remains provide the most direct means for exploring these issues in terms of life experience (Agarwal and Glencross 2011; Buikstra and Scott 2011; Knudson and Stojanowski 2008; Martin et al. 2013; Sofaer 2006). The human skeleton responds in observable ways to dietary practices, exposure to disease and trauma, and patterns of physical activity; precisely those aspects of experience expected to vary between different social groups. By establishing the normal range of variation in mortuary treatment and recognizing patterns in relation to time and basic components of identity revealed through skeletal analyses (such as age, sex, and geographic origins), it becomes possible to better recognize how other aspects of identity, such as social status, occupation, gender, and ethnicity may have been symbolized archaeologically (Brown 1995). When archaeological and skeletal data are combined, we can test hypotheses about the interaction of identity and experience, and such approaches continue to deepen our understanding of, for example, the division of labor, changes to peoples' quality of life over time in response to economic and/or political shifts, and the health consequences of belonging to particular status groups (Cohen and Armelagos 1984; Farnum 2002; Klaus et al. 2017a; Klaus et al. 2009; Klaus and Tam 2009; Meskell 1996; Shimada et al. 2004; Wesp 2014).

In 1977 when Jane Buikstra first outlined bioarchaeology as practiced in North America, she characterized the approach as part and parcel of processual archaeology's scientific agenda to move beyond the descriptive, typological, and often racialized research that characterized both archaeology and physical anthropology for the first half of the twentieth century (Armelagos 2003; Buikstra 1977; Buikstra 2006a). Processual archaeology was revolutionary in its attempt to explain – rather than just identify – cultural change in the archaeological record. Processualists defined culture as an adaptive strategy for dealing with specific environmental challenges, and were therefore interested in modeling relationships between culture, human organisms, and their environments that could be tested with archaeological data. Stuart Struever, an early and very influential proponent of processualism, argued this was only possible through multidisciplinary collaboration and greater integration of the natural sciences into archaeology (Brown and Struever 1973). His ground-breaking Illinois Valley Archaeology Program in Kampsville Illinois, of which Buikstra was a part, brought together a range of specialists who contributed to every phase of the project from inception to completion, perhaps for the first time in

archaeology's history. The goal at Kampsville was to reconstruct the ecological, biological, and cultural landscape of prehistoric populations living in the Lower Illinois Valley just prior to and during the transition to agriculture (i.e. the "Hopewell Tradition"), upon which explanatory models of social organization and cultural change could be built (Brown and Struever 1973). Buikstra envisioned bioarchaeology as a way to further promote and standardize this integrative approach, focusing her attention specifically on human skeletal remains and their value for addressing archaeological problems in a more "holistic" way.

Buikstra argues that the experiential information encoded in human skeletal remains is essential for reconstructing the "social worlds and personae" of ancient peoples (Buikstra 2006b; Buikstra et al. 2011), and her allegiance to a social-archaeological agenda likely stems from involvement with the Kampsville project early-on in her career. These experiences helped to articulate and strengthen her argument that the physical body, as an integral component of the cultural "landscapes" past peoples inhabited, is equally shaped by social processes and adaptive strategies detectible in other kinds of archaeological materials. Buikstra not only sees bioarchaeology as a way to realize the potential of human skeletal data for understanding past social dynamics, she also advocates for their careful contextualization, recognizing that social, cultural, and ideological processes not only influence patterns of health and behavior identified in mortuary samples, but also who is, and who is not, represented within them (Knüsel 2010).

Buikstra's bioarchaeological approach attempts to synthesize method and theory from two fields of study, skeletal biology and archaeology, traditionally subsumed under rather disparate branches of anthropology. It is therefore perhaps no surprise that despite resounding support, the discipline has been slow to fully implement her agenda (Goldstein 2006; Kakaliouras 2017; Rakita 2014). For example, efforts to engage social theory in a more critical, self-reflexive way have lagged far behind those put forth in archaeology, a legacy that bioarchaeologists continue with increasing fervor to distance themselves from (Agarwal and Glencross 2011; Baker and Agarwal 2017; Becker and Juengst 2017; Kakaliouras 2017). Part of the reason for this stems from competing ideas about what bioarchaeology should be that emerged as the discipline was coming into full fruition during the late 1990's (Rakita 2014). Because these two viewpoints have provided an important impetus for more recent trends, they are worthy of further exploration.

In contrast to Buikstra's approach that advocated bioarchaeology as a way to address social questions within specific sociocultural, temporal, and spatial contexts, others saw the approach primarily as a strategy for identifying the biological consequences of major behavioral shifts in human history, such as the transition to agriculture (e.g. Cohen and Armelagos 1984) or European colonialism (e.g. Larsen and Milner 1994). Because the questions raised by this approach are fundamentally biological, there is less consideration of archaeological context beyond that required to establish basic population parameters (i.e. ecology, subsistence, political structure), with models of physiological stress and adaptation taking precedence over social theory (Agarwal and Glencross 2011; Baker and Agarwal 2017; Goldstein 2006; Goodman 1998; Kakaliouras 2017; Rakita 2014). Clark Larsen, often characterized as the "face" of this approach, emphasizes methodological rigor and sophistication – rather than engagement with social theory – as key to bioarchaeology's vitality. Larsen promotes bioarchaeology as a strategy for understanding the origins and development of the modern world and issues that continue to affect us today (e.g. inequality, migration, epidemics), and therefore aligns the approach more broadly with disciplines outside of archaeology such as history, economics, public health, and even "hard" sciences like chemistry and physics (Knüsel 2010; Larsen 1997; 2017; 2018).

Buikstra and Larsen's approaches to bioarchaeology have historically been cast in opposition to one another, and while the resulting debates over engagement with social theory and adaptationist perspectives still loom large in present discussions, this distinction is becoming less relevant as the discipline matures (Baker and Agarwal 2017; Larsen 2018; Rakita 2014). Larsen's methodological focus, reliance on biological theory, and promotion of bioarchaeology's relevance to present-day concerns, have helped pave the way for the renaissance the discipline has been experiencing since the late 1990's (Rakita 2014). For example, the extensive collection of human skeletal data amassed to explore adaptive strategies in response to large-scale change was also used to test assumed relationships between skeletal lesions and overall health, ushering in a more critical stance regarding the representativeness of bioarchaeological samples (e.g. Wood et al. 1992; Wright and Yoder 2003). This in turn spurred the development of improved methods for estimating age and sex from skeletal remains and technological advances such as stable isotope and DNA analyses that have greatly expanded bioarchaeology's interpretive power (Baker and Agarwal 2017; Larsen 2018). Theories of stress and adaptation continue to play an important role in the biocultural approach, which promotes the inclusion of political-economic perspectives to critically explore "the relationship between biological well-being and inequalities in access to and control of resources" (Goodman 1998: 148). The biocultural program, as articulated in Alan Goodman and Thomas Leatherman's seminal book Building a New *Biocultural Synthesis* (1998), argued that conventional adaptive approaches – with their emphasis on biological reactions to large-scale factors (e.g. ecology, subsistence) - tend to cast societies as homogenous, functional units and obscure important within group differences linked to ideology and power relations. A political-economic approach not only helped to better align bioarchaeological research with themes of identity and power incorporated into archaeology as

part of the post-processual movement (discussed further in chapter 2), it was also seen as a way for bioarchaeology to become more sensitive and responsive to the issues facing present day indigenous peoples, descendant communities, and other marginalized groups (Armelagos 2003; Blom 2017; Goodman 1998; Kakaliouras 2017; Knüsel 2010; Martin 1998). This second point became exceedingly important in the wake of the Native American Graves Protection and Repatriation Act (NAGPRA), a law enacted in 1990 in response to mounting tensions between indigenous groups and researchers that requires explicit consent from affiliated tribes before excavating or analyzing Native American cultural items, including human remains.

There is little doubt that Buikstra's vision for bioarchaeology, with emphases on context and the application of social theory, continues to exert the strongest influence on how the discipline is practiced today, and even Larsen himself has more recently acknowledged the value of incorporating social theory into bioarchaeological studies (Baker and Agarwal 2017; Larsen 2018). Researchers are beginning to expand engagement with social theory, moving beyond a consideration of skeletal data with well-known socio-cultural significance to address "messier" concepts such as community (e.g. Becker and Juengst 2017), caregiving (e.g. Tilley 2015), or personhood (e.g. Boutin 2011), to give a few examples. As Kakaliouras (2017) succinctly puts it:

"Bioarchaeologists have all too often employed a Tylorian comparative method to ground their interpretations in something that is already known, such as a cultural practice that leaves marks on bone, or a set of behaviors that are expressed skeletally in similar ways ... here I wish to stretch a smidgen past that ... even if that means being left in a place of not knowing the culturally-specific significance of a feature, adaptation, or cultural modification" (pg. 14).

This move within bioarchaeology to address more complex sociocultural phenomena has been important for challenging some of the assumptions and biases influencing archaeological work on identity, particularly in relation to ideas about the relationship between sex and gender, biological age and "adulthood," as well as ethnicity and status (Buikstra and Scott 2011; Knudson and Stojanowski 2008; Sofaer 2006). And yet, despite the potential for bioarchaeological research to explore identity in a more nuanced way, few studies have integrated skeletal and archaeological data to test conventional ideas about labor demands and social status (Milella et al. 2015; Porčić and Stefanović 2009; Robb et al. 2001; Schrader 2014). I highlight this under-utilized avenue of bioarchaeological inquiry in my research, contributing to current debates regarding the future of activity studies within the field, and offering new data. The research presented here highlights this under-utilized avenue of bioarchaeological research, contributing to current debates regarding activity studies within the field and offering new data to enhance current understandings of labor division and social stratification in Middle Sicán and Ychsma societies.

## Organization of the Dissertation

This section provides a synopsis of the dissertation chapters, outlining the major themes to be addressed in each. Chapter 1 introduces the research objectives, hypotheses, and problem context for this study. Here I discuss common perceptions about social status and manual labor based on historical and modern examples, and the need to test whether such assumptions have relevance for understanding social experience and the division of labor in ancient societies. I outline the bioarchaeological approach employed to test assumed relationships between social status and manual labor using archaeological human remains and their associated mortuary contexts, and explain why these lines of evidence are particularly useful for addressing this problem.

Chapter 2 details the theoretical framework for my research. I define social status as understood here, discuss the theory of embodiment that relates status identity to biological wellbeing, outline the rationale for using mortuary treatment to infer social status, and present theories adopted to understand relationships between bone biology and physical activity.

Chapter 3 reviews the current literature pertinent for understanding the strengths and limitations of activity reconstructions based on patterns of skeletal trauma, enthesial changes, and degenerative joint disease. Recently proposed changes to the scoring and interpretation of ECs are explored, and then commented on further in the final chapter. The second section of this chapter summarizes previous research on social stratification, labor organization, and quality of life differences between elites and commoners in Sicán and Yschma societies just prior to Inca and subsequent European conquest. These works provide the foundation upon which my hypotheses are built and are useful for identifying gaps in our understanding of status identity and labor activities to be explored in this study.

Chapter 4 describes the research sample used, as well as the specific methods employed to infer status identities from mortuary treatment and record skeletal evidence of trauma, ECs, and DJD associated with strenuous physical activity. Also detailed in this chapter are the statistical techniques and associated data treatments used to explore the dataset and test my hypotheses.

Chapter 5 explains the results of the statistical analyses, and patterns of trauma, DJD, and ECs are considered separately. The dissertation concludes with Chapter 6, which summarizes the results, offers an interpretation of the data with respect to social status and the division of

labor in pre-Hispanic coastal Peru, discusses the implications of this research for bioarchaeological studies of activity, and highlights future directions for research.

#### CHAPTER 2

## THEORETICAL ORIENTATION

#### Social Status & Life Experience

Conventional understandings of social status rely upon Marx's idea that power and inequality are essentially the products of economic relationships, and there is often little distinction made between concepts such as wealth, status, and class (Babić 2005). This perspective tends to cast social stratification in a dichotomous light – the privileged elite "bourgeoisie" (i.e. the owners of the means of production) versus the lowly proletariat (i.e. the working class), but nevertheless establishes a direct connection between material objects and social relations, with obvious relevance to archaeology (McGuire 1992).

Recognizing that social life is more than just the sum of economic activities, Max Weber extended Marx's view of social stratification to consider the role of prestige, or one's ability to live up to a certain set of principles held important by society at large or by social groups (Kerbo 2009). For Weber, social classes were not just economic units, but rather collections of people sharing a common lifestyle based on similarities in material as well as social conditions (Kerbo 2009). Informed by these understandings of social stratification, I consider social status to be the cumulative index of one's access to material and social capital (i.e. wealth and prestige) that manifests as differential behaviors and life experiences. These differences are encoded not only in the objects made and used throughout the course of daily life and the spaces that people inhabit, but also in the physical body. My research adopts the perspective that status is paramount to social identity and life experience, relying on the concept of embodiment to explain interactions between social life and biological well-being. Embodiment as defined here is the process by which social conditions and experiences are incorporated into the physical body. Embodiment is the recognition that the way we move, our postures, how we decorate and adorn ourselves, the kinds of foods we eat, the environments we are exposed to, the challenges we face, and the strategies we use to survive are the products of the societies in which we live, our cultural values, and our social relations.

Archaeological theories of embodiment draw largely from the works of Maurice Merleau-Ponty, Marcel Mauss, and Pierre Bourdieu (Crossland 2010; Csordas 1994; Joyce 2005; Meskell 1996). Merleau-Ponty's ideas reflect the phenomenological approach that attempts to break down traditional philosophical dichotomies such self-object and mind-body (Csordas 1994). According to Merleau-Ponty, the body, through experience, provides the basis of perception, and this experience or "being-in-the-world" in turn shapes the body (Csordas 1994). Because humans come to understand and engage with the world through corporeal experience, the body is "both the medium and the product of social relations" as well as the "locus of identity formation" (Joyce, 2005: 142). Mauss and Bourdieu offer the concept of *habitus*, which argues that bodily practices are not only a manifestation of personal history, they are also unconsciously patterned by socio-cultural values and expectations (Crossland 2010; Knapp and van Dommelen 2008). Bourdieu's work in particular focuses on social hierarchy, arguing that patterned behaviors (especially aesthetic tastes) emerge as a result of early indoctrination into a particular social class (Bourdieu 1984). From these ideas comes the realization that our bodies are not simply biological entities formed by genes and natural processes, they are the cumulative result

18

of our individual experiences and histories, as well as our social and material realities (Krieger 1999; Krieger 2001).

The plasticity of the human body – our biological responsiveness to environmental stimuli- has been key to our success as a species, but also leaves us extremely vulnerable to circumstance. The most poignant example of this is the decades of clinical research showing that those with marginalized social status are disproportionately affected by disease and premature death (Chandola and Marmot 2011; CSDH 2008; Marmot et al. 1978; Marmot and Smith 1991). Furthermore, these health disparities tend to emerge even when wealth and educational factors are held constant, highlighting the role of beliefs and value systems in social stratification. Take as a an example the observation that African American and Latino communities in the United States - regardless of income and level of education - tend to experience poorer health than their white counterparts (Gravlee 2009). Genetic studies disprove the idea that these health disparities are the result of innate biological difference, and researchers have instead focused on the ways in which ideological factors, such as institutionalized racism and discrimination, can affect stress levels and in turn, immune function and overall health (Gravlee 2009; Krieger 1999). Clinical research on social health disparities situate human bodies within the appropriate context of social experience, and offer an important model for bioarchaeologists of how status identities are embodied.

## Social Status & Mortuary Treatment

One of the greatest debates within archaeology is the degree to which mortuary contexts are reliable indicators of social organization and individual social status identities. At one

19

extreme are proponents of the "New Archaeology" school, or the "processualists," whose approach is often characterized by the works of Arthur Saxe (1970) and Lewis Binford (1971). New Archaeology emerged during the early 1960's with the goal of moving the discipline away from cultural and chronological typologies to a more scientific approach that could test hypotheses about past human cultures and societies and offer explanations for processes responsible for cultural variation (Pearson 2000). Using cross-cultural ethnographic data to establish behavioral correlates for archaeological patterns, processualists argued that the degree of variability in mortuary contexts reflected the level of social complexity within the society, as well as the status and rank of individuals during life (Binford 1971; Saxe 1970; Tainter 1978). Brown (1995) characterized this perspective as "representationalist" because proponents tended to view mortuary contexts as a "mirror" of social processes and the status identity of the deceased. Although often accused of over-simplification, logical positivism, and indiscriminate use of ethnographic analogy (Hodder 1980; Shanks and Tilley 1987; Ucko 1969), Brown and others have noted that many processual views and cross-cultural generalizations continue to play a dominant role in archaeological studies, especially with regards to mortuary contexts (Brown 1995; Hegmon 2003; Pearson 2000).

At the other end of the spectrum are the "post-processualists," who questioned the assumed one-to-one relationship between mortuary variability and social complexity, as well as between burial treatment and social status. Pointing to the fact that "the dead do not bury themselves," post-processualists did not view mortuary contexts as a passive reflection of social organization or the "social personae" of individuals, they instead viewed them as dynamic spaces where ideologies of power and domination were negotiated, constructed, and even concealed or misrepresented (Arnold and Jeske 2014; Brown 1995; Pearson 2000). Rather than relying on

social factors to explain mortuary variability, post-processualists emphasized ritual and symbolic aspects (Brown 1995). In its most extreme form, post-processualism questioned whether it was even possible to accurately interpret mortuary contexts, given their tremendous variability and significance only in relation to specific socio-cultural and historical contexts (Hodder 1980; Trigger 1991; Ucko 1969). However, rather than sealing the nail in the coffin of mortuary archaeology, so to speak, post-processualist critiques provided valuable insights for improving the field. They emphasized human action and experience, and championed holistic, contextual analyses (Pearson 2000). Out of this sprang greater consideration of the sentiments, actions, memory, and processes involved in creating mortuary assemblages (Tarlow 1999), and a growing awareness that valid interpretations – especially with regards to social status – are only possible when mortuary data are analyzed in concert with other lines of evidence, such as settlement patterns and indices of skeletal health (Ashmore and Geller 2005; Goldstein 2006; Trigger 1991).

By the 1970's and 1980's, the chasm between processual and post-processual archaeologies was on the wane as researchers began advocating for "middle ground" approaches (Pearson 2000), or what Hegmon (2003) referred to as "processual-plus." While recognizing that ethnographic data should be used cautiously, that mortuary contexts may reflect idealized rather than realistic versions of the social order, and that mortuary data are just one of many lines of evidence needed to make sound inferences, there is still general consensus that mortuary assemblages are patterned in socially-meaningful ways, so that distinctions in power and prestige are discernible (Arnold and Jeske 2014; Brown 1995; Carr 1995). For example, there is wide ethnographic and archaeological support for Saxe and Binford's assertion that the allocation of resources expended on funerary activities will be directly related to the number of individuals

with social obligations to the deceased (Brown 1995). Brown (1995) argues that because every society has a finite amount of resources that can be expended, it is possible to establish the "normal" range of variation in mortuary treatment and thus identify unusual cases suggestive of special social significance. Elite burials can be identified archaeologically using estimates of the amount of time and resources dedicated to the treatment of the corpse, the architecture of the internment facility, and the material contribution to the funerary ritual (Tainter 1978). It is assumed that because elites occupy a larger number of social roles than the average citizen, their deaths will inspire greater community involvement (i.e. more energy expended on funerary treatment), and symbolic representation in the mortuary context will be more diverse (Howell and Kintigh 1996).

Bioarchaeology provides an important platform for integrating processual and postprocessual perspectives, and the field has contributed to, and benefited from, both approaches. New Archaeology's call to be more "scientific" and the post-processual emphasis on experience resulted in greater appreciation for human skeletal remains as an important source of social information (Sofaer 2006). By highlighting the body's role in social practice, post-processualism has also helped generate a more socially-relevant bioarchaeology, serving as a catalyst for enhanced dialogue between the field and the social sciences more broadly conceived (Knudson and Stojanowski 2008). Today's bioarchaeologists are trying harder than ever not only to apply social theories to human skeletal datasets, but also to actively engage in developing and testing them. Bioarchaeological approaches have been particularly fruitful with regards to ancient social identities, and my study of status and labor in pre-Hispanic coastal Peru is part of a larger push to situate interpretations of the past within the appropriate context of daily life and experience.
### Bone Biology and Physical Activity

My research is guided by the basic premise that labor organization – and by extension, patterns of physical activity – are socially and culturally meaningful, particularly with regards to the embodiment of status identity. Status divisions organize relations of power to determine one's rights, responsibilities, and expected behaviors, and therefore one's social position is often expressed most clearly in terms of the activities they engage in (Porčić and Stefanović, 2009). Social status, patterns of physical activity, and human skeletal data are linked together here by the underlying principle of biological plasticity. Just like any other living tissue, bone is malleable, and responsive to both the internal and external environment. This plasticity is precisely what enables bioarchaeologists to draw meaningful inferences about life experience from human skeletal remains (Sofaer 2006).

Bioarchaeological studies of physical activity can take several forms, all with their own set of strengths and limitations. Each technique relies on the concept of bone functional adaptation, which states that bone will remodel in the direction of mechanical loading, and will increase or decrease to reflect mechanical requirements (Kennedy 1989; Ruff et al. 2006). Detailed descriptions of the skeletal traits used in my analysis (fractures, Schmorl's nodes, degenerative joint disease [DJD], and enthesial changes [EC]) are provided in the Literature Review and Methods sections.

The skeletal trauma considered in this study – Schmorl's nodes and fractures –are common sources of pain and debility in living populations, and clinical studies can therefore offer important interpretive models for bioarchaeological data. Careful assessment of skeletal trauma offers an opportunity to address aspects of life experience that are typically not accessible

through other lines of evidence, although identifying the precise causal mechanism – accidental injury, violence, cultural modification, or funerary processing – can be difficult. While the relationship between biomechanical stress and skeletal injury is rather straight-forward, interpreting patterns of DJD and ECs requires more caution (Cardoso and Henderson 2010; Jurmain and Kilgore 1999; Kennedy 1989; Weiss 2007; Weiss and Jurmain 2007). Cadaveric studies and research in occupational and sports medicine have supported the idea that strenuous physical activity can induce these osseous changes (Bramhall et al. 1994; Brown et al. 2008; Chen and Bohrer 1990; Fahey et al. 1998; Parfitt 2004; Partridge and Duthie 1968; Rapillard et al. 2006; Schlecht 2012a; Swartz et al. 2005; Villotte et al. 2010; Weiss 2015). However, in addition to being influenced by physical activity, age, sex, body size, dietary factors, hormones, and genetics can also play important roles (Jurmain et al. 2012; Weiss 2007; Weiss and Jurmain 2007; Wilczak 1998).

Out of necessity, inferences about activity using evidence of DJD and ECs are rather general. It is not possible to determine the specific occupation(s) individuals engaged in during their lives, as a wide range of activities can use the same joints and muscles in similar ways (Jurmain and Kilgore 1999; Kennedy 1989; Robb 1998). At best, one may be able to use these traits to distinguish those who performed high intensity tasks from everyone else, with finer distinctions being more difficult to discern (Jurmain et al. 2012; Molnar 2006; Weiss 2007; 2015). Furthermore, there is some evidence that muscle loading intensity in adulthood beyond a threshold established during skeletal growth and development (particularly in adolescence) is more important to the development of ECs than are loading frequency and duration, so that "low-intensity" activities (i.e. do not reach the peak loading capacity of the muscles) or those within the limits of one's "adaptive range" established during adolescence – even when performed

habitually – may not lead to ECs at all (Acosta et al. 2017; Chapman 1972; Churchill and Morris 1998; Henderson 2013; Schlecht 2012a; Toyne 2002).

•

### CHAPTER 3

### LITERATURE REVIEW

# **Biological Indicators of Physical Activity**

Bone is made of inorganic (hydroxyapatite) and organic (collagen) components that provide both strength and flexibility to support the body and facilitate movement. Bone undergoes constant cycles of proliferation, resorption (destruction of bony tissue) and maintenance throughout the lifespan – often simultaneously –to adapt to physical stresses, access calcium and phosphate reserves, and also as a normal part of growth and development (White and Folkens 2000). Bone proliferation is achieved by osteoblast cells, which produce a collagenrich matrix that calcifies as hydroxyapatite crystals are incorporated. Once encased in the calcified matrix, these cells serve to maintain bone tissue and are referred to as osteocytes (White and Folkens 2000). Resorption occurs as osteoclast cells dissolve collagen and remove the mineral matrix. Normal patterns of bone deposition, resorption, and maintenance can be disrupted by nutritional inadequacies, as part of the inflammatory response to infectious agents and trauma, and as a result of mechanical loading (Larsen 1997). This can result in bony overgrowth and/or resorptive lesions at specific sites throughout the skeleton that are often permanent and therefore visible in bioarchaeological assemblages.

Bioarchaeologists have long recognized the need to consider the effects of age, body size, and perhaps sex when using patterns of DJD and EC to understand activity patterns, and have accounted for these factors using a variety of statistical procedures (Cheverko and Hubbe 2017; Henderson and Nikita 2016; Nikita 2014). DJD and ECs are almost always more common and more severe in older and larger individuals than in younger and smaller ones, with size usually referring to body mass for DJD and to long bone measurements for ECs. While DJD tends to be more common in females compared to males, the opposite is true for ECs. Sex-based differences appear - at least for EC - to largely be a function of size (i.e. males tend to be larger than females), but clinical studies of DJD suggest hormonal factors are also at play (Jurmain et al. Specific genes that affect DJD severity have been identified, although thus 2012; Weiss 2015). far none have been implicated in determining whether the condition develops or not in the first place (Spector and MacGregor 2004). Hormones and dietary factors have been shown to play a role in the development of DJD, but these influences have remained largely unexplored in bioarchaeological contexts and with respect to EC (Jurmain et al. 2012). In order to produce interpretations of these skeletal traits that are meaningful with regards to patterns of activity, it is essential that bioarchaeologists account for as many confounding variables as possible, which in most studies has been age, body size, and/or sex. Different strategies for dealing with these issues will be discussed further in the methods section (chapter 4).

# Skeletal trauma

Skeletal fractures occur when tensile, compressive, twisting, bending, and/or shearing forces exceed the bone's natural strength, causing displacements (when two surfaces are no longer in anatomical alignment) or the appearance of fracture line (Davidson et al. 2011). When fracture lines occur, they originate at the point of impact and radiate out in a concentric pattern, usually intersecting preexisting lines of weakness (e.g. cranial sutures; Davidson et al. 2011). Bending stresses often lead to transverse fracture lines, while shearing causes horizontal fracture

lines and torsion produces lines that are spiral-patterned (Aufderheide and Rodríguez-Martín 1998). Tension may result in joint dislocations or avulsions (the tearing away of small, irregular bone fragments), while compressive forces can lead to crushing or impactation (Aufderheide and Rodríguez-Martin 1998). Fractures can be the result of a single traumatic event, repeated stress under static or dynamic loading, or abnormal weakening of the bone due to pathological processes, and are often accompanied by degenerative changes in the corresponding joints (Aufderheide and Rodríguez-Martin 1998).

In the long bones, fractures can be recognized by the presence of a bony callus that forms as part of the healing process, shortening and/or mis-alignment of the shaft, or the formation of a "pseudo-joint" (Aufderheide and Rodríguez-Martin 1998). Figure 1 provides a comparative perspective of a normal and fractured clavicle to illustrate these changes. A pseudo-joint forms when fracture causes non-union between two portions of long bone or between two joint surfaces. As a result, the two ends of the bone become rounded with sclerosis of the medullary cavity (Aufderheide and Rodríguez-Martin 1998). Vertebral fractures are largely the result of compressive forces, and are recognized by crushing or impactation of one or more vertebral bodies, oftentimes with subsequent fusion to adjacent vertebral bodies (Figure 2). Cranial fractures can be recognized by the presence of beveled or smooth punctures, indentations, and circular depressions that may extend inward into the cranial space (Aufderheide and Rodríguez-Martin 1998).

Schmorl's nodes are lytic-like lesions in the superior and/or inferior surface of vertebral bodies (Figure 3), and have been associated with acute trauma to the spine in the clinical literature (Faccia and Williams 2008; Fahey et al. 1998; Peng et al. 2003). These lesions are commonly found in the elderly and juvenile gymnasts and are thought to occur when there is a rupture in the intervertebral discs between two adjacent vertebral bodies. In some cases Schmorl's nodes can be rather painful and in others they are asymptomatic (Peng et al 2003).

# Degenerative joint disease (DJD)

The best support for a relationship between DJD and physical activity comes from experimental and clinical research in occupational or sports medicine, particularly studies that focus on specific patterns of involvement for professional athletes and farmers (Brown et al. 2008; Sulsky et al. 2012; Weiss and Jurmain 2007). However, clinical data has also suggested that this relationship is more clear for some regions of the body than others, with degenerative changes in the spine being considered least reflective of differential activity patterns, given its unique susceptibility to the stress and strains of the bipedal gait common to all humans (Weiss and Jurmain 2007). While there is lack of clinical support for a relationship between osteophytosis (the development of bony spurs at the margins of vertebral bodies) and patterns of physical activity (Rogers and Waldron 1995; Weiss and Jurmain 2007), Brown et al. (2008) did find statistically significant, positive associations between load-bearing, cartilage destruction, and DJD scores in the facet joints of cadaveric spines. Their work does offer some support for including DJD of the facet joints in bioarchaeological studies of activity.

Clinical research has also been helpful for outlining "best practices" in the scoring and interpretation of DJD. Skeletal changes associated with DJD include osteophytes (bony protrusions at the margins of joint surfaces), porosity, and eburnation (polished, opaque appearance on the articular surface). Although standard bioarchaeological protocol for documenting DJD outlined in Buikstra and Ubelaker (1994) calls for the evaluation of all three of these characteristics, there is little clinical support to suggest that osteophytes and porosity in



Figure 2 - Normal right clavicle (left); fractured right clavicle (right)



Figure 3 - Thoracic compression fracture (left), Schmorl's node (right)



Figure 4 - Eburnation of the distal femur

isolation are sufficient evidence of DJD (Rogers and Waldron 1995; Rothschild 1997; Weiss and Jurmain 2007). Several researchers specializing in DJD advocate a conservative approach that identifies the condition only when two ormore skeletal signatures are present, with eburnation being the sole definitive indicator (Rogers and Waldron 1995; Rothschild 1997).

# Enthesial changes (ECs)

Entheses are junction points of hard and soft tissues where muscles insert into bone, either directly via tendons or indirectly via the periosteum, a layer of vascular connective tissue that surrounds long bones (Benjamin et al. 2002). Their primary functions include anchoring tendons to resist static and dynamic loading, and balancing tensile loads to dissipate stress outward into the bone and/or tendon (Benjamin et al. 2002). Entheses can be grouped into one of two categories, based on general morphological characteristics. Fibrous entheses tend to be located along large sections of long bone shafts, and the soft tissue components attach directly into the bone via collagen "Sharpey's" fibers, which are physically anchored to the bone matrix (Shaw and Benjamin 2007). Fibrocartilaginous entheses, in contrast, tend to be located in smaller regions of rather thin cortical bone devoid of periosteum at the ends of long bones, with muscles attaching via tendons (Shaw and Benjamin 2007). Fibrocartilaginous entheses are composed of four histological "zones," including: 1) the tendon or ligament; 2) a layer of uncalcified fibrocartilage; 3) calcified fibrocartilage; and 4) subchondral bone (Benjamin et al. 2002). The surface of zone 3, the calcified fibrocartilage, is the portion of the entheses that is visible in dry bone (Villotte et al. 2010). While the distinction between entheses types as presented here has important implications for the scoring and interpretation of ECs (see below for further discussion), it is important to remember that they are not absolute and different portions of the

same entheses can have different characteristics in relation to tensile loading (Cardoso and Henderson 2010; Shaw and Benjamin 2007). For example, collagen fibers at the peripheral margins of fibrocartilaginous entheses often merge with the periosteum, leaving only the innermost portion avascular (Villotte et al. 2010).

The idea that enthesial changes (ECs) at fibrous attachment sites are indicative of intensive muscle use and therefore reflect (at least to some extent) differential activity patterns is based on the premise that forceful muscle contractions increase blood flow to the periosteum, which in turn initiates the proliferation of bone cells and thus, surface irregularities (e.g. bony ridges, porosity, lipping at enthesial margins; Schlecht 2012a; 2012b; Shaw and Benjamin 2007). Because fibrocartilaginous entheses are largely devoid of periosteum – except at the peripheral margins as explained above – and the associated cortical bone may be incapable of supporting deeply penetrating Sharpey's fibers, an alternative explanation for their response to mechanical loading is required (Shaw and Benjamin 2007; Villotte and Knüsel 2012). Shaw and Benjamin (2007) suggest that the fibers crossing the uncalcified and calcified zones of a fibrocartilaginous entheses may be equivalent to the Sharpey's fibers found in fibrous insertions. In this case, bony changes similar to those found with degenerative joint disease (i.e. pitting and other surface irregularities) may occur as soft tissue structures wear down, placing more stress and strain on the bone (Shaw and Benjamin 2007). However, unlike DJD which is thought to reflect habitual and/or repetitive episodes of biomechanical stress and strain, there is some research to suggest that ECs are more likely the result of sporadic incidences of high intensity loading beyond one's adaptive capabilities (Acosta et al. 2017; Henderson 2013; Schlecht 2012a; 2012b). Entheses are, after all, designed specifically to deal with the stress and strains of mechanical loading, and given bones' plastic and adaptive nature, it should perhaps be expected that a significant

morphological response would only occur in the presence of "atypical" and intermittent loading regimes (Schlecht 2012b).

Some support for the idea that increased bone remodeling occurs at muscle attachment sites in response to biomechanical stress was found by Schlecht (2012a) in his study of insertion sites using a sample of radii from human cadavers, where elevated rates of bone remodeling were observed histologically. Previous histological studies testing the relationship between enthesial morphology and mechanical loading have largely been conducted using animal models, with conflicting results and limited relevance to humans. Schlecht's (2012a) pioneering research lends credence to activity studies using EC, and future work that considers histological assessment of bone remodeling and enthesial morphology in tandem will help to further clarify the role of mechanical stress and strain in EC etiology (Schlecht 2012a).

Bioarchaeological and experimental research on ECs have yielded mixed results, with several studies failing to discern meaningful patterns despite known or inferred activity differences (Cardoso and Henderson 2010; Churchill and Morris 1998; Toyne 2002; Zumwalt 2006). Likely problems with scoring techniques and a lack of consistency in statistical procedures have complicated this issue further, making it hard to compare different studies in a meaningful way (Cheverko and Hubbe 2017; Henderson and Nikita 2016). The most widely used scoring method for EC provided by Hawkey (1988) and Hawkey and Merbs (1995) do not take into account significant differences in the anatomy, "normal" range of variation, and responses to biomechanical stress found between different kinds of muscle insertions (Jurmain et al. 2012; Villotte et al. 2010; Weiss 2015). Importantly, Villotte et al. (2010) and Henderson et al. (2016) presented a new method that does address these concerns and has in some cases yielded greater correspondence between manual labor and severity of EC scores for skeletal samples where age, sex, size, and occupation are known (Michopoulou et al. 2015; Niinimäki 2011; Villotte et al. 2010; Weiss 2015). Dubbed the "Coimbra Method," this approach developed out of a series of workshops between 2009 and 2013 at the University of Coimbra's Research Center for Anthropology and Health in Portugal with the goal of tackling some of the most pressing concerns with regards to the utility of ECs as skeletal markers of activity (Santos et al. 2011). These workshops and the resulting publications represent a huge paradigm shift in the way ECs are analyzed and interpreted, with three dominant themes emerging: 1) all ECs – regardless of the insertion type – demonstrate positive correlations with age; 2) some insertion types perform better with regards to activity reconstruction than others; and 3) more research is needed to fully understand the etiology of ECs and better evaluate their utility in bioarchaeological studies of activity (Goode et al. 2018; Henderson and Cardoso 2013; Jurmain et al. 2012; Niinimäki 2011). Methodological considerations resulting from these discussions will be addressed more thoroughly in chapter 4.

Clinical evidence of correspondence between musculoskeletal over-use injuries and osseous changes at fibrous insertion sites is rare, and it may be exceedingly difficult to differentiate between normal variation in skeletal robusticity at these sites, and that related to biomechanical stress (Villotte 2006; Cardoso and Henderson, 2010; Villotte et al., 2010). Given this information, Villotte (2006), Cardoso and Henderson (2010), and Villotte et al (2010) argue against the inclusion of fibrous attachment sites in bioarchaeological reconstructions of activity patterns. It is quite possible that previous studies of ECs have obtained conflicting results because what was being measured was actually normal variation, rather than pathological conditions associated with increased biomechanical stress (Villotte et al. 2010). In fact, while Cardoso and Henderson (2010) were not able to distinguish between occupational groups in a

well-documented skeletal collection (i.e. known age, sex, and occupation) using Hawkey and Merbs' (1995) methods, when only fibrocartilaginous insertions were considered in the same samples, Villotte and colleagues (2010) found significantly more ECs among workers involved in forceful tasks (i.e. where muscles were loaded to near-peak capacity).

Villotte et al.'s (2010) study is important because it demonstrates that - contrary to recent criticisms of activity studies in bioarchaeology – careful, clinically-informed evaluations of ECs can provide accurate information about differential exposure to mechanical stress in past populations. By providing a way to enhance archaeological models of socio-economic organization with information about life experience, bioarchaeological activity studies can contribute greatly to our understanding of the past. The Coimbra method presents a step in the right direction towards enhancing the utility of ECs as markers of physical activity, but there are some important caveats that warrant further attention. The method has only recently been made available for widespread use, and its applicability to muscle sites and skeletal samples outside of those used to initially create and test it remains to be seen. Furthermore, Villotte et al. (2010) made no attempt to control for body size even though numerous studies have found this variable to exert strong and statistically significant influence on EC scores (Michopoulou et al. 2015; Weiss 2007; 2015). Examining statistical associations between EC scores, size, and age for the different enthesial types will allow me to shed some light on these issues and assess the merit of Villotte's claims.

#### Middle Sicán and Ychsma Social Organization & Burial Practices

The emergence of the Sicán and Ychsma archaeological cultures corresponds to the transition from the Middle Horizon (550 – 900 CE) to the Late Intermediate Period (900-1476 CE), a time characterized by significant sociopolitical re-organization following the decline of the expansive Wari (south and central coasts, highlands) and Moche (north coast) polities in Peru (Covey 2008). Despite changes in sociopolitical organization, Late Intermediate Period sites on the central and northern coasts demonstrate clear evidence of settlement hierarchy (i.e. distinctions between civic/ceremonial structures, and elite vs. common-class residences), urban centers, extensive road and irrigation networks, and occupational specialization (Covey 2008; Shimada 1994). These features demonstrate well the large-scale monopolization of land and labor, offering valuable clues about Sicán and Ychsma socio-economic organization.

The Middle Sicán archaeological culture (900 – 1100 CE) is known for a distinct ceramic style including burnished blackware vessels, copper-arsenic and gold alloy objects, a distinct religious ideology centered on the Sicán Diety and his human equivalent the Sicán Lord, monumental religious architecture, elite shaft tombs containing an unprecedented amount of material wealth, and the development of extensive trade networks with Ecuador, Colombia, and Amazonia (Shimada 1994; Shimada 2000). The emergence of Sicán culture corresponds to the decline of the Moche polity, which at its height covered roughly one third of the entire Peruvian coast (Shimada 1994). The Late Moche capital at Pampa Grande was abruptly abandoned following a severe flood spurred by El Niño events sometime around 700 CE (Shimada 1994), and this marked a period of population movements and transition intensified by Wari expansion from the highlands into coastal areas (Segura and Shimada 2010). Wari expansion resulted in a

synthesis of local and foreign elements in the north coast, and this melding of traditions in Middle Sicán iconography and material culture is perhaps responsible for its rapid spread throughout the north coast and further south (Segura and Shimada 2014; Shimada and Craig 2013).

The Sicán Precinct is a complex of six monumental mounds (Huacas Loro, Las Ventanas, La Merced, Lercanlech [also known as Rodillona], El Corte, and Soltillo) surrounding a large plaza (Shimada 2010). Each of the six ceremonial structures and associated cemeteries at the Sicán Precinct are thought to have represented an elite lineage that although united by shared religious ideology, were perhaps distinct with regards to their economic endeavors, as well as their social and political identities (Shimada 2014). Lacking evidence of intensive domestic occupation, the resident population of the Sicán Precinct appears to have been composed of a rather small number of elites, although excavations in the Great Plaza at the center of the site reveal an intensive and diverse set of activities took place here, including the construction and maintenance of buildings and large-scale food preparation, perhaps for funerary events and/or the provisioning of laborers (Matsumoto 2014; Shimada et al. 2015).

The activities documented at the Sicán Precinct were fueled by and dependent upon residential and production sites (e.g. ore mining and smelting complexes, ceramic and metal workshops) in the surrounding area, and archeometric analysis has linked vessels excavated from Huaca Loro to the Huaca Sialupe ceramic and metal workshop located some 23 kilometers to the southwest (Shimada et al. 2015). Shimada (in press) describes the network of interaction between the Precinct and its hinterland as a "dispersed city."

Although Middle Sicán burial practices, iconography, and material culture have been found beyond the north coast, there is no evidence to suggest that elites engaged in militaristic conquest or exerted much direct control over the local populations with which they interacted (Segura and Shimada 2014; Shimada 2014). The Precinct lacks defensive structures, no Middle Sicán "enclaves" have been discovered, militaristic iconography is rare, and while the practice of human sacrifice is well-documented, little evidence of skeletal trauma has been found in the human remains excavated thus far (Farnum 2002; Klaus 2008; Klaus et al. 2017; Shimada 2014). Rather it appears that Sicán influence was tied to religious ideology surrounding the Sicán Deity/Lord and motivated by economic interests to establish and maintain trade networks and access to human labor for the acquisition, production, and exchange of resources – including exotic or "sumptuary" goods (Shimada 2014; in press).

The Sicán Deity, a ubiquitous symbol in Middle Sicán art, is a supernatural being with avian-like characteristics including almond-shaped and upturned or "winged" eyes with dilated pupils, claws, wings, a tail, and feathers with a human-like face and/or body shape (Shimada and Samillán 2014). This figure is often found in association with images linked to life-giving water and fertility such as *spondylus princeps* – whose ritual significance is widespread throughout the Andes – and fauna present on the north coast during the rainy season (e.g frogs, iguanas, bees, and parrots; Shimada and Samillán 2014). The human embodiment of this being, known as the "Sicán Lord," shares the same basic physical characteristics, although supernatural elements like wings, feathers, and a tail are replaced by anthropomorphic traits and personal adornments including elaborate headdresses and clothing (Shimada and Samillán 2014). It is believed that this "Sicán Lord" symbolized the elite leaders of Middle Sicán society, who upon death would assume the identity of the Sicán Deity (Matsumoto 2014; Shimada and Samillán 2014). It is unclear whether this all-powerful being was a symbol of Naylamp, the legendary progenitor of the Lambayeque dynasty who, based on ethnohistorical accounts, arrived in the region from

coastal Ecuador, or a unique blending of the supreme masculine god common in Wari and Tiwanaku cosmology with north coast fisherfolk traditions (Shimada and Samillán 2014). In either case, symbolic associations between the Sicán Deity and Lord and Middle Sicán elites were likely important for naturalizing their authoritative role and privileged status (Shimada 2000; Shimada 2010; Shimada and Samillán 2014). Furthermore, the blending of familiar religious/ritual elements and variation in the themes associated with images of the Sican Deity and Lord (i.e. rainfall, aquatic and terrestrial flora and fauna) would have had broad appeal to the general populace composed primarily of fishing and agricultural groups, perhaps contributing to the spread of Middle Sicán influence.

Based on mitochondrial DNA, patterns of inherited dental traits, and extensive excavations of production and mortuary contexts, Shimada and his colleagues have identified at least two ethnic groups within Middle Sicán society (Cervantes et al. 2011; Klaus et al. 2017; Shimada 1994; Shimada et al. 2004; Shinoda 2014). Inferred elites were more closely related to one another than they were to the local populations living in settlements surrounding the Precinct, perhaps reflecting a distinct "Sicán" ethnic identity with possible origins in Ecuador (Corruccini and Shimada 2002; Shimada 2010; Shimada et al. 2004; Shinoda 2014). In contrast, patterns of inherited dental traits, burial practices, and material culture found for those interred in surrounding residential sectors and workshops or as sacrificial offerings associated with elite tombs aligned more closely with the earlier Moche culture, perhaps denoting the enduring presence of a "Mochica" ethnic identity (Cervantes et al. 2011; Klaus 2003; 2008; Klaus 2014; Klaus et al. 2017; Shimada 2010; Shimada et al. 2004). The persistence of an ethnic "Mochica substratum" (Klaus 2014) within Middle Sicán society as inferred from mortuary and production contexts (discussed further below) provides further support for the idea that local populations sharing religious and economic ties with elites likely enjoyed a significant degree of autonomy in their daily operations (Shimada 2015; Shimada and Craig 2013; Shimada and Wagner 2007). Shimada and his colleagues have argued that Middle Sicán elite strategies for control were focused primarily on aligning themselves with pre-existing religious and economic structures to develop relationships that could be seen as mutually beneficial (Klaus and Shimada 2016; Segura and Shimada 2014; Shimada 2014). This perhaps helps to explain the perceived willingness on the part of non-elites to participate in Middle Sicán society, despite what appears to have been pervasive social inequalities (Klaus et al. 2017; Shimada et al. 2004).

Burials excavated at the Sicán Precinct and surrounding areas provide the most striking evidence for the hierarchical organization of Middle Sicán society, and emphasize the role of religious ideology, and perhaps ethnicity, in legitimizing and maintaining the power structure. Observed mortuary patterns suggest the presence of at least four different social classes including high level elites, low level elites, commoners, and sub-commoners, with differential access to metals being a primary mechanism of distinction (Klaus et al. 2017; Shimada 2014; Shimada 2015; Shimada et al. 2000). Contrasts between inferred elites (regardless of ranking) and non-elites correspond to the suggested Sicán and Muchic ethnic distinctions outlined above, giving support to the idea that social status and ethnic identity were intertwined in Middle Sicán society (Klaus et al. 2017; Shimada 2014; Shimada et al. 2000).

At the very bottom of the social scale (i.e. commoners and sub-commoners), individuals were interred in flexed or extended positions in simple pits below the floors of humble domestic structures and workshops, or served as dedicatory and/or sacrificial offerings in Precinct plazas and elite tombs (Shimada, 2000). Ceramic associations, body positioning and orientation, as well as the kinds of grave goods (if any) associated with these people reflect Moche traditions. For example, at the Middle Sicán production sites of Huaca Sialupe and Batán Grande, as well as at the Precinct and other Sicán-affiliated locales, inferred commoners were often buried in extended position with the head facing south in simple pits, wrapped in plain cotton cloth, and accompanied by ceramic vessels, llama offerings, with copper objects sometimes placed in the mouth and/or hands (Farnum 2002; Klaus 2003; Klaus 2014; Klaus et al. 2017; Shimada et al. 2004). An example of this type of burial is provided in Figure 5. Ceramic vessel associations rarely exceeded ten offerings and were largely reflective of Moche traditions, although some goods depicting the Sicán Deity or Lord on characteristic burnished, blackware (or imitations thereof) have also been found (ibid). Interestingly, vessels bearing images of the Sicán Deity in inferred commoner burials near the Precinct largely derive from workshop contexts and are found with individuals in a seated or flexed position more reminiscent of the pattern documented for Sicán elites than of Moche practices, perhaps denoting close social ties with elite groups and/or status emulation (Klaus et al. 2017). Inferred dedicatory or sacrificial burials ("subcommoners") are those that lack any evidence of grave goods, demonstrate "atypical" body positioning (e.g. supine position), and/or osteological evidence of violent death (e.g. decapitation or strangulation; Eeckhout and Owens 2008). An example of this type of burial is provided in Figure 6. Individuals interred in commoner or sub-commoner type burials were also disproportionately burdened by dietary and developmental insults (Farnum, 2002; Shimada et al., 2004), supporting inferred relationships between mortuary treatment and position within the social hierarchy.

High- level elites were generally interred in seated positions at major Precinct temples in deep shaft tombs, and their burial regalia were rich with imagery of the Sicán Lord, the earthly alter ego of the Sicán Deity (Shimada, 2000). These individuals were frequently associated or

adorned with masks depicting a standardized image of the Sicán Lord's face made of gold, silver, and/or copper alloys (depending on rank) and decorated with other highly-valued or exotic items (Matsumoto 2014; Shimada et al. 2000; Shimada and Samillán 2014). These masks are interpreted as a symbolic expression of elite leaders' transformation from the Sicán Lord to the Sicán Deity upon death, and were perhaps part of the mechanisms used by elites to establish and legitimize their power (Matsumoto 2014; Shimada and Samillán 2014).

High elite shaft tombs contained large quantities (as much as 1.2 tons) of high karat gold, silver, and copper alloy objects, as well as many imported items such as cinnabar pigment, emeralds, amber beads, feathers, and *Spondylus princeps* and *Conus fergusoni* shells (Shimada et al., 2000). Other high elite items included painted cloth, semi-precious stones, double and single spouted bottles, and human retainers (Shimada et al., 2000; Shimada, 2010). Low-level elite burials share many features with the previous category, although ochre pigment was used rather than cinnabar, and neither high karat gold alloy objects nor emeralds were present (Shimada, 2010). An example of a Middle Sicán elite burial is provided in Figure 7.

Middle Sicán burials and material culture have also been found at Huaca Cao Viejo, a major Moche civic-ceremonial mound at the El Brujo Archaeological Complex in the Chicama valley about 230 km south of the Sicán "heartland". The Huaca Cao Viejo mound sits atop a bluff overlooking the Pacific Ocean at the right margin of the Chicama river and is composed of at least four superimposed structures built over a span of about 500 years (Mujica et al. 2007). Painted with polychrome reliefs and murals, the most well-known depicting warriors parading naked captives, Huaca Cao Viejo was a major focal point for Moche ritual and administrative activities in the region, including the burial of high-ranking elites, until its eventual abandonment and re-purposing as a burial ground by local populations some time around 650 CE (Franco et al. 2007; Mujica et al. 2007). The period between Moche abandonment and the appearance of Middle Sicán-style burials and material culture around 950 CE has been dubbed the "Transitional" phase, characterized by a ceramic style blending Moche and foreign (Wari?) elements (Mujica et al. 2007).

Sicán-affiliated internments at Cao Viejo are highly concentrated and superimposed on the northeast slope of the mound, and while many of them have been disturbed by extensive looting, 100 intact and partially-intact burials dating to the late Middle Sicán and Late Sicán periods (ca. 1000 – 1100 CE) have been excavated thus far (Franco et al. 2007; Segura and Shimada 2014). Individuals were bundled in plain cotton cloth, typically in a seated flexed position or with their legs crossed, and sometimes with the addition of supporting cane or wooden frames and "false heads" (Franco et al. 2007), a practice also seen in Ychsma burials (discussed further below). The sample is thought to be composed primarily of local commoners, although several burials demonstrate similarities with elite internments at the Precinct including the presence of retainers, copper arsenic funerary masks, shiny blackware bottles with images of the Sicán Deity, finely woven textiles with Sicán-style iconography, and *spondylus princeps* shells (Franco et al. 2007). Segura and Shimada (2014) infer that these individuals were local, low-ranking elites that had adopted Middle Sicán ideology.

Most of the Sicán-affiliated burials at Cao Viejo include a single individual interred in a simple pit and accompanied by camelid remains (particularly the head and/or legs), woven clothing, blankets, and bags, Moche-style ceramics and locally-made copies of Sicán blackware bottles, copper sheets, textile tools, and pieces of chalk (Franco et al. 2007). Interestingly, there does not seem to have been any spatial differentiation between inferred elite and commoner burials at Cao Viejo, with individuals of different status designations, adults, and children being

interred in close proximity to one another (Franco et al. 2007; Mujica et al. 2007).

The Ychsma archaeological culture is characterized by distinct ceramic, architectural, and burial styles distributed throughout the Lurín and Rimac valleys of Peru's central coast, particularly during the Late Intermediate Period (900 – 1470 CE), which as mentioned previously marks the period between the end of Wari influence ("Middle Horizon") and the rise of the Inca state ("Late Horizon;" Marsteller 2015; Uhle and Shimada 1991). Ychsma architecture includes ramped pyramids (discussed further below) and monumental platforms with adjacent patios (Eeckhout 2004a). Ychsma-style ceramics were produced using relatively simple technology, a limited range of colors (cream, black, white and red pastes), with simple stamped, incised, or appliqued decorations (Vallejo 2004). Although examples of Ychsma iconography are rare, common images found on ceramics vessels and figures, textiles, and wooden implements include geometric designs and stylized images of fish, mammals (e.g. camelid, dog or fox, monkeys, felines), snakes, plants such as maize, and anthropomorphic characters, often with prominent and hooked noses (Marsteller 2015; Vallejo 2004). There is no evidence that Ychsma communities utilized molds in ceramic production until the Inca intrusion into the central coast, and vessels demonstrate a high degree of variability with regards to morphological characteristics (i.e. specific dimensions of shape and size) and production strategies (i.e. the type of clay used, cleaning and firing techniques), likely reflecting the artistic preferences of the makers (Vallejo 2004).

In Spanish historical documents, Ychsma society is described as an ethnic lordship (*señorío*) consisting of *parcialidades* grouped into several chiefdoms (*curacazgos*) in the Lurín and Rimac Valleys, who were aligned to a common religious leader in residence at Pachacamac, the "creator god" in Quechua (an indigenous language in Peru) and progenitor of

the world, humans, and agriculture associated with the night and earthquakes (Marsteller 2015; Shimada 1991). Pachacamac was the site of a legendary oracle believed to have been made some time during the Middle Horizon, perhaps ca. 800 – 900 CE, and a major focus of religious/ritual activities (Shimada 1991). The oracle is a seven-foot tall carved wooden staff depicting two human figures back-to-back and adorned with images of maize and possibly yucca, as well as animal and arc designs. Shimada (1991) and Eeckhout (2008a) have argued these bear resemblance to "moon animal" and "sky serpent" motifs common on the north coast during the Middle Horizon. The idol was found with remnants of a door covered by cloth bearing aquatic motifs (e.g. spondylus and strombus shell, marine creatures) outside of the "Painted Temple," which is located in the center of Pachacamac's main religious/ritual and burial sector (the "Sacred Precinct;" Eeckhout 2013; Shimada 1991). Several competing interpretations of the idol have been developed based on historically-documented folklore, including the idea that it symbolized the god Pachacamac (and perhaps also his brother, Vichama, the god of the day) and his supernatural power, or that it reflected mythical and natural explanations for the origins of the cosmos (Shimada 1991). Regardless of the specific meaning behind the oracle, it has a largely "agrarian" theme (Shimada 1991).

During the Late Intermediate Period following the intrusion of Wari and Sicán religious and stylistic influences on the central coast, monumental architecture proliferated at the site of Pachacamac and surrounding areas, offering clues about the organization of Ychsma society. Most of these structures share a common set of features (ex. rectangular platform mounds each with a short, central access ramp, and a front patio surrounded by a wall with access from the outside) and are known as "Pyramids-with-Ramps", or PCRs (Eeckhout 1999; Eeckhout 2004a; Eeckhout 2004b; Eeckhout 2008b; Eeckhout 2013). Eeckhout's extensive excavation of the largest PCR in Pachacamac (PCR III) indicated the platform had restricted access and, although food remains were less numerous than in the plaza, the high proportion of charred camelid and deer remains suggests pyramid occupants had preferential access to high quality foods (Eeckhout 2004a; Eeckhout 2004b; Eeckhout 2008b; Eeckhout 2013). The enclosures surrounding the platform contained benches, niches, and possible windows, but yielded little traces of activity save for the deposition of elaborate burials, perhaps of high status individuals (Eeckhout 2004b). In contrast, the plaza was the site of extensive occupation and activity, yielding numerous hearths, food remains, and tools associated with weaving and cooking (Eeckhout 1999).

The occupation of PCR III lasted for about 30 to 40 years, and it was abandoned shortly after the deposition of an inferred elite burial (Eeckhout 1999). Based on this information, Eeckhout (1999) interprets PCRs as the homes of elite individuals who held periodic ceremonies for their subjects in the plazas. The abandonment of one PCR and subsequent construction and occupation of another is seen to reflect the transfer of power from one lord to their successor (Eeckhout 2004b). This interpretation of Ychsma contexts accords well with the *parcialidad* model outlined previously and discussed in more detail below. To summarize, Eeckhout and others have argued that Ychsma chiefdoms were organized into different land-owning kin groups (i.e. *parcialidades*) sharing labor, resource, and ritual obligations (Díaz and Vallejo 2005; Díaz 2004; Eeckhout 1999; Eeckhout 2004a; Eeckhout 2004b; Eeckhout 2008b). Tribute (i.e. agricultural products, herd animals, raw materials) was sent from the *parcialidades* in the provinces to Pachacamac, where it was exchanged and stored by elites to maintain their wealth and fuel public ritual events (Eeckhout 1999).

While archaeological data seem largely consistent with the characterization of Ychsma society as a confederation of ranked chiefdoms comparable to the *parcialidades* described in

historical documents, associated burial assemblages have yet to reflect the standard "hallmarks" commonly used to identify and distinguish between different status groups (Díaz and Vallejo 2005; Díaz 2015; Marsteller 2015; Owens and Eeckhout 2015; Shimada et al. 2010). For example, although early historical accounts identify silversmithing as an important craft on the central coast, archaeological evidence of this practice remain elusive, likely due in part to extensive looting and destruction of sites in this region (Díaz and Vallejo 2005; Díaz 2015; Owens and Eeckhout 2015). Therefore, access to metals, a useful indicator of social status in Sicán society, has limited applicability in Ychsma contexts. Current knowledge regarding Ychsma mortuary practices suggests symbols of status were likely expressed through different (often less durable) means such as textiles, wood implements, dedicatory offerings, or even the extent of mortuary care and maintenance (Díaz and Vallejo 2005; Eeckhout and Owens 2008; Owens and Eeckhout 2015; Shimada et al. 2007b; Shimada et al. 2010).

Ychsma funerary practices have been characterized based on cemetery assemblages from the sites of Armatambo and Rinconada Alta in the Rimac valley, as well as Pachacamac in the Lurin valley (Díaz and Vallejo 2005; Owens and Eeckhout 2015; Shimada et al. 2010; Takigami et al. 2014). Individuals were positioned in a seated, flexed position, bundled in layers of woven cotton and/or braided reeds with the addition of padding and/or filling made of plant materials to provide shape and support, particularly around the head (Díaz and Vallejo 2005). These bundles were often grouped into simple pits and accompanied by food offerings, ceramics, and craft tools, either arranged around the exterior of the bundle or tucked inside the various layers (Díaz and Vallejo 2005; Takigami et al. 2014). While most of the bundle wrappings were relatively simple and plain, textile layers closer to the body tended to be more colorful and elaborately designed (Díaz and Vallejo 2005). Women were often dressed in tunics while men wore *unkus* 



Figure 5 – Example of a non-elite burial at the Sicán Precinct; drawing by Izumi Shimada and César Samillán



Figure 6 - Example of a possible sacrificial burial at the Sicán Precinct; drawing by Izumi Shimada and César Samillán



Figure 7 – Example of an elite burial at the Sicán Precinct; drawing by Izumi Shimada and César Samillán

(ponchos) and loincloths. Bodies and clothing were sometimes stained with a red pigment, perhaps cinnabar, and adorned with simple shell or metal necklaces (Díaz and Vallejo 2005).

Díaz and Vallejo (2005) have recognized two main types of Ychsma funerary bundles including a "simple" type that took on the shape of the wrapped body, and those that were reinforced laterally with supportive cane or wooden frames, sometimes with the addition of decorated "false heads" (Shimada et al. 2007b; Shimada et al. 2010; Takigami et al. 2014). Presumably, the frames and false heads served to give the bundle the appearance of an upright, seated person and to provide structural support when being moved or transported (Shimada et al. 2010). While Díaz and Vallejo (2005) have suggested funerary bundles with supportive frames may have been reserved for people of high rank or elite status, the possibility that these features are a marker of age or the result of temporal change cannot be eliminated (Owens and Eeckhout 2015). For example, in comparing the Ychsma burial assemblages at Armatambo and Rinconata Alta, Marsteller (2015) found more cane-reinforced bundles and higher frequencies of ceramic vessels and other grave goods in the former. However, because grave goods (including "prestige" items such as metal objects and spondylus shells) become more numerous and funerary bundles become more elaborate over time at both sites, caution is required when using these characteristics as a basis for status inferences (Díaz and Vallejo 2005; Marsteller 2015).

Previous studies of the Ychsma funerary bundles included in my sample offer important models for how estimates of the time, effort, and energy expended to prepare and maintain funerary bundles – "tomb biographies", so to speak – may be helpful for identifying individuals with special social significance (Shimada et al. 2010; Takigami et al. 2014). For example, Takigami et al. (2014) found that the most elaborate of the 34 funerary bundles recovered (Bundle B; described in more detail in Chapter 6) from an adobe brick and stone-lined pit, also maintained a central position within the tomb despite evidence that bundles were added, removed, and rearranged on at least three occasions over the roughly five hundred years it was in use. This, combined with the tomb's location just outside the Painted Temple in Pachacamac's Sacred Precinct and its intrusion upon burials associated with an earlier, non-Ychsma occupation, suggest competition for space was high in this area (Shimada et al. 2010). Shimada et al (2010) and Takigami et al (2014) use this in addition to the elaborate design of Bundle B to argue the individual was afforded a significant amount of social prestige, at least in death.

## Labor Organization in Middle Sicán and Ychsma Societies

Models of pre-Hispanic labor organization on Peru's north and central coasts rely heavily on the *parcialidad* system described in early Spanish chronicles and colonial records. *Parcialidades* were territorially-bounded, lineage-based socio-economic groups headed by a lord or *principal* and ordered into a nested hierarchy under a central political and/or religious authority (Netherly 1977). These *parcialidades* provided the labor and/or goods necessary to sustain infrastructure and elite ideologies in exchange for access to corporately-held resources, particularly land and water (Netherly 1984). *Principales* were responsible for coordinating and supervising labor activities within their respective territories and serving as intermediaries between their subjects and higher-level leaders (Netherly 1977). They also filled an important role in religious life, and subjects' allegiance to their *principal* was contingent upon his ability to sustain communal well-being, not only during life but also in death as a venerated ancestor (Netherly 1977; Ramírez-Horton 1981).

According to Netherly (1984), the territory of each *parcialidad* was defined by irrigation canals, and placement within the network (i.e. whether located along primary, secondary, or tertiary canals) as well as access to other corporately-held resources, corresponded to the rank or status of the group. Parcialidades were hierarchically ordered according to the size of their labor force, as well as the kinds of goods and services they provided (Netherly 1977; Ramírez-Horton 1981). While most *parcialidades* were composed of farmers, other specialized groups identified in historical records included fisherfolk, potters, leather-workers, brick-layers, salt-makers, metalsmiths, chicha-makers (a fermented corn drink), carpenters, shoemakers, and traders (Netherly 1977; Rostworowski 1975). Rostworowski (1975) and Netherly (1977) have argued that *parcialidades* on the central and northern coasts were extremely specialized – so much so that they relied heavily upon exchange to fulfill basic needs. They conclude that the bulk of tasks completed to fulfill labor obligations to the regional polity, such as extracting and transporting resources to administrative centers, provisioning non-producing elites, or building and maintaining agricultural and administrative infrastructure, fell to those engaged in seasonal occupations such as farming and fishing (Netherly 1977; 1984; Rostworowski 1975). Artisan groups, and in some cases fishers, were thought to have worked on a more full-time basis, exchanging their goods in lieu of their labor to fulfill tax obligations and meet basic subsistence requirements. This interpretation is based on legal records of fishermen, artisans (e.g. potters, leather-workers, metalsmiths), and traders or merchants asking for exemption from Spanishimposed labor tax (mita) on the grounds that they did not own land or know how to farm.

The argument for full-time specialization is particularly strong for those involved in the production of regalia for elite consumers, such as potters, metalsmiths, and weavers, as well as the merchants or traders that acquired the necessary materials (Netherly 1977; Ramírez 2007;

2008; Ramírez-Horton 1981). These individuals appear to have not only been exempt from "menial" labor, but likely enjoyed some social privilege as a result of their close association with the top rungs of the social hierarchy (Netherly 1977; Ramírez 2007). Although Spanish records are somewhat ambiguous with regards to the social status of artisans and other specialists, the Inka practice of co-opting and re-settling skilled potters and metal smiths from occupied coastal areas suggests they were highly-valued members of their local communities (Netherly 1977; Ramírez 2007).

*Parcialidad* membership, at least during historical times, appears to have been not only a marker of occupational identity and a source of social prestige for some, but also of ethnic identity. *Parcialidades* were often described as distinct with regards to their style of dress, how and where they practiced religion, and even the dialects they spoke (Netherly 1977; Ramírez-Horton 1981; Rostworowski 1975). For example, ethnic differences between fisherfolk and the larger agrarian population are suggested by historical reference to '*la lengua pescadora*,' a distinct dialect thought to have been shared by fishing communities all along the Peruvian coast (Netherly 1977).

Ramírez (2007; 2008) has been more critical in her reading of historical documents, arguing that *parcialidades* may not have been as specialized or as territorially-bounded as conventionally assumed. She notes that historical records provide some indications that families often moved around to take advantage of both terrestrial and marine resources as weather and circumstance dictated, dividing their time between farming and/or fishing, craft production, and labor tax obligations. Marsteller (2015) and Marsteller et. al's (2017) bioarchaeological study of dietary practices at the Ychsma sites of Armatambo and Rinconata Alta (Rímac valley, Peruvian central coast), assumed to be fishing and farming *parcialidades* (respectively) based on their

locations, also bring the extent of specialization into question. Using stable isotope analyses of human remains excavated from the sites, Marsteller and colleagues' findings were largely consistent with the notion of specialized *parcialidades* in that the Armatambo individuals consumed more marine or higher-quality marine foods than those at Rincondad Alta, who in turn relied more heavily on terrestrial products (Marsteller 2015; Marsteller et al. 2017). Nevertheless, they notes some deviations from this general pattern to suggest a more complex situation was at play than offered by the above mentioned model (Marsteller 2015; Marsteller et al. 2017). Several "outliers" were identified, with some individuals at Armatambo consuming more terrestrial foods relative to the average for the site, and some at Rinconata Alta consuming more marine resources than expected (Marsteller 2015; Marsteller et al. 2017). Based on their data, Marsteller and colleagues conclude that there was either more internal specialization within each *parcialidad* than traditionally assumed, or alternatively that there was a greater degree of fluidity between groups (Marsteller 2015; Marsteller et al. 2017).

Previous studies on Peru's north and central coasts suggest *parcialidades* or comparable socio-economic systems dominated from at least the Middle Horizon (Cornejo 2000; Eeckhout 1999; 2008b; 2013; Klaus et al. 2017; Rengifo and Castillo 2015; Shimada 1994; 2001; Uceda and Rengifo 2006). Eeckhout's (1999; 2004b; 2008b; 2013) interpretation of the Ychsma "Pyramids-with-Ramps" using this model was discussed in the previous section. Unfortunately, little in the way of direct evidence regarding labor organization, such as workshops and other production contexts, is available for Ychsma society. This is not the case for Middle Sicán, and here I outline current models of labor derived from experimental studies and production site excavations in areas surrounding the Sicán Precinct.

Shimada and colleagues have described a "modular" approach for Middle Sicán construction, ceramic production, and metallurgy (Shimada and Craig 2013; Shimada and Wagner 2007). The modular approach is described as:

"Relatively small-scale production units sharing basic technology and achieving equivalent or similar outputs with a great deal of technical, artistic, and perhaps even political autonomy" (Shimada and Craig 2013:25,27).

All Precinct monuments were built using a chamber-fill technique that required a large-scale, unified construction effort (Cavallaro and Shimada 1988; Shimada 2000). Variations in the brick type, size, shape, and markings, as well as their distribution in Precinct structures suggest distinct groups were responsible for their production, transport, and ultimate use (Cavallaro and Shimada 1988). Almost all bricks were marked with a distinct design (e.g. lines, dots, hand and foot prints), and even bricks marked with the same or similar designs varied in terms of the materials used to make them, size, and shape (Bezúr 2003; Cavallaro and Shimada 1988). Furthermore, instead of entire segments of structures being built with similar bricks, they were distributed throughout the structures in a seemingly random pattern (Cavallaro and Shimada 1988). Cavallaro and Shimada (1988) and Shimada (2000) use these data to postulate that bricks were commissioned from various sources by groups or individuals who would then contribute them to elite administrators at the Precinct for construction projects, perhaps as a way to gain privileged access to the structures upon completion, or to assert social prestige.

Strategies used to construct monumental architecture provide strong evidence that labor was managed by the Middle Sicán elite lineages and involved significant task specialization, and Cleland and Shimada's (1998) identification of three distinct spheres of craft production provide a useful perspective on how tasks were divided among different social groups. According to Cleland and Shimada's (1998) research, sumptuary items (ex. fine ceramic and gold-alloy objects decorated with rare and/or imported materials and imagery of the Sicán Lord and Deity) appear to have been produced under the direct supervision of highly-specialized artisans and their apprentices in workshops directly associated with elite compounds. In contrast, more widely distributed crafts (i.e. copper alloy objects and mold-made ceramics) were produced in nucleated – but relatively autonomous – workshops, with laborers participating in all phases of production and sharing resources with relatively little over-sight by higher authorities (Cavallaro and Shimada 1988; Goldstein and Shimada 2007; Shimada 2000; Shimada and Craig 2013). Finally, production of paddle-and-anvil or *palateada* utilitarian ceramics appears to have been organized at the household level and performed in domestic workshops on a seasonal or part-time basis (Cleland and Shimada, 1998).

Excavations at Huaca Sialupe, a ceramic and metalworking complex just 22 km. from the Sicán Precinct associated with residential and burial contexts, offers the most striking evidence that labor was specialized and likely divided along status and/or ethnic lines (Goldstein and Shimada 2010; 2007; Klaus 2003; Klaus et al. 2017; Shimada and Craig 2013). The workshop produced both "high-status" and more broadly distributed goods, including mold-made ceramics (elite blackware bottles and utilitarian redware bowls), as well as thin copper-arsenic and gold alloy sheets, tools, and ornaments. Similar to the adobe bricks used to make Precinct structures, ceramic molds found in workshop rooms had "makers marks" (Shimada and Wagner 2007). Different markings were found on molds of nearly identical form and paste, and molds with the same or similar marking were in some cases clustered by room (Shimada and Wagner 2007). Based on this evidence, Shimada and Wagner (2007) suggest at least two distinct groups of artisans were working in close proximity to produce the same kinds of goods. Goldstein and Shimada's (2007) analysis of fuel remains from kilns and furnaces indicate that ceramic and

metal-working were conducted at the workshop in a complimentary fashion, with charcoal produced during ceramic firing being re-utilized for smelting purposes. Technological and stylistic details recall earlier Moche traditions and were highly variable, although the resulting goods were uniform with respect to product type and the raw materials used (Shimada and Wagner 2007). This suggests artisans at Huaca Sialupe maintained a significant degree of autonomy, and perhaps a distinct social identity (Shimada 2014; Shimada and Craig 2013; Shimada and Wagner 2007).

Refuse from domestic hearths at Huaca Sialupe and proximity to residential and burial grounds suggest the workshop was in use year-round (Goldstein and Shimada 2010). Food remains included terrestrial products (e.g. agricultural crops, llama, dog) supplemented by marine resources (e.g. fish and mollusk), and remnants of a canal system indicate those living in the area engaged in both farming and crafting activities (Goldstein and Shimada 2010). The burial ground included a combination of 15 adults and children of both sexes, and largely reflect local, Moche burial traditions (discussed in detail in the previous section of this chapter).

The technological skills required to produce classic Middle Sicán burnished blackware vessels and many of the metal objects found in elite tombs suggest the presence of highly-skilled artisans (Bezúr 2003; Shimada 2015). Middle Sicán metallurgy was rather labor-intensive, and the sustained high temperatures and drafts required for smelting – all achieved without the use of bellows – would have required substantial manpower (Shimada and Craig 2013). Based on excavations at the Cerro Blanco mine and associated smelting sites used during Middle Sicán times, ore was largely extracted by hand from deposits at or near the surface using simple stone tools (Shimada and Craig 2013). Although physically-taxing, mining and initial processing of mineral deposits would have required minimal skills and is thought to have been accomplished
by common laborers or apprentices (Bezúr 2003; Shimada and Craig 2013). Producing the characteristic Middle Sicán thin, uniform metal sheets and working them into intricate ornaments such as the elaborate headdresses and masks found in elite tombs at the Precinct, on the other hand, surely involved a great deal of specialized skill (Shimada and Griffín 2005). There is some evidence to support the idea that highly-skilled artisans involved in the production of elite goods may have been afforded some social prestige, either because they were drawn from the elite class or as a result of their technological and/or artistic abilities. For example, while most individuals buried in association with Middle Sicán production sites are interred in a manner consistent with inferred commoner status and a Mochica ethnic identity, a few burials at the Huaca Sialupe and Batán Grande workshops possess features found among inferred Middle Sicán elites, both within the Precinct and at sites outside the "core", including El Brujo (Franco et al. 2007; Klaus 2003; Klaus et al. 2017; Segura and Shimada 2014).

Given the aesthetic mastery and scale of Middle Sicán craft production, full-time artisans would be expected, and therefore the burden of raw material extraction, transport, and food production may not have been shared evenly. While evidence of llama husbandry (Cavallaro and Shimada 1988) suggest Sicán people were perhaps exempt from physically-demanding transportation tasks, non-elite fishers and farmers whose participation in craft production may have been intermittent were potentially incorporated under the elite lineages and perhaps largely responsible for construction activities and resource extraction, in addition to subsistence activities.

Klaus et al. (2009) and Klaus and Chang (2011) found increased prevalence of degenerative joint disease in burial samples from the north coast during the colonial era (ca. 1532 – 1600 CE) thought to reflect intensified labor demands under Spanish rule, but we still know very little about how the social division of labor outlined above contributed, if at all, to previously identified health inequalities between status groups during the Middle Sicán era. We know even less about the division of labor and degree of social hierarchy present in Ychsma society, and this study provides insights that will help to address these issues.

#### **CHAPTER 4**

## METHODS AND MATERIALS

My primary research question is whether non-elite members of Middle Sicán and Ychsma society were over-burdened with manual labor tasks relative to elites, as often assumed based on Marxist notions of social class. To address this issue, I focus on human skeletal remains and mortuary characteristics, testing the hypotheses that skeletal trauma, DJD, and ECs will be more frequent and more severe in non-elites. An important first step for these analyses is the ability to identify status groups based on mortuary treatment, and I begin this chapter with a discussion of the parameters considered to accomplish this. I then discuss each of my hypotheses and the statistical procedures used to test them, and describe the research sample.

## Methods for Determining Status

Sufficient archaeological data have been collected to outline the normal range of variation present in Sicán - and Ychsma - affiliated burials so that status-related differences in treatment are discernible. Mortuary information for sampled individuals was gathered from site reports, detailed burial drawings, artifact assemblages, and works published by various members of the Sicán, El Brujo, and Pachacamac Archaeological Projects. Social status designations were made only after osteological analyses were complete to avoid any potential biases that might influence data collection. Shimada's (2010) characterization of Sicán social classes is modified to accommodate the range of variation found within Ychsma burials, and a summary of the

criteria used to infer social status is presented in Table 1 and described in more detail in the following sections.

## Burial type, feature, and location

Characteristics considered in this section include burial type (i.e. whether multiple or single internment), feature (e.g. built tomb and earthen pit), and location (near monuments, in public plazas, nucleated workshops, or domestic contexts). Elite burials should reflect the greatest energy expenditure in terms of labor and materials, yielding characteristics with a relatively limited range of distribution (Tainter 1978). For Middle Sicán and Ychsma contexts, single or multiple internments (3 or more individuals) in large tombs associated with major monuments that include dedicatory or sacrificial individuals (criteria for identifying these cases is discussed below) fit these requirements and are therefore considered reliable indicators of high social status.

In contrast, non-elite burials should reflect the least amount of energy expenditure and present characteristics that are more widely distributed in Middle Sicán and Ychsma contexts. In terms of burial type, feature, and location, this includes single pit internments located in areas peripheral to major monuments where access was less exclusive (ex. public plazas, nucleated workshops, and domestic contexts), although they could also be present as "dedicatory offerings" or "sacrificial individuals" in elite tombs and/or the columns and walls of major monuments. Dedicatory and sacrificial individuals can be identified based on archaeological evidence of being bound, buried alive, strangled, or deposited in supine position, lack of personal grave associations, and/or osteological evidence of stabbing or decapitation (Klaus and Shimada 2016).

## Grave goods

Grave goods with limited range in Middle Sicán and Ychsma contexts that have been associated with elites throughout pre-Hispanic coastal Peru include religious iconography, various types of metal including high karat gold and silver, *tumbaga* (low karat gold-silvercopper alloy), and copper (Díaz and Vallejo 2005; Díaz 2015; Franco et al. 2007; Shimada 2010; Shimada et al. 2010). Other high-status objects include imported materials such as cinnabar pigment, amber, emeralds, feathers, *Spondylus princeps* and *Conus fergusoni* shells, and elaborately decorated textiles (ibid). Copper objects, floral and faunal remains, and craft tools have the widest distribution in Middle Sicán and Ychsma mortuary contexts, and are thus expected in both elite and non-elite burials.

## Intentional post-mortem alterations

Sustained contact between the living and the dead and the veneration of ancestors (real or mythical) are well-documented on the north and central coasts during the Late Intermediate Period, and ethnohistorical data point to the close association between those in leadership positions within the *parcialidad* system (discussed in detail in chapter 3) and the supernatural world (Eeckhout 2013; Eeckhout and Owens 2008; Klaus and Shimada 2016; Lopez-Hurtado 2015; Matsumoto 2014; Shimada et al. 2007b; Shimada et al. 2015). Leaders were believed to have literally transformed into powerful ancestors who would periodically be visited, fed, and cared for by the living in exchange for their protection of communal well-being (Netherly 1977). In Middle Sicán and Ychsma contexts, evidence of tomb re-visiting as well as intentional comingling and disarticulation of human remains has been found in burials considered high status based on type, feature, location, and grave good assemblage, supporting an association

between these characteristics. The death of an average citizen would likely not be associated with the same kind of social upheaval that can occur with the death of key political and/or religious figures (Metcalf and Huntington 1991), and therefore evidence of grave re-visiting and post mortem alterations is not expected in non-elite mortuary contexts.

# Methods for Determining Activity Patterns

To test the assumed relationship between low social status and manually-intensive activities and better understand how life experience varied for different status groups in Middle Sicán and Ychsma societies, several osseous changes were considered including post-cranial skeletal trauma (fractures and Schmorl's nodes), DJD in the shoulder, elbow, wrist, knee, and ankle, and ECs for fibrous and fibrocartilaginous insertions on the humerus, radius and ulna. Information about the specific muscle attachment sites considered in my EC analysis is presented in Table 2. These skeletal markers have been selected because they: 1) are associated with strenuous activities in the clinical and anthropological literature; 2) are commonly observed in archaeological populations; and; 3) can be scored according to clearly defined standards that are either widely used in bioarchaeology (fractures, Schmorl's nodes, and DJD) or in need of further testing (ECs).

Activity assessments include all adult individuals whose skeletons are complete enough to: 1) determine sex using pelvic morphology and/or cranial features (i.e. extent of cranial robusticity) as outlined by standard bioarchaeological methods (Buikstra and Ubelaker 1994); 2) determine age according to standards developed for the cranium (i.e. extent of suture closure), and/or pelvis (the auricular surface and/or pubic symphysis; Buikstra and Ubelaker 1994);

Mortuary Characteris	tics	Elite	Non-elite
	Single or multiple internments in tombs closely associated with major monuments	Х	
Burial type,	Multiple internments in tombs associated with public plazas	Х	X
structure, and	Sacrificial or dedicatory internment in tombs associated with monuments and/or		
location	public plazas		X
	Single pit internments in plazas, nucleated workshops, or domestic contexts		X
	Religious iconography	X	+
	High karat gold/silver	X	+
	Tumbaga (low karat gold-silver-copper alloy)	X	
Crows acada	Copper alloy	X	X
Grave goods	Elaborately decorated textiles	X	<u> </u>
	Imported items	X	
	Craft tools	X	X
	Floral/faunal remains	X	X
Intentional Post			1
mortem alterations		X	

Table 1 - Mortuary characteristics used to infer social status

3) evaluate several major joints for degenerative changes; 4) create size variables using measurements of the humerus (maximum length, vertical head diameter, and epicondylar breadth); and 5) evaluate both fibrous and fibrocartilaginous attachment sites of the upper limbs (humerus, radius, ulna). Prior to reaching full skeletal maturity, normal variations in muscle attachment morphology encompass characteristics also associated with physical strain (ex. rough surface texture, pitting), and therefore individuals with incomplete fusion of the upper and/or lower extremities were excluded from my sample.

A sexual division of labor is common in both traditional and contemporary societies, and it may therefore be warranted to investigate whether statistical associations between sex (male/female) and participation in strenuous labor activities are present in my sample. While not the focus of this research, a sexual division of labor may influence my ability to discern meaningful differences between status groups, and statistical testing of association between sex and skeletal markers of activity are included in the Results chapter.

## Hypothesis 1 - skeletal trauma

Based on the assumption that status groups in Middle Sicán and Ychsma societies differed with regards to engagement in manual labor tasks, I predict that there will be a significant association between inferred social status and evidence of skeletal trauma, with postcranial fractures and Schmorl's nodes being more prevalent among non-elites in my sample. I also explore whether any differences in fracture prevalence are discernible between males and females. Evidence of cranial fractures and other skeletal trauma often attributed to violent acts such as decapitation or stabbing is not considered here, as it has been discussed elsewhere for the Middle Sicán sample (Farnum 2002; Klaus and Shimada 2016) and was not observed in the Ychsma group.

Fractures and Schmorl's nodes were scored as either present or absent and the specific bone, side, and location affected were recorded, according to standard bioarchaeological protocol outlined in Buikstra and Ubelaker (1994). Statistical testing of the association between two dichotomous categorical variables, in this case status (elite/non-elite) or sex (male/female) and skeletal trauma (present/absent), is best achieved with the Chi-square procedure, which I performed using the Crosstabs function in SPSS 21. In circumstances where counts were low (i.e. less than five individuals in any category), results for Fisher's exact test were considered in place of Pearson's Chi-square test, as is the standard procedure (Howell 2007). The p-value was set at 0.05; the standard threshold used to reject the null hypothesis of no association between two categorical variables in social science research (Howell 2007). P-values extrapolated using Chi-square and Fisher's exact tests below 0.05 would indicate a statistically significant association between sex or status and skeletal trauma, supporting my hypothesis.

The chi-square statistic determines whether two variables are associated, but does not offer any information about the strength of relationships – which is known as the effect size. Effect size for associations between two dichotomous variables are determined using the phi statistic. Phi values that are 0.5 or greater denote a "strong" effect, while phi values between 0.3 and 0.5 are considered "medium" or "typical" effects, and those below 0.3 reflect a "small" effect (Morgan et al. 2007). Phi values were determined for each chi-square test employed, and are reported in the Results chapter along with statistically significant chi-square tests.

Prevalence rates of skeletal trauma were low in my sample as a whole (discussed further under the "Research Sample" heading), and therefore data for the individual bones affected were

67

Bone		Fibrous entheses	Fibrocartil	aginous entheses
	muscle	location	muscle	location
	Deltoideus	Deltoid tuberosity		
Humerus	Pectoralis major	Lateral lip intertubercular groove		
	Teres major	Medial lip; intertubercular groove		
	Latissimus dorsi	Floor of the intertubercular groove		
Radius		Medial portion of lateral radial	Biceps brachii	Radial tuberosity
	Pronator teres	shaft	Brachioradialis	Styloid process
Ulna			Triceps brachii	Olecranon process
			Brachialis	Coronoid process

Table 2 - Entheses evaluated by bone and type

combined under three broad categories for statistical testing. The upper limb category includes data for the clavicle, humerus, and radius. The torso category includes the ribs and vertebrae, and the tibia is the only lower limb bone for which trauma was observed.

## Hypothesis 2 – DJD

If non-elites engaged in physically-demanding tasks on a more regular basis than elites, it is predicted that degenerative joint disease (DJD) will be more prevalent and more severe in this group. I test this hypothesis by searching for statistically significant associations between social status and the presence of DJD in the shoulder, wrist, knee, hip, and ankle joints, as well as the facet joints of the cervical, thoracic, and lumbar spines. Osteophytosis, the development of osteophytes at the margins of vertebral bodies, was omitted from consideration due to its tenuous relationship with physical activity (Jurmain et al. 2013; Weiss and Jurmain 2007).

While Buikstra and Ubelaker's (1994) standards provide ordinal scoring criteria to measure DJD severity, this was based on the premise that all bony traits associated with DJD are equally indicative of the condition, which we now know not to be the case. I therefore also scored DJD as a dichotomous variable (present/absent), which is the most widely adopted approach in bioarchaeological studies (Cheverko and Hubbe 2017; Henderson and Nikita 2016). Because eburnation is the only definitive indicator of DJD but can be difficult to identify in archaeological bone, I opted for a conservative approach that scores DJD as present only when two or more skeletal signatures (i.e. lipping at the margins and porosity on the surface of joint articulations) and/or eburnation were present, as recommended by Cheverko and Hubbe (2017); Rogers and Waldron (1995); Rothschild (1997). The same statistical procedures used for Hypothesis 1, Pearson's Chi-square, Fisher's exact, and the phi tests were also employed to test for associations between sex or status and DJD, as well as to determine the strength of relationship for statistically significant results. However, because DJD is an age-related process, comparing prevalence rates between status groups and the sexes without first partitioning the samples into age categories can lead to spurious results. Therefore, I partitioned the samples into two age groups: 1) under 30 years; and 2) 30 years and older, following the recommendations of Cheverko and Hubbe (2017). This accords well with clinical studies that find the effects of age on DJD expression increase significantly following the third decade of life (Cheverko and Hubbe 2017; Felson et al. 2000; Weiss and Jurmain 2007), and will ensure that adequate sample sizes are maintained even after partitioning individuals by age.

## Hypothesis 3 - ECs

If the labor activities performed by non-elites were more physically-demanding than those performed by elites, aggregate enthesial (EC) scores should also be significantly higher in the former group, even after accounting for variations related to age and size. This hypothesis is tested using a non-parametric version of ANCOVA, the preferred procedure for determining whether two or more groups – in this case inferred elites and non-elites – differ significantly in a response variable (i.e. aggregated EC scores) while accounting for additional contributing factors ("covariates"; i.e. age and size; Howell 2007). For EC analyses, I did not include sex as a covariate, as previous studies suggest sex-associated influences are largely related to size (Vilotte et al 2010; Weiss 2015), which is considered. Instead, the sample was divided into male and female groups so that ANCOVA could be employed to test for significant differences in aggregate EC scores between the sexes as a separate analysis.

Table 2 shows the muscle sites evaluated in this study by bone and enthesial type, and EC scores for every individual included in this sample are provided in Appendices C - F. ECs were not recorded for the lower limbs because body weight – a measure difficult to estimate and account for with archaeological human remains – is a particularly strong influence on these sites, and previous bioarchaeological studies have had little success in discerning differences in activity patterns using EC data from the lower limbs (Churchill and Morris 1998; Weiss, 2004).

ECs were scored using a four-category ordinal scale where a score of zero indicates no expression, and score of one indicates faint expression, a score of two indicates moderate expression, and a score of three indicates strong expression. Criteria for each category were adapted from the methods outlined by Hawkey (1988), Hawkey and Merbs (1995) for fibrous entheses and Villotte (2006) and Villotte et al. (2010) for fibrocartilaginous entheses. The Coimbra method was not yet finalized or published at the time of data collection, but does not vary significantly from the Villotte (2006) and Villotte et al. (2010) schemes used here. My scoring criteria is as follows: a) 0 = no expression; slightly rounded cortical surface (fibrous), smooth, well-defined imprint with regular margins (fibrocartilaginous), and an absence of surface irregularities (ridges, crests, body outgrowths); b) 1 = faint expression; irregular cortical surface (slight indentations; fibrous), irregular margins (fibrocartilaginous); c) 2= moderate expression; uneven cortical surface with a distinct ridge (fibrous), irregular margins accompanied by surface irregularities (pitting, bony outgrowths) covering less than half of the insertion surface (fibrocartilaginous); and d) 3= strong expression; distinct ridges and crests separated by a slight indentation (fibrous), clearly delineated and irregular margins accompanied by surface pitting or bony outgrowths covering more than half of the insertion surface (fibrocartilaginous). Figure 8

demonstrates "strong" ECs (score of 3) at the fibrous attachment for the deltoideus muscle on the humerus and the fibrocartilaginous attachment for the biceps brachii on the radius, respectively. Note the distinct ridges present for the fibrous insertion, and the irregularity and pitting at the surface of the fibrocartilaginous insertion.

There are two general data treatment approaches used to handle ordinal response data such as EC scores. One approach collapses the ordinal information into a binary category, while the other assigns numerical values to the ordinal scores and treats them as a continuous variable (Agresti 2010). The latter approach will be used for this research, and EC scores will be aggregated by summing ordinal scores for the left and right side, so that each individual is represented by a single value that reflects their overall level of biomechanical stress and strain. This is guided by the premise that the labor activities of status groups differed primarily in terms of physical requirements, and also acknowledges that associations between EC scores and specific activities are extremely tenuous. Treating ordinal EC scores as a continuous variable ensures no data are lost by collapsing categories into a dichotomy and also provides more statistical power (Stiger et al. 1998). The continuous approach is generally preferred when one can reasonably assume a latent continuous variable underlying the ordinal categories (Agresti 2010). This assumption is acceptable in circumstances where: 1) the range of possible responses actually lies on a continuum from one ordinal category to the next; and 2) the ordinal variable could be measured on a continuous scale if sufficient measurement instruments were available (Stiger et al. 1998). ECs conform well to these assumptions. Muscle attachment sites are difficult to measure because they are three-dimensional, and in some cases, it is not possible to pinpoint their borders precisely. Nevertheless, numerical scoring of ECs is not arbitrary in that



Figure 8 - "Strong" fibrous EC on the humerus (left) and "strong" fibrocartilaginous EC on the radius (right)

larger numbers reflect increasing size and complexity, and actual observations – if measured precisely – would vary along a continuum from one category to the next.

For statistical tests involving ECs, age will be scored for each individual using the average of their estimated age range. For example, if based on skeletal aging techniques, an individual is classified between the ages of 35-50 years, their average age would be (35 + 50)/2 = 42.5 years. The size variable was calculated by summing the z-scores of humeral measurements including maximum length, vertical head diameter, and epicondylar breadth, as recommended by Weiss (2003; 2007). Z-scores denote how many standard deviations away from the sample mean a particular observation is, with positive values denoting cases above the mean, negative values denoting cases below the mean, and 0 denoting cases that are at the mean (Howell 2007). In this study, the combined z-scores provide a comparative index of humeral size for each individual included in my sample, and this is crucial for testing the hypothesis that elites and non-elites will vary with regards to aggregate EC score beyond what can be explained by differences in size.

Although the lower limbs (particularly the pelvis and femur) may yield better estimates of overall body size due to their weight-bearing responsibilities, humeral dimensions were chosen here because previous studies have shown them to correlate strongly with upper limb EC scores, and they are the most commonly used proxies for size in EC analyses (Michopoulou et al. 2015; Niinimäki 2011; Weiss 2007; 2015). Additionally, focusing on humeral measurements allowed me to increase sample size and include individuals whose lower limbs were either fragmentary or incomplete.

Cheverko and Hubbe (2017) have recently highlighted ANCOVA's potential value for bioarchaeological studies of age-related skeletal traits like the ones studied here because it provides a way to address the age bias without having to partition the sample into smaller agespecific groups. In datasets like mine, where there are: 1) small sample sizes; 2) non-normally distributed data; 3) unbalanced groups and; 4) unequal variance between groups, non-parametricalternatives outperform standard ANCOVA and are preferred (Rheinheimer and Penfield 2001). Quade's (1966) ranked ANCOVA works particularly well when parametric assumptions are violated, especially if there is a positive relationship between group size and variance (i.e. the largest group should also be the most variable with respect to the covariates; Howell 2007; Rheinheimer and Penfield 2001). As Table 31 in Chapter 5 shows, my sample meets these conditions because the largest group (non-elites) also has the greatest variance with respect to size and aggregate EC scores.

Quade's (1966) ranked ANCOVA follows the same procedures of combining linear regression and ANOVA as parametric ANCOVA, except the ranked variables are used in place of the raw data. For this analysis, average age and size are first ranked and then regressed against ranked aggregate EC scores, ignoring the grouping variable (status). The unstandardized residuals were saved as a new variable, REC. A REC score was calculated for each individual, representing the amount of their aggregate EC score adjusted for confounding influences. Finally, a one-way ANOVA test was performed using the unstandardized REC residuals as the dependent variable, and the grouping variable (status or sex) as the factor. The resulting F-test determines whether there is a significant difference in aggregate EC scores between inferred elites and commoners, or between males and females, beyond what has been accounted for by variations in age and size. The threshold for rejecting the null hypothesis of no significant difference between groups was set at p = 0.05. Effect size measures the strength of the relationship between the variables, and is determined from the Eta test statistic. Standard thresholds used in social science

research to determine whether the effect is "small" (eta = 0.14), "medium" (eta = 0.36), or large (eta = 0.51) were employed. (Howell 2007).

## Hypothesis 4 - ECs

The final research question to be addressed in this analysis is whether fibrocartilaginous entheses are less prone to systemic influences such as age and size and thus better markers of activity than the fibrous type. Based on previous research done by Villotte (2006), Villotte et al. (2010) and Weiss (2015), I hypothesize that statistical correlations between EC scores, age, and size will be positive and stronger for fibrous entheses than for fibrocartilaginous entheses. This hypothesis is tested using the Spearman's rho statistic, which is a non-parametric version of the Pearson's correlation recommended when the data are not normally-distributed, which is the case with my dataset. Because previous studies have found statistically significant and positive correlations between EC scores, size, and age, I employed a one-tailed Spearman's rho test, as recommended for directional hypotheses (Howell 2007). The value of Spearman's rho is equivalent to the correlation coefficient (r), which provides an indication of effect size, or the strength of the relationship between the variables. I employed Cohen's (1988) thresholds to evaluate the effect size, as is standard for the social sciences. A correlation coefficient at or below 0.10 is considered a "small" effect; values from 0.30 to 0.4 are considered "medium" or "typical;" and those 0.5 and over are considered "large" or "larger than typical" (Morgan et al. 2007).

Site			Se	X	Total	
			Female	Male		
	Status	Non-elite	9	13	22	
El Brujo		Elite	1	1	2	
	Total		10	14	24	
	Status	Non-elite	4	7	11	
Pachacamac	Status	Elite	2	3	5	
	Total		6	10	16	
	Status	Non-elite	2	5	7	
Sicán Precinct	Status	Elite	17	3	20	
	Total		19	8	27	

Table 3 - Sample sizes by site, status, and sex

Site	Status	Sex	Ν	Min	Max	Mean	Std. Deviation
	Non alita	Female	9	21.00	55.00	34.3333	10.33501
El Bruio	Non-ente	Male	13	20.50	50.00	32.3846	8.58330
El Blujo	Flita	Female	1	35.00	35.00	35.0000	
	Ente	Male	1	30.00	30.00	30.0000	
	Non alita	Female	4	19.50	47.00	30.1250	13.28768
Dachacamac	Non-ente	Male	7	27.50	44.50	35.7857	7.56008
1 actiacatitae	Flita	Female	2	40.00	57.00	48.5000	12.02082
	Line	Male	3	27.00	47.00	35.5000	10.33199
	Non alita	Female	2	17.50	23.50	20.5000	4.24264
Sicán Precinct	Non-ente	Male	5	22.00	45.00	34.9000	11.72817
Sical Freehict	Flita	Female	17	18.50	40.00	23.7353	5.39999
	Diffe	Male	3	25.00	42.50	32.5000	9.01388

Tab	le 4	· - I	Descri	iptive	statistics	for a	age by	v site,	status,	and	sex
							<u> </u>		,		

## **Research Sample**

Relationships between social status and physical activity are investigated using burial samples from three pre-Hispanic Peruvian archaeological sites including: 1) the Sicán Precinct in Pomac (La Leche Valley, north coast), Middle Sicán's religious capital; 2) El Brujo in Magdalena de Cao overlooking the mouth of the Chicama River (Chicama Valley, north coast), a religious/ceremonial center established by the local Moche polity and later reutilized as a major burial ground; and 3) the cemetery in front of the Painted Temple at Pachacamac, the major religious and pilgrimage center in pre-Hispanic Peru (Lurín Valley, central coast). Included burials date from the Terminal Middle Horizon to the Late Intermediate Period (c. 900-1400 CE; Shimada et al., 2004, Shimada et al., 2006), and are associated with the Sicán and Provincial Sicán (Sicán Precinct and El Brujo), or Ychsma (Pachacamac) archaeological cultures. The sample includes 67 males and females ranging in age from 19-55+ years. Complete inventories of the individuals comprising the non-elite and elite groups considered here are provided in Appendices A and B and the composition of the sample with regards to site, status, sex, and average age are presented in Tables 3 and 4.

This sample was selected to obtain a sufficient number of elites and non-elites of both sexes, and to reflect the age range at which full integration into the labor system would be most likely (i.e. young adulthood to late adulthood). Although the sample can be considered representative in that individuals from the different status groups, sexes, and ages are included, females under the age of 30 years make up seventy percent of the entire elite group from the Sicán Precinct. This may reflect preferential burial for elite females in the West Tomb of Huaca Loro, where the bulk of my elite sample comes from. Their young age is a bit more difficult to explain, and it is unclear whether this reflects sex-based differences in life expectancy for Middle Sicán elites, selective burial practices at the Precinct, or some other factor.

## **CHAPTER 5**

## RESULTS

#### Social Status

Status inferences for the Sicán sample were drawn from previous research (Farnum 2002; Franco and Gálvez 2014; Franco et al. 2007; Klaus 2003; Shimada et al. 2007a; Shimada et al. 2004) and the primary criteria used were discussed Chapter 4 (refer to Table 1). This portion of the sample includes individuals from the Huaca Loro, Las Ventanas and Lercanlech [also known as Rodillona] Precinct monumental mounds, a sacrificial individual bound to a wooden pole atop the North Platform of Huaca Loro, as well as people from the Huaca Sialupe workshop and associated burial ground. All but the platform sacrifice, two inferred sacrificial burials incorporated into columns at Huaca Lercanlech, and four individuals buried in and around the Huaca Sialupe workshop are inferred elites, with the majority of Sicán "commoners" coming from Cao Viejo at El Brujo. While all of the Cao Viejo individuals in this sample were thought to have been buried alone in simple pits without precious metals or imported items, looting and the superimposed and concentrated burial context may have in some cases led to misidentification using criteria such as grave goods and burial type.

The inclusion of the potential sacrificial burials in the non-elite category also deserves further consideration. Possible sacrificial burials did not contain any of the characteristics associated with inferred elites, but I cannot rule out the possibility that these burials lacked "elite" characteristics due to the circumstances surrounding their death, and not because of their position in the social hierarchy. Nevertheless, available archaeological and osteological data do seem to support the inclusion of potential sacrificial victims in the non-elite group. First of all, there is no evidence to suggest that Middle Sicán elites engaged in the kind of militaristic endeavors that would involve the capture and sacrifice of foreign warriors and/or dignitaries, as has been documented on the north coast during earlier Moche times (Toyne et al. 2014; Sutter and Verano 2007). Additionally, Farnum (2002) argued that in terms of skeletal health, many of the inferred sacrifices at the Sicán Precinct were more similar to non-elites derived from less questionable contexts than they were to elites.

The Ychsma individuals come exclusively from a stone and adobe- walled tomb (Tomb 1-2) in Cemetery 1 right outside the Painted Temple in Pachacamac's Sacred Precinct and just a few meters north of the area known during Inka times as Pilgrim's Plaza (Shimada et al. 2010). Social status was inferred using a combination of previous work (Shimada et al. 2007b; Shimada et al. 2006a; Shimada et al. 2010; Takigami et al. 2014) and my own review of bundle inventories. As mentioned already, the general absence of metals in Ychsma burials (both in and outside of Pachacamac) necessitates a slightly different strategy than that used to infer social status in Middle Sicán contexts, and I outline my approach below.

Based on archaeological excavations in the Pilgrims Plaza, Shimada et al. (2006a); Shimada et al. (2010) argued that during Ychsma occupation, the site was an area of intensive, seasonal ritual activities for common farmers and fisherfolk from littoral and inland regions of the central coast. Food remains found at the site (maíz, beans, potato, yucca, squash, llama, cuy [guinea pig], deer, fish, and seal) were abundant, although signs of processing and consumption were rare. Many of the excavated contexts included large numbers of fly pupae casings, suggesting the food remains served as ritual offerings rather than food for the living (Shimada et al. 2010). In addition to food offerings, there were also weaving and metalworking tools, as well as evidence suggesting funerary bundles were prepared and/or repaired on site. This included sunken enclosures with decaying cloth, bundle filling (unspun cotton), and other plant materials, the small bones of human fingers and toes, as well as weaving tools, scraps of copper sheets, and shells containing cinnabar (Shimada et al. 2007b; Shimada et al. 2010).

Most of the Ychsma funerary bundles considered here consisted of a single, flexed individual (i.e. knees drawn up to the chest) wrapped in several layers of plain cloth and accompanied by food items (e.g. corncobs, cuy, and fish remains), as well as tools for weaving and fishing such as needles, spindles, nets, and floats. This is consistent with Shimada et al. (2006a) and Shimada et al.'s (2010) characterization of the nearby Pilgrim's Plaza as a ritual gathering area for humble fishers and farmers, and similar burial patterns have been documented at the Ychsma sites of Armatambo and Rinconata Alta (Díaz and Vallejo 2005). While bundles of this type make up the majority of the sample, "atypical" patterning with respect to grave goods and funerary processing techniques are apparent in some, perhaps reflecting high rank or status. In fact, there seems to be an association between evidence of intentional commingling, disarticulation, and the presence of grave goods recognized as prestige items throughout pre-Hispanic coastal Peru. Certainly the proximity of Tomb 1-2 to the Painted Temple, a site imbued with great sacred significance, would have made it a highly desirable resting place, an assertion supported by the density of burials and remnants of ritual activities in the area (Shimada et al. 2006a; Shimada et al. 2010; Takigami et al. 2014). In fact, Owens and Eeckhout (2015) noted a direct association between cemetery crowding, tomb disturbance, and proximity to Pachacamac's Sacred Precinct based on their excavations in a portion of Cemetery 1 just outside the walls of this sector.

Tomb 1-2 was close to the surface and easily accessible during the time it was in use, pupae casing were found in and around bundles, and evidence that bundles had been removed, rearranged, and added over time suggests periodic visits by the living (Shimada et al. 2006a; Shimada et al. 2010; Takigami et al. 2014). For example, when comparing placement within the tomb and radiocarbon dates for the bundles, expectations that the oldest bundles would be placed in the innermost portions of the tomb were not met (Takigami et al. 2014). Rather, the patterns suggest that on at least three occasions, existing bundles were re-arranged to accommodate new additions (Takigami et al. 2014). Additional evidence suggesting tomb re-visiting include basins found just below the cemetery surface and in association with the bundles that may have been receptacles for ritual offerings (Shimada et al. 2006a). It should be noted, however, that genetic and isotopic studies indicate individuals in Tomb 1-2 originated from perhaps three distinct regions of central Peru, including both coastal and highland areas (Takigami et al. 2014). It is therefore unlikely they reflect a single group of venerated ancestors.

All but one of the Tomb 1-2 individuals with "prestige" items not widely distributed in Ychsma mortuary contexts, including finely woven and brightly colored textiles, exotic feathers, *Spondylus* shell, and cinnabar, were bundled with remains from other individuals and/or were missing their heads. Neither Díaz (2015) nor Marsteller (2015) noted any evidence of intentional comingling or disarticulation in hundreds of Ychsma funerary bundles from Armatambo and Rinconata Alta, so this is perhaps exceptional. Aside from missing crania, these individuals are remarkably complete, with even the small bones of the hands and feet and some hair, skin, and cartilage still present. In contrast, the duplicate skeletal elements were isolated bones completely devoid of any tissues, typically placed in the innermost bundle layer in direct association with the body. This suggests co-mingling and disarticulation were intentional, and likely occurred during the bundling of relatively "fresh" corpses. Although small sample size limits my ability to draw any conclusions about the expression of status identity in Ychsma funerary contexts more broadly, I do believe that the combination of intentional comingling, disarticulation, and grave good "wealth" observed in some individuals has significance with regards to their social identities, especially given that these practices seem "atypical" compared to patterns described in other Ychsma contexts (e.g. Díaz and Vallejo 2005; Díaz 2015; Marsteller 2015; Owens and Eeckhout 2015) It would not be surprising to find elite individuals in Tomb 1-2, given its close proximity to the Painted Temple, evidence that it was likely re-visited and re-arranged on several occasions, and the density of ritual activity in the area. Based on the criteria outlined above, I identified five "elites" within the Tomb 1 - 2 sample, and they are discussed in more detail below.

Four of the five inferred elites in the Ychsma sample (Bundles H, N, Q, and W) demonstrate evidence of comingling, elaborate funerary processing, and/or disarticulation of the head prior to or after bundling without any associated vertebral cutmarks that would suggest decapitation. In contrast, there was no evidence of intentional comingling and only one case of cranial disarticulation in those inferred as commoners, based on their bundle assemblages. A duplicate right scapula was present in Bundle H, which contained a male aged 21 - 34 years with metal fragments, maíz, reed tubes, a ball of yarn, textile tools (needles and whorls), and shell beads that the excavator identified as *Spondylus*. Bundle N, another male aged 27 - 34 years, was headless and wrapped with an elaborately colored and decorated textile (Figure 9), as well as fishing tools (nets and floats), botanical remains, fish remains, and wooden tablets. Bundle W, a



Figure 9 - Decorated textile associated with inferred elite Ychsma Bundle N



Figure 10 – Inferred elite Ychsma Bundle B; photo by Izumi Shimada. Original source: Takigami et al. 2014. Assessing the chronology and rewrapping of funerary bundles at the prehispanic religious center of Pachacamac, Peru. Latin American Antiquity 25(3): 322-343

female aged 50 – 61 years was also missing her head, an extra sacrum was placed in correct anatomical position on her pelvis, and her pelvic cavity had apparently been filled with unspun cotton. Her grave associations included decorated textiles (camelid and cotton) and metal sheets as well as weaving tools in addition to corncobs and faunal remains. With Bundle Q, a headless female aged 35 - 44 years, there was evidence that one of her decorated bundle wrappings had been cut, possibly to remove her head after initial bundling (Takigami et al. 2014). The cutting of textiles has rarely been documented in pre-Hispanic Andean contexts (Takigami et al. 2014), and may therefore denote special or "unusual" social circumstances. Although Bundle Q lacked a head, her bundle was fastened with a false head, and additional grave goods included metal sheets (perhaps silver), ceramic fragments, and textile tools.

Bundle B (Figure 10), the largest of the group and the most elaborate Ychsma bundle described to date, was the only inferred elite without evidence of intentional comingling or disarticulation. This bundle included a man aged 40 - 49 years, a wooden cinnabar-painted false head, a headdress decorated with feathers likely from an Amazonian macaw, and elaborate textiles (Takigami et al. 2014). He was also dressed in a tunic made of camelid fiber, and strips of cloth were stitched onto the outer layer of his bundle.

## Patterns of Physical Activity

#### Hypothesis 1 - skeletal trauma

There were low frequencies of skeletal trauma in my sample as a whole, and data for the bones of the upper limbs, torso, and lower limbs are reported by status and sex in Tables 5 - 10. Bones for which no evidence of skeletal trauma was present in my sample are excluded from these tables. Figures 10, 11, and 12 offer a visual comparison of skeletal trauma frequencies by body region for the three archaeological sites considered, males and females, and elites and nonelites, respectively. The torso was the most common site of skeletal trauma for all groups considered. Torso trauma included Schmorl's nodes, vertebral compression fractures and rib fractures – lesions that are generally attributed to accidental injury rather than interpersonal violence. Skeletal injuries in the Middle Sicán sample attributed to violence, largely associated with human sacrifice, have been discussed at length elsewhere (Farnum 2002; Klaus and Shimada 2016) and are not considered here.

My data do not support the hypothesis that skeletal trauma was more common in nonelites, and no statistically significant associations were found between inferred status and trauma of the upper limbs, torso, or lower limbs (Tables 11, 12, and 13). No discernible differences in skeletal trauma were found between the sexes, either (Tables 14, 15, and 16). Failure to detect any statistically significant differences between the sexes and status groups is likely due to the low frequencies of skeletal trauma in my sample as a whole.

#### Hypothesis 2 – DJD

DJD was not prevalent in my sample. Tables 17 through 22 display frequencies of DJD in the joints of the upper limbs, spine, and lower limbs by age group and status, as well as by age group and sex. When DJD was present, it was most common in the spine, particularly in the cervical region (Tables 18 and 21). Graphic comparisons of DJD frequencies by site, sex, and status for each body segment and age group are presented in Figures 14 through 19.

I hypothesized that DJD would be more prevalent in the upper limbs, spine, and lower

			Clavicle				Humerus	5		Radius	
			Absent	Present	Total	Absent	Present	Total	Absent	Present	Total
	Non-	Count	38	2	40	39	1	40	39	1	40
<b>G</b> 4 4	elite	% within Status	95.0%	5.0%	100.0%	97.5%	2.5%	100.0%	97.5%	2.5%	100.0%
Status	Elita	Count	26	1	27	26	1	27	26	1	27
	Ente	% within Status	96.3%	3.7%	100.0%	96.3%	3.7%	100.0%	96.3%	3.7%	100.0%
Total		Count	64	3	67	65	2	67	65	2	67
		% within Status	95.5%	4.5%	100.0%	97.0%	3.0%	100.0%	97%	3.0%	100.0%

Table 5	Engange			1:1.	+	1	1	A	atataa
Table 5 -	Frequer	icies oi	upper	ump	trauma	DV -	none a	na	status
1 4010 0			• pp • •			~ )	00110 0		

Table 6 - Frequencies of torso trauma by bone and status

			Cervical			Thoracic		Lumbar			Rib		
		Absent	Present	Total	Absent	Present	Total	Absent	Present	Total	Absent	Present	Total
Neu	Count	38	2	40	33	7	40	38	2	40	36	4	40
Non-	% within	95.0%	5.0%	100.0%	82.5%	17.5%	100.0%	95.0%	5.0%	100.0%	90.0%	10.0%	100.0%
ente	Status												
	Count	27	0	27	25	2	27	27	0	27	26	1	27
Elite	% within	100.0	0.0%	100.0%	92.6%	7.4%	100.0%	100.0%	0.0%	100.0%	96.3%	3.7%	100.0%
	Status	%											
	Count	65	2	67	58	9	67	65	2	67	62	5	67
Total	% within	97.0%	3.0%	100.0%	86.6%	13.4%	100.0%	97.0%	3.0%	100.0%	92.5%	7.5%	100.0%
	Status												

				Tibia	
			Present	Total	
	NT	Count	39	1	40
<b>C</b> + +	Non-elite	% within Status	97.5%	2.5%	100.0%
Status	E1:4-	Count	27	0	27
	Elite	% within Status	100.0%	0.0%	100.0%
T-4-1		Count	66	1	67
Total		% within Status	98.5%	1.5%	100.0%

# Table 7 - Frequencies of lower limb (tibia) trauma by status

 Table 8 - Frequencies of upper limb trauma by bone and sex

				Clavicle			Humerus		Radius			
			Absent	Present	Total	Absent	Present	Total	Absent	Present	Total	
	Famala	Count	34	1	35	33	2	35	34	1	35	
Carr	Female	% within Sex	97.1%	2.9%	100.0%	94.3%	5.7%	100.0%	97.1%	2.9%	100.0%	
Sex	Mala	Count	30	2	32	32	0	32	31	1	32	
	Male	% within Sex	93.8%	6.3%	100.0%	100.0%	0.0%	100.0%	96.9%	3.1%	100.0%	
Total		Count	64	3	67	65	2	67	65	2	67	
		% within Sex	95.5%	4.5%	100.0%	97.0%	3.0%	100.0%	97.0%	3.0%	100.0%	

		Rib			Cervical			Thoracic			Lumbar		
		Absent	Present	Total	Absent	Present	Total	Absent	Present	Total	Absent	Present	Total
Female	Count	35	0	35	34	1	35	31	4	35	31	4	35
	% within Sex	100.0%	0.0%	100.0%	97.1%	2.9%	100.0%	88.6%	11.4%	100.0%	88.6%	11.4%	100.0%
Male	Count	27	5	32	31	1	32	27	5	32	29	3	32
	% within Sex	84.4%	15.6%	100.0%	96.9%	3.1%	100.0%	84.4%	15.6%	100.0%	90.6%	9.4%	100.0%
Total	Count	62	5	67	65	2	67	58	9	67	60	7	67
	% within Sex	92.5%	7.5%	100.0%	97.0%	3.0%	100.0%	86.6%	13.4%	100.0%	89.6%	10.4%	100.0%

 Table 9 - Frequencies of torso trauma by bone and sex

Table 10 - Frequencies of lower limb (tibia) trauma by sex

				Tibia				
			Absent Present Tota					
	Famala	Count	34	1	35			
C	Female	% within Sex	97.1%	2.9%	100.0%			
Sex	N / 1	Count	32	0	32			
	Male	% within Sex	100.0%	0.0%	100.0%			
Tatal		Count	66	1	67			
Total		% within Sex	98.5%	1.5%	100.0%			



Figure 11 - Graphic comparison of skeletal trauma frequencies by site



Figure 12 – Graphic comparison of skeletal trauma frequencies by sex



Figure 13 – Graphic comparison of skeletal trauma frequencies by status
	Ν	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	1.000	.594

# Table 11 - Test of association between status and upper limb trauma

Table 12 - Test of association between status and torso trauma

	Ν	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	.599	.383

Table 13 - Test of association between status and lower limb (tibia) trauma

	Ν	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	.149	.081

Table 14 - Test of association between sex and upper limb trauma

	Ν	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	.431	.253

Table 15 - Test of association between sex and torso trauma

	N	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	.594	.306

	N	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Fisher's Exact Test	67	1.000	.522

# Table 16 - Test of association between sex and lower limb (tibia) trauma

Table 17 - Frequencies of upper limb DJD by joint, status, and age group

Age Group					Shoulder			Wrist	
				Absent	Present	Total	Absent	Present	Total
		Non alita	Count	15	0	15	15	0	15
	Status	Non-ente	% within Status	100.0%	0.0%	100.0%	100.0%	0.0%	100.0%
< 20 maara	Status	Elita	Count	13	1	14	14	0	14
< 30 years		Ente	% within Status	92.9%	7.1%	100.0%	100.0%	0.0%	100.0%
	Total		Count	28	1	29	29	0	29
	Total		% within Status	96.6%	3.4%	100.0%	100.0%	0.0%	100.0%
		Non alita	Count	22	3	25	23	2	25
	Status	Non-ente	% within Status	88.0%	12.0%	100.0%	92.0%	8.0%	100.0%
20   voora	Status	Elita	Count	10	3	13	13	0	13
30+ years		Ente	% within Status	76.9%	23.1%	100.0%	100.0%	0.0%	100.0%
	Total		Count	32	6	38	36	2	38
	Total		% within Status	84.2%	15.8%	100.0%	94.7%	5.3%	100.0%

Age Group	1			C	ervical D	JD	Tł	noracic D.	JD	L	umbar D.	ID
_				Absent	Present	Total	Absent	Present	Total	Absent	Present	Total
		Non alita	Count	12	3	15	12	3	15	13	2	15
	Status	Non-ente	% within Status	80.0%	20.0%	100.0%	80.0%	20.0%	100.0%	86.2%	13.3%	100.0%
< 20	Status	Elite	Count	10	4	14	12	2	14	14	0	14
< 30 years Total		Elite	% within Status	71.4%	28.6%	100.0%	85.7%	14.3%	100.0%	100.0%	0.0%	100.0%
	Tatal		Count	22	7	29	24	5	29	27	2	29
		% within Status	75.9%	24.1%	100.0%	82.8%	17.2%	100.0%	93.1%	6.9%	100.0%	
		Non alita	Count	20	5	25	17	8	25	19	6	25
	Status	Non-ente	% within Status	80.0%	20.0%	100.0%	68.0%	32/0%	100.0%	76.0%	24.0%	100.0%
< 30 years Star Tot 30+ years Tot	Status	<b>D1</b> .	Count	11	2	13	11	2	13	13	0	13
		Ente	% within Status	84.6%	15.4%	100.0%	84.6%	15.4%	100.0%	100.0%	0.0%	100.0%
	T ( 1		Count	31	7	38	28	10	38	32	6	38
_	Total		% within Status	81.6%	18.4%	100.0%	73.7%	26.3%	100.0%	84.2%	15.8%	100.0%

Table 18 - Frequencies of spinal DJD by vertebral segment, status, and age group

		Age Group	)	ŀ	Knee DJD	
				Absent	Present	Total
		N	Count	14	1	15
	Status	Non-ente	% within Status	93.3%	6.7%	100.0%
< 30	Status	alita	Count	14	0	14
years		ente	% within Status	100.0%	0.0%	100.0%
		Tatal	Count	28	1	29
		Total	% within Status	96.6%	3.4%	100.0%
		Non alita	Count	22	3	25
	Status	Non-ente	% within Status	88.0%	12.0%	100.0%
30+	Status	alita	Count	12	1	13
years		ente	% within Status	92.3%	7.7%	100.0%
		Tatal	Count	34	4	38
		Total	% within Status	89.5%	10.5%	100.0%

Table 19 - Frequencies of lower limb (knee) DJD by status and age group

Table 20 - Frequencies of upper limb DJD by joint, sex, and age group
---

Age (	Group			Sh	oulder DJ	D	V	Vrist DJI	)
				Absent	Present	Total	Absent	Present	Total
		Famala	Count	19	1	20	20	0	20
	Corr	Female	% within Sex	95.0%	5.0%	100.0%	100.0%	0.0%	100.0%
< 30	) Sex N rs <u>N</u> Total	Mala	Count	9	0	9	9	0	9
years		Male	% within Sex	100.0%	0.0%	100.0%	100.0%	0.0%	100.0%
		1	Count	28	1	29	29	0	29
	Tota	l	% within Sex	96.6%	3.4%	100.0%	100.0%	0.0%	100.0%
		Famala	Count	14	1	15	14	1	15
	Corr	Female	% within Sex	93.3%	6.7%	100.0%	93.3%	6.7%	100.0%
30+	Total + Sex -	Mala	Count	18	5	23	22	1	23
years		Male	% within Sex	78.3%	21.7%	100.0%	95.7%	4.3%	100.0%
	Tata	1	Count	32	6	38	36	2	38
	Tota	L	% within Sex	84.2%	15.8%	100.0%	94.7%	5.3%	100.0%

Age G	roup			С	ervical D.	ID	Thoracic DJD			Lumbar DJD		
				Absent	Present	Total	Absent	Present	Total	Absent	Present	Total
		Female	Count	15	5	20	18	2	20	20	0	20
	C		% within Sex	75.0%	25.0%	100.0%	90.0%	10.0%	100.0%	100.0%	0.0%	100.0%
< 30	Sex	N/-1-	Count	7	2	9	6	3	9	7	2	9
years	_	Male	% within Sex	77.8%	22.2%	100.0%	66.7%	33.3%	100.0%	77.8%	22.2%	100.0%
_	T-4-1	Total Count		22	7	29	24	5	29	27	2	29
	I otal		% within Sex	75.9%	24.1%	100.0%	82.8%	17.2%	100.0%	93.1%	6.9%	100.0%
		E	Count	13	2	15	11	4	15	14	1	15
	C	Female	% within Sex	86.7%	13.3%	100.0%	73.3%	26.7%	100.0%	93.3%	6.7%	100.0%
30+	Sex	N/-1-	Count	18	5	23	17	6	23	18	5	23
years		Male	% within Sex	78.3%	21.7%	100.0%	73.9%	26.1%	100.0%	78.3%	21.7%	100.0%
	T-4-1		Count	31	7	38	28	10	38	32	6	38
	i otal		% within Sex	81.6%	18.4%	100.0%	73.37	26.3%	100.0%	84.2%	15.8%	100.0%

Table 21 - Frequencies of spinal DJD by vertebral segment, sex, and age group

Age Gro	up			Knee			
	_			Absent	Present	Total	
		Female	Count	20	0	20	
< 30 years	Carr		% within Sex	100.0%	0.0%	100.0%	
	Sex	Male	Count	8	1	9	
			% within Sex	88.9%	11.1%	100.0%	
	Tata	1	Count	28	1	29	
	Tota	1	% within Sex	96.6%	3.4%	100.0%	
		Female	Count	13	2	15	
	Carr		% within Sex	86.7%	13.3%	100.0%	
30+	Sex		Count	21	2	23	
years		Male	% within Sex	91.3%	8.7%	100.0%	
	Tata	1	Count	34	4	38	
	1 ota	1	% within Sex	89.5%	10.5%	100.0%	

Table 22 - Frequencies of lower limb (knee) DJD by sex and age group



Figure 14 - Graphic comparison of DJD by site for < 30 years age group



Figure 15 – Graphic comparison of DJD by site for 30+ years age group



Figure 16 – Graphic comparison of DJD by sex for < 30 years age group



Figure 17 – Graphic comparison of DJD by sex for 30+ years age group



Figure 18 – Graphic comparison of DJD by status for < 30 years age group



Figure 19 – Graphic comparison of DJD by status for 30+ years age group

Age Range			Exact Sig. (2-sided)	Exact Sig. (1-sided)
< 20	Fisher's Exact Test		.598	.473
< 30 years	N of Valid Cases	29		
20± voors	Fisher's Exact Test		1.000	.640
50+ years	N of Valid Cases	38		

Table 23 - Test of association between upper limb DJD and status by age group

Table 24 - Test of association between spinal DJD and status by age group

Age Range			Exact Sig. (2-sided)	Exact Sig. (1-sided)
	Fisher's Exact Test		.272	.180
< 30 years	N of Valid Cases	29		
	Fisher's Exact Test		.203	.164
30+ years	N of Valid Cases	38		

Table 25 - Test of association between lower limb DJD and status by age group

Age Range			Exact Sig. (2-sided)	Exact Sig. (1-sided)
	Fisher's Exact Test			
			.598	.473
< 30 years				
	N of Valid Cases	29		
	Fisher's Exact Test			
			.643	.433
30+ years				
	N of Valid Cases	38		

Age Group			Exact Sig. (2-sided)	Exact Sig. (1-sided)
	Fisher's Exact Test		.532	.312
< 30 years	N of Valid Cases	29		
	Fisher's Exact Test		1.000	.490
30+ years	N of Valid Cases	38		

Table 26 - Tests of association between upper limb DJD and sex by age group

Table 27 - Tests of association between spinal DJD and sex by age group

Age Group			Exact Sig. (2-sided)	Exact Sig. (1-sided)
	Fisher's Exact Test		.454	.336
< 30 years	N of Valid Cases	29		
	Fisher's Exact Test		.681	.419
30+ years	N of Valid Cases	38		

Table 28 - Tests of association between lower limb DJD and sex by age group

Age Group			Exact Sig. (2-sided)	Exact Sig. (1-sided)
	Fisher's Exact Test		.310	.310
< 30 years	N of Valid Cases	29		
	Fisher's Exact Test		1.000	.520
30+ years	N of Valid Cases	38		

that they bore the brunt of the most strenuous tasks associated with subsistence, infrastructure, and craft production. Tests of association between inferred status and DJD and sex and DJD were conducted separately for the two age groups. No statistically significant associations were found between status and DJD or sex and DJD for any of the body segments considered in either age group (Tables 23 through 28), and therefore the data do not support my hypothesis. This is likely due in part to low frequencies of DJD for the sample as a whole, something that Farnum (2002) also recognized in her analysis of skeletal remains from the Sicán Precinct and El Brujo. She did note more vertebral osteophytosis at El Brujo (primarily inferred commoners) compared to the Precinct (primarily inferred elites), a skeletal marker that I excluded from this analysis because its association to physical activity is spurious.

# Hypothesis 3 - ECs

I predicted that status groups would differ significantly in aggregate EC scores, even after statistically controlling for variations in size and age between groups. Aggregate EC scores, which are the sum total of all EC scores for each individual, will better clarify whether inferred commoners experienced greater biomechanical stress overall, as hypothesized. The minimum and maximum values, as well as the means and standard deviations for aggregate fibrous and fibrocartilaginous EC scores by site, sex, and status are reported in Tables 29 through 31. Mean aggregate fibrous and fibrocartilaginous EC scores for both sexes. Mean aggregate scores for males and females within each status group, however, were very similar. Interestingly, a search of the data revealed that inferred commoners from Cao Viejo had the highest scores in both fibrous and fibrocartilaginous insertion types for males as well as females. This is consistent with my

hypothesis that commoners were over-burdened by mechanical stress and strain relative to their elite counterparts, but requires further testing that accounts for variations in age and size, which are known influences on EC scores.

The first step for statistically testing my hypothesis that status differences will account for variation in aggregate EC scores beyond what is accounted for by humeral size and average age involved ranking these variables and performing a regression analysis. Residuals from the regression analysis were saved to create new variables, RECFib and RECCart. RECFib is the amount of aggregate fibrous EC scores not explained by average age and humeral size and RECCart denotes the same for the aggregate fibrocartilaginous scores. These variables were then subjected to One-way ANOVA, with RECFib or RECCart being the dependent variable, and status being the independent variable. P-values did not achieve statistical significance for aggregate fibrous as well as fibrocartilaginous ECs (Table 32), and thus the hypothesis that inferred status groups would vary with respect to aggregate EC scores was also not supported.

#### Hypothesis 4 - ECs

A final objective in this study was to test recent assertions put forth about how ECs should be scored and interpreted to yield information about physical activity. Henderson and Cardoso (2013); Villotte et al. (2010); and Weiss (2015) contend that fibrocartilaginous insertions are more reliable indicators of activity because, among other things, they are less sensitive to the influences of age and size. Based on their previous work, I predicted that correlations between aggregate fibrous EC scores (variable AgFibEC), average age, and humeral size would be positive, statistically significant, and stronger than those found for aggregate fibrocartilaginous ECs (variable AgCartEC).

Site			Min	Max	Mean	Std. Deviation
	Aggregate fibrous EC		9.00	19.00	14.6667	2.23931
El Brujo	Aggregate fibrocartilaginous EC		6.00	14.00	9.5833	2.46571
	Ν	24				
	Aggregate fibrous EC		2.00	16.00	9.5938	3.85235
Pachacamac	Aggregate fibrocartilaginous EC		.00	10.00	5.8125	2.94887
	Ν	16				
	Aggregate fibrous EC		.00	15.00	7.2593	5.19314
Sicán Precinct	Aggregate fibrocartilaginous EC		.00	11.00	4.5556	3.78594
	Ν	27				

Table 29 - Descriptive statistics for aggregate EC scores by enthesial type and site

Table 30 - Descriptive statistics for aggregate EC scores by enthesial type and sex

Sex						Std.
		Ν	Min	Max	Mean	Deviation
	Aggregate fibrous EC		.00	19.00	9.6286	4.76560
Female	Aggregate fibrocartilaginous EC		.00	14.00	6.4286	3.90539
	Ν	35				
	Aggregate fibrous EC		.00	17.00	11.3906	5.47242
Male	Aggregate fibrocartilaginous EC		.00	14.00	6.9063	3.83834
	Ν	32				

Status			Min	Max	Mean	Std. Deviation
	Aggregate fibrous EC		.00	19.00	11.3125	5.40025
Non- elite	Aggregate fibrocartilaginous EC		.00	14.00	7.3250	3.88546
	Ν	40				
	Aggregate fibrous EC		.00	16.00	9.2222	4.57698
Elite	Aggregate fibrocartilaginous EC		.00	12.00	5.6667	3.64797
	Ν	27				

Table 31 - Descriptive statistics for aggregate EC scores by enthesial type and status

Table 32 – One-way ANOVA for status and residual aggregate fibrous and fibrocartilaginous EC scores

Fibrous	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	484.500	1	484.500	1.542	.219
Within Groups	20427.414	65	314.268		
Total	20911.914	66			
Fibrogertiloginoug	a 6a	10		-	~ '
Fibrocartinaginous	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	Sum of Squares	df 1	Mean Square 536.796	F 1.526	Sig. .221
Between Groups Within Groups	Sum of Squares     536.796     22858.027	df 1 65	Mean Square 536.796 351.662	F 1.526	Sig. .221
Between Groups Within Groups Total	Sum of Squares   536.796   22858.027   23394.	df 1 65 66	Mean Square 536.796 351.662	F 1.526	.221

			AgFibEC	Average Age
		Correlation Coefficient	1.000	.318**
Spearman's rho	AgFibEC	Sig. (1-tailed)		.004
		Ν	67	67

Table 33 – Correlation of aggregate fibrous EC scores and average age

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table 34 - Correlation of aggregate fibrous EC scores and humeral size

			AgFibEC	Size
Spearman's rho	AgFibEC	Correlation Coefficient	1.000	.349**
		Sig. (1-tailed)		.002
		Ν	67	67

\*\*. Correlation is significant at the 0.01 level (1-tailed).

Table 35 - Correlation of	f aggregate fibro	cartilaginous EC	scores and	average age
	aggiegate more	our magnitous De	beeres and	a vorage age

		AgCartEC	Average age
	Correlation Coefficient	1.000	.235*
Spearman's rho AgCartEC	Sig. (1-tailed)		.028
	Ν	67	67

\*. Correlation is significant at the 0.05 level (1-tailed).

Table 36 -	Correlation	of aggregate	fibrocartilaginous	EC scores and	humeral size
			0		

		AgCartEC	Size
	Correlation Coefficient	1.000	.091
Spearman's rho AgCartEC	Sig. (1-tailed)		.231
	Ν	67	67

Fibrous EC scores did have a statistically significant relationship to age (p=0.004; Table 33) and humeral size (p=0.002; Table 34), as predicted. Age explained about 10% of the variation in aggregate fibrous EC scores, and the effect size based on the correlation coefficient of 0.318 indicates a medium or typical strength to their relationship. Humeral size explained about 12% of aggregate fibrous EC scores, and the strength of relationship between these variables is also considered medium (r = 0.349), although may be slightly stronger than that identified for age in my sample. Aggregate fibrocartilaginous EC scores were also significantly correlated with age (p=0.28; Table 35), although age only explained about 5% of the variation and the strength of the relationship is not strong (r = 0.235). No statistical correlation between aggregate fibrocartilaginous EC scores and humeral size was detected (Table 36), and therefore my data are consistent with Villotte (2006), Villotte et al. (2010) and Weiss' (2015) claims that fibrocartilaginous EC scores are less influenced by size. Additionally, while age did seem to have a significant influence on fibrocartilaginous EC scores in my sample, the relationship was much weaker than that found with fibrous entheses. Overall, my results support the claims that fibrocartilaginous entheses are less prone to systemic factors, and thus may yield better information about physical activity than fibrous entheses.

#### **CHAPTER 6**

# DISCUSSION

#### Summary of Results

In this study, I hypothesized that skeletal trauma, degenerative joint disease (DJD), and enthesial changes (ECs) would be more common and more severe among inferred Middle Sicán and Ychsma commoners relative to their elite counterparts. By testing commonly held assumptions about the relationship between manual labor and marginal social status based on western, Marxist notions of social class, I provide some clarity regarding how social inequalities were actually experienced as part of daily life in pre-Hispanic coastal Peru.

I found no significant relationship between inferred social status or sex and the prevalence or severity of skeletal trauma, DJD, or ECs in my sample. Evidence of trauma and DJD (especially in regions other than the spine) was so rare in this sample these measures were rendered ineffective for discerning patterns of activity with any degree of specificity. About as much as can be said in this regard is that overall, patterns of skeletal trauma and DJD in this sample suggest Ychsma and Middle Sicán –affiliated communities were either well-adapted to biomechanical stress and strain, or not subject to the kind of physical demands imposed during the colonial era when skeletal lesions such as DJD become dramatically more prevalent, at least on the north coast (Farnum 2002; Klaus 2008; Klaus and Chang 2011; Klaus et al. 2009). Contrary to conventional assumptions, elites did not appear to have had any discernible privilege compared to non-elites in terms of exposure to biomechanical stress and strain. Based on these

results, it is clear that models which postulate a simple dominant/subordinate relationship between status groups with regards to labor have little relevance for understanding social dynamics in Middle Sicán and Ychsma societies. This is perhaps to be expected for Ychsma society, as clear archaeological evidence of marked social classes has yet to be found. Individuals in the Ychsma sample considered here (i.e. Tomb 1-2) and identified as "elite" based on atypical funerary treatment and grave good "wealth" were largely indistinguishable from inferred commoners with regards to patterns of physical activity, and although sample size is small this does seem contrary to expectations for a class-based, economically-specialized society. This provides further support to the growing body of literature on Ychsma society which argues that although differences in mortuary treatment may reflect special relationships between the living and the dead and perhaps some form of social ranking, there is little reason to believe this is indicative of status or "social class" as conventionally understood by western standards (Diaz 2015; Diaz and Vallejo 2005; Owens and Eeckhout 2015). Lack of correspondence between inferred social status and physical activity as measured here in the Middle Sicán context where there is good evidence of status-based health inequalities and huge discrepancies in access to resources, however, requires further exploration. Below I synthesize the results of this study with recent interpretations of Middle Sicán labor organization based on archaeological and ethnohistorical data.

#### Social Identity & Labor in Middle Sicán Society: A Reappraisal

Historical accounts of *parcialidades;* defined in a general sense as hierarchically ordered networks of interdependent and specialized socio-economic units, are often used as a model for

interpreting labor organization in pre-Hispanic coastal Peru. Placement within the hierarchy depended to a large extent on the goods and services provided by each respective *parcialidad*, with more "specialized" groups associated with the production of elite insignia such as potters, weavers, metalsmiths, and traders, being afforded a certain degree of social prestige and exemption from menial labor tasks (Netherly 1977; Ramírez 2007). *Parcialidades* were linked together under larger regional authorities by economic exchange and interdependence, with the power of authorities being contingent upon their ability to provide for the spiritual and material well-being of their subjects (Netherly 1977). In the *parcialidad* system, aspects of social identity such as status and ethnicity were intimately tied to occupational activities, and we may reasonably expect a similar correspondence in archaeological contexts where this model seems appropriate.

Excavations in and around the Sicán Precinct attest to the presence of at least three distinct social classes, possibly divided to some extent along ethnic lines (Shimada 2014; Shimada et al. 2004). Bioarchaeological evidence of marked disparities in developmental and nutritional health between inferred ethnic Middle Sicán elites and the local, ethnically "Mochica" (i.e. affiliated with the earlier Moche archaeological culture) base attest to the overriding influence of status identity on life experience (Farnum 2002; Shimada et. al 2004), and there was good reason to assume that labor activities were an integral part of these dynamics. The sheer quantity and quality of material goods recovered from elite tombs at the Precinct suggests leaders were able to monopolize labor on a large-scale, and production contexts identify local Mochica agrarian communities as the likely "backbone" of these economic enterprises (Bezúr 2014; Klaus 2014). And yet as this study demonstrates, Middle Sicán-affiliated communities seem to have been equally protected from activity related injury and biomechanical stress, regardless of their

social status or biological sex. This compliments Shimada and Wagner's (2007) assertion based on extensive excavations of Middle Sicán production contexts that elite administrators exerted little if any direct control over labor, with workers being afforded a great deal of autonomy with respect to their day-to-day operations (Shimada and Wagner 2007).

The success of the Middle Sicán power structure has been attributed to their synthesis of pre-existing belief systems and lifeways into a newly configured religious ideology, serving to justify or naturalize their position of privilege and sphere of influence (Segura and Shimada 2014; Shimada and Samillán 2014). They did not achieve power through militaristic force or total subjugation, but through religious and economic cooption. Shimada and Craig (2013) and Shimada and Wagner's (2007) characterization of Middle Sicán labor organization as "modular;" or based on a collection of small, independent, local production networks pooling resources to produce on a large scale, provides a useful model for how this economic cooption may have been achieved without placing oppressive labor burdens on any particular social group(s). If the most physically demanding tasks such as ore extraction, construction, or the cleaning and maintenance of irrigation canals were completed in a segmentary fashion by a bunch of small work collectives sharing religious and/or ethnic ties, labor obligations may have been an integral, integrated, and rewarding component of everyday life, rather than an oppressive burden or digression.

Low frequencies of skeletal trauma and degenerative joint disease in my sample as a whole, as well as the lack of variation in aggregate EC scores between elites and non-elites more specifically, suggests that labor activities played little to no role in differential life experience for Middle Sicán status groups. While contrary to assumptions about labor division and social hierarchy based on Marxist notions of social class, the data presented here are consistent with Shimada and colleagues' (Shimada and Craig 2013; Shimada and Wagner 2007) characterization of Middle Sicán labor organization. If skeletal trauma, degenerative joint disease and ECs are the result of "high impact" and "strenuous" activities performed on a habitual basis as conventional interpretations postulate, then it is clear that non-elites affiliated with Middle Sicán society were not over-burdened by oppressive labor regimes, despite significant differences in the quality of life with regards to dietary patterns and exposure to developmental stressors between inferred status groups. A more equitable distribution of labor tasks than Marxist models of social class allow for was likely present in Middle Sicán society, which may help to explain why despite marked difference in access to resources, local populations incorporated into the Middle Sicán sphere were willing to engage with "the-powers-that-be." As long as relationships between the "rulers" and the "ruled" could be perceived of as mutually beneficial, systems of social stratification could remain in place.

My research accords well with current archaeological models of labor organization in Middle Sicán society and helps to clarify the role of labor activities in differential life experience for status groups. Nevertheless, although the underlying premise that habitual muscle and joint strain will initiate bone adaptive/remodeling responses and bone hypertrophy has strong clinical support, the specific magnitude, duration, and frequency of stress episodes necessary to initiate such responses are not well understood and may be highly variable (Schlecht 2012b; Villotte and Knüsel 2013). This is particularly true for ECs, where clinical correlates and experimental studies are few. There are multiple interpretations regarding what it is that EC scores actually measure, and this can greatly limit their interpretive power for understanding patterns of activity in ancient societies. I address this issue in some detail in the following section.

## Bioarchaeological Studies of Activity

# ECs as markers of physical activity

Contrary to conventional ideas about the relationship between habitual muscle strain and bony changes, Schlecht (2012a; 2012b) and Villotte and Knüsel (2013) postulate that EC development in adults is largely dependent upon mechanical loading and adaptive mechanisms established during adolescence when rates of bone growth and turnover are at their peak. They argue that ECs develop only when stress levels exceed some threshold established during this critical period, and when exposure is intermittent, so as to preclude further adaptive adjustments. According to this line of reasoning, if the labor requirements fulfilled by non-elites as part of their social obligations to community leaders required similar levels of physical exertion as activities performed on a more regular basis (so that the hypothesized adaptive thresholds were not exceeded), we might not expect to see marked differences in EC scores between status groups, despite any real or significant variations in their respective range of activities and physical demands. And thus, we may explain the apparent lack of variation in EC scores between elites and non-elites in this study to be a result of consistency in the kinds of activities completed by different status groups, rather than any similarities in the kind of physical exertion required to perform them. If non-elites worked significantly "harder" than elites, but did so on a regular basis from adolescence onward, they may potentially be indistinguishable from elites or others who did not engage in physically-demanding tasks at all. While in the case of this study low rates of skeletal trauma and degenerative joint disease do support the general conclusion that non-elites were no more exposed to biomechanical stress and strain than their elite counterparts, the lack of specificity regarding the best way to interpret EC data can pose some serious

limitations for their use in bioarchaeological studies of activity. Furthermore, as this and other studies have demonstrated, fibrous entheses are not particularly useful for reconstructing patterns of physical activity because they are strongly influenced by age and size. In light of these conclusions, EC studies should be limited to fibrocartilaginous entheses, as Villotte (2006); Villotte et al (2010) and Weiss (2015) have suggested. Additionally, EC scores in and of themselves may not be particularly informative about differential patterns of activity in ancient societies, because they can be interpreted in a number of different ways. Until we know more about EC etiology, ECs should be considered only in conjunction with other skeletal measures of biomechanical stress, and interpreted accordingly. In this study, correspondence between EC data, patterns of skeletal trauma and degenerative joint disease, and models of Middle Sicán and Ychsma societies based on extensive archaeological research helped to strengthen my conclusions, but such data may not always be readily available for a given context.

#### Directions for future research

Bioarchaeologists recognize human skeletal remains as valuable sources of information about the past precisely because of their potential to reveal aspects of human experience not typically accessible with other kinds of archaeological data. As the discipline has struggled to become more responsive to social theory and more relevant to larger anthropological issues as well as the general public, issues of identity, inequality, and embodiment have emerged as important sources of common ground. Greater engagement with post-processual ideas has ushered in a new era of self-reflection, as witnessed by the re-hashing of old debates and unresolved issues surrounding bioarchaeological studies of activity coming out of the recent Coimbra Workshop in Musculoskeletal Stress Markers. This workshop, a "task force" of sorts, was designed to outline best practices for how enthesial changes should be scored and analyzed. While these efforts and the resulting publications are a step in the right direction especially with regards to expanding dialogue between bioarchaeologists and those working in a clinical setting, many of the issues raised during initial critiques of EC studies in the mid 1990's have yet to be adequately addressed.

The biggest limitation to bioarchaeological studies of activity using ECs is lack of understanding with regards to the precise mechanisms responsible for their development and expression. Do they reflect habitual, sustained stresses, or more sporadic instances of "atypical" loading? What is the relative influence of loading intensity versus loading duration? What about nutritional, genetic, and hormonal influences, and how can these be best dealt with in a bioarchaeological context?

This study supports Villotte et. al. (2010; 2013) and Weiss' (2015) assertion that fibrocartilaginous insertion sites are less sensitive to the effects of size than the fibrous type, but it remains to be seen whether that translates into more precise, reliable behavioral interpretations. Better lines of communication between bioarchaeologists and those working in a clinical setting are a step in the right direction towards resolving these issues, as experimental studies will be essential and have thus far been rather limited (Schlecht 2012b). Until a better understanding of EC etiology is achieved, they will have limited utility for reconstructing patterns of behavior in bioarchaeological settings.

# REFERENCES

- Acosta M, Henderson CY, and Cunha E. 2017. The effects of terrain on entheseal changes in the lower limbs. International Journal of Osteoarchaeology 27:828-838.
- Agarwal SC, and Glencross BA. 2011. Building a Social Bioarchaeology. In: Agarwal SC, and Glencross BA, editors. Social Bioarchaeology. West Sussex, UK: Wiley-Blackwell. p 1-11.
- Agresti A. 2010. Analysis of ordinal categorical data, second edition. Hoboken, NJ: WIley & Sons, Inc.
- Armelagos GJ. 2003. Chapter 3. Bioarchaeology as Anthropology. Archeological Papers of the American Anthropological Association 13(1):27-40.
- Arnold B, and Jeske RJ. 2014. The Archaeology of Death: Mortuary Archaeology in the United States and Europe 1990-2013. Annual Review of Anthropology 43(1):325.
- Ashmore W, and Geller PL. 2005. Social DImensions of Mortuary Space. In: Rakita GFM, Buikstra JE, Beck LA, and Williams SR, editors. Interacting with the Dead: Perspectives on Mortuary Archaeology for the New Millennium. Gainsville, FL: University Press of Florida. p 81-92.
- Aufderheide AC, and Rodríguez-Martín C. 1998. Cambridge Encyclopedia of Human Paleopathology. Cambridge, UK: Cambridge University Press. 478 p.
- Babić Sa. 2005. Status identity and archaeology. In: Diaz-Andreu, Lucy S, Babić Sa, andEdwards DN, editors. The Archaeology of Identity: Approaches to Gender, Age, Status,Ethnicity, and Religion. New York, NY: Routledge. p 67-85.

- Baker BJ, and Agarwal SC. 2017. Stronger together: Advancing a global bioarchaeology. Bioarchaeology International 1(1-2):1-18.
- Becker SK, and Juengst SL. 2017. Introduction: Establishing a Bioarchaeology of Community. Archaeological Papers of the American Anthropological Association 28:6-12.
- Benjamin M, Kumai T, Milz S, Boszczyk BM, Boszczyk AA, and Ralphs JR. 2002. The skeletal attachment of tendons--tendon `entheses'. Comparative Biochemistry and Physiology -Part A: Molecular & Integrative Physiology 133(4):931-945.
- Bezúr A. 2003. Variability in Sicán copper alloy artifacts: Its relation to material flow patterns during the Middle Sicán Period in Peru, AD 900 - 1100 [Dissertation]. Tucson, AZ: University of Arizona.
- Bezúr A. 2014. La tecnología y organización de la producción de aleación de cobre de Sicán. In: Shimada I, editor. Cultura Sicán: Esplendor preincaico de costa norte. Lima, PE: Fondo Editorial del Congreso del Perú. p 93-106.
- Binford LR. 1971. Mortuary Practices: Their Study and Their Potential. Memoirs of the Society for American Archaeology(25):6-29.
- Blom D. 2017. A Bioarchaeological Perspective on Community and the Tension between Iindividual and Population. Archaeological Papers of the American Anthropological Association 28:104-111.
- Bourdieu P. 1984. Distinction: A Social Critique of the Judgement of Taste. Cambridge, MA: Harvard University Press.
- Boutin AT. 2011. Crafting a Bioarchaeology of Personhood: Osteobiographical Narratives from Alalakh. In: Baadsgaard A, Boutin AT, and Buikstra JE, editors. Breathing New Life into

the Evidence of Death: COntemporary Approaches to Bioarchaeology. Santa Fe, NM: SAR Press. p 109-133.

- Bramhall J, Scarpinato D, and Andrews J. 1994. Injuries in throwing sports. In: P R, editor. Clinical practice of sports injury prevention and care. Oxford, UK: Blackwell Scientific Publications. p 446-449.
- Brown J. 1995. On Mortuary Analysis with Specific Reference to the Saxe-Binford Research Program. In: Beck LA, editor. Regional Approaches to Mortuary Analysis. New York, NY: Plenum Press. p 3-26.
- Brown J, and Struever S. 1973. The organization of archaeological research: An Illinois example.In: Redman CL, editor. Research and theory in current archaeology. New York, NY:John Wiley & Sons. p 261- 280.
- Brown K, Pollintine P, and Adams M. 2008. Biomechanical implications of degenerative joint disease in the apophyseal joints of human thoracic and lumbar vertebrae. American Journal of Physical Anthropology 136:318-326.
- Brumfiel E. 1998. The multiple identities of Aztec craft specialists. In: Costin C, and Wright R, editors. Craft and social identity. Arlington, VA: The American Anthropological Association. p 145 152.
- Brumfiel E, and Earle T. 1987. Specialization, Exchange, and Complex Societies: An Introduction. In: Brumfiel E, and Earle T, editors. Specialization, Exchange, and Complex Societies. Cambridge, MA: Cambridge University press. p 1-9.
- Buikstra JE. 1977. Biocultural dimensions of archaeological study: A regional perspective. In:Blakely RL, editor. Biocultural adaptation in prehistoric America. Athens, GA:University of Georgia Press. p 67-84.

- Buikstra JE. 2006a. A Historical Introduction. In: Buikstra JE, and Beck LA, editors.Bioarchaeology: The Contextual Analysis of Human Remains. Burlington, MA: Elsevier.p 7-25.
- Buikstra JE. 2006b. Preface. In: Buikstra JE, and Beck LA, editors. Bioarchaeology: The contextual analysis of human remains. Burlington, MA: Academic Press. p xvii-xx.
- Buikstra JE, Baadsgaard A, and Boutin AT. 2011. Introduction. In: Buikstra JE, Baadsgaard A, and Boutin AT, editors. Breathing New Life into the Evidence of Death: COntemporary approaches to Bioarchaeology. Santa Fe, NM: SAR Press. p 3-26.
- Buikstra JE, and Scott RE. 2011. Key concepts in identity studies. In: Knudson KJ, andStojanowski CM, editors. Bioarchaeology and identity in the Americas. Gainsville, FL:University Press of Florida. p 24-55.
- Buikstra JE, and Ubelaker DH, editors. 1994. Standards for Data Collection from Human Skeletal Remains. Fayetteville, AR: Arkansas Archaeological Survey. 206 p.
- Cardoso F, and Henderson C. 2010. Enthesopathy formation in the humerus: Data from known age-at-death and known occupation skeletal collections. American Journal Of Physical Anthropology 141:550-560.
- Carr C. 1995. Mortuary practices: Their social, philosophical-religious, circumstantial, and physical determinants. Journal of Archaeological Method & Theory 2(2):105-200.
- Cavallaro R, and Shimada I. 1988. Some Thoughts on Sican Marked Adobes and Labor Organization. American Antiquity 53(1):75-101.
- Cervantes G, Shimada I, Klaus HD, Knudson KJ, and Shinoda K-i. 2011. Multi-ethnicity in the Sicán world: Figurines and other lines of evidence. Annual Meeting of the Society for American archaeology. Sacramento, CA.

- Chandola T, and Marmot MG. 2011. Socioeconomic status and stress. In: Contrada RJ, and Baum A, editors. The handbook of stress science: Biology, psychology, and health. New York, NY: Springer Publishing. p 185-193.
- Chapman F. 1972. Vertebral osteophytosis in prehistoric populations of central and southern Mexico. American Journal of Physical Anthropology 36:31-38.
- Chen YM, and Bohrer SP. 1990. Coracoclavicular and coracoacromial ligament calcification and ossification. Skeletal Radiology 19(4):263-266.
- Cheverko CM, and Hubbe M. 2017. Comparisons of statistical techniques to assess age-related skeletal markers in bioarchaeology. American Journal of Physical Anthropology 163:407-416.

Childe V. 1950. The Urban Revolution. The Town Planning Review 21(1):3 - 17.

- Christiansen CH. 1999. Defining Lives: Occupation as Identity: An Essay on Competence, Coherence, and the Creation of Meaning. American Journal of Occupational Therapy 53:547-558.
- Churchill S, and Morris A. 1998. Muscle marking morphology and labour intensity in prehistoric Khoisan foragers. International Journal of Osteoarchaeology 8:390-411.
- Cleland KM, and Shimada I. 1998. Paleteada Potters: Technology, Production Sphere, and Sub-Culture in Ancient Peru. In: Shimada I, editor. Andean Ceramics: Technology, Organization, and Approaches. Philadelphia, PA: The University Museum if Archaeology and Anthropology, University of Pennsylvania. p 111 150.
- Cohen MN, and Armelagos GJ, editors. 1984. Paleopathology at the Origins of Agriculture. Orlando, FL: Academic Press, Inc.

- Cohen J. 1988. Statistical power and analysis for the behavioral sciences, Second edition. Hillsdale, NJ: Lawrence Erlbaum Associates.
- Cornejo MA. 2000. La Nación Ischma y La Provincia Inka de Pachacamac. Arqueológicas 24:149-172.
- Corruccini R, and Shimada I. 2002. Dental relatedness corresponding to mortuary patterning at Huaca Loro, Peru. American Journal of Physical Anthropology 117:113-121.
- Costin C. 1998a. Housewives, chosen women, skilled men: Cloth production and social identity in the Late Prehispanic Andes. In: Costin C, and Wright R, editors. Craft and Social Identity. Arlington, VA: American Anthropological Association. p 123-141.
- Costin C. 1998b. Introduction: Craft and Social Identity. In: Costin C, and Wright R, editors. Craft and Social Identity. Arlington, VA: The American Anthropological Association. p 3-16.
- Costin C. 2006. Craft Production. In: Maschner H, editor. Handbook of methods in Archaeology. Lanham, MD: AltaMira Press. p 1032 - 1105.
- Costin C. 2016. Who Benefits? Structural Change and Lived Experience in the Late Prehispanic Andes. Archaeological Papers of the American Anthropological Society 27:120-142.
- Covey A. 2008. Multiregional perspectives on the archaeology of the Andes During the Late Intermediate Period (c. AD 1000-1400). J Archaeol Res 16:287-338.
- Crossland Z. 2010. Materiality and Embodiment. In: Hicks D, and Beaudry MC, editors. The Oxford Handbook of Material Culture Studies. New York, NY: Oxford University Press. p 386-405.

- CSDH. 2008. Closing the gap in a generation: health equity through action on the social determinants of health: final report of the commission on social determinants of health. Geneva, CH: World Health Organization.
- Csordas TJ. 1994. Introduction: The body as representation and being-in-the-world. In: Csordas TJ, editor. Embodiment and Experience: The existential ground of culture and self. New York, NY: Cambridge University Press. p 1-24.
- Davidson K, Davies C, and Randolph-Quinney P. 2011. Skeletal trauma. In: Black S, and Ferguson E, editors. Forensic anthropology: 2000 - 2010. Boca Raton, FL: CRC Press. p 183 - 236.
- Derevenski JRS. 2000. Sex differences in activity-related osseous change in the spine and the gendered division of labor at Ensay and Wharram Percy, UK. American Journal of Physical Anthropology 111(3):333-354.
- Díaz L, and Vallejo F. 2005. Clasificación del patron funerario Ychsma identificado en Armatambo y la Rincondad Alta. Corriente Arqueológica 1:223-322.
- Díaz LA. 2015. The preparation of corpses and mummy bundles in Ychsma funerary practice at Armatambo. In: Eeckhout P, and Owens LS, editors. Funerary practices and models in the ancient Andes. New York, NY: Cambridge University Press. p 186—209.
- Díaz LEA. 2004. Armatambo y la sociedad Ychsma. Bulll Inst fr études andines 33(3):571-594.
- Eeckhout P. 1999. Pirámide con rampa No. III, Pachacamac nuevos datos, nuevas perspectivas. Bull Inst fr études andines 28:169-214.
- Eeckhout P. 2004a. La Sombra de Ychsma: Ensayo Introductorio sobre la arqueología de la costa central del Perú en los Periodos Tardíos. Bulletin de L'Institut Français d'Etudes Andines 33(3):403-423.

- Eeckhout P. 2004b. Pachacamac y el proyecto Ychsma (1999-2000). Boletín del Instituto Francés de Estudios Andinos 33:425-448.
- Eeckhout P. 2008a. The Coastal Icon and press-molded Casma-Supe vessels at the MRAH. Bulletin des Musées Royaux d'Art et d'Histoire de Bruxelles 79:321-332.
- Eeckhout P. 2008b. Poder y Jerarquías Ychsmas En El Valle De Lurín. Arqueología y Sociedad 19:223-240.
- Eeckhout P. 2013. Change and permanency on the coast of ancient Peru: the religious site of Pachacamac. World Archaeology 45(1):137-160.
- Eeckhout P, and Owens LS. 2008. Sacrifice at Pachacamac. Latin American Antiquity 19(4):375-398.
- Faccia KJ, and Williams RC. 2008. Schmorl's Nodes: Clinical significance and implications for the bioarcaheological record. International Journal of Osteoarchaeology 18:22048.
- Fahey V, K O, Silberstein M, Anderson R, and Briggs C. 1998. The pathogenesis of Schmorl's nodes in relation to acute trauma: An autopsy study. Spine 23(21):2272-2275.
- Farnum J. 2002. Biological Consequences of Social Inequalities in Prehistoric Peru [dissertation]. Columbia, MO: University of Missouri, Columbia.
- Felson D, Lawrence R, Dieppe P, Hirsch R, and Helmick G. 2000. Osteoarthritis: New insights. Part I: The disease and its risk factors. Annals of Internal Medicine 133:635-646.
- Fisher G, and Loren DD. 2003. Introduction: Embodying Identity in Archaeology. Cambridge Arcaheological Journal 13:225-230.
- Franco R, and Gálvez C. 2014. Contextos funerarios de Transición y Lambayeque en el Complejo El Brujo, Valle Chicama. In: Alvarado J, and La Torre E, editors. Cultura

Lambayeque en el contexto de la costa norte del Perú. Chiclayo, PE: Universidad Católica Santo Toribio de Mogrovejo. p 139-165.

- Franco R, Gálvez C, and Vásquez S. 2007. El Brujo: Prácticas funerarias post mochicas. Lima, PE: Fundacion Augusto N. Wiese.
- Goldstein DJ, and Shimada I. 2010. Feeding the fire: Food and craft production in the Middle Sicán Period (AD 950 - 1050). In: Klarich EA, editor. Inside ancient kitchesn: New directions in the study of daily meals and feasts. Boulder, CO: University Press of Colorado. p 161-189.
- Goldstein J, and Shimada I. 2007. Middle Sicán multicraft production: Resource management and labor organization. In: Shimada I, editor. Craft production in complex societies. Salt Lake City, UT: University of Utah Press. p 44-67.
- Goldstein L. 2006. Mortuary Analysis and Bioarchaeology. In: Buikstra JE, and Beck LA, editors. Bioarchaeology: The COntextual Analysis of Human Remains. Burlington, MA: Elsevier. p 375-387.
- Goode K, Taylor RW, and Gutierrez C. 2018. Entheseal changes and demographic/health indicators in the upper extermity of modern Americans: associations with age and physical activity. International Journal of Osteoarchaeology 28:285-293.
- Goodman AH. 1998. The Biological Consequences of Inequality in Antiquity. In: Goodman AH, and Leatherman TL, editors. Building a New Biocultural Synthesis: Political-Economic Perspectives on Human Biology. Ann Arbor, MI: The University of Michigan Press. p 147-169.
- Gravlee CC. 2009. How race becomes biology: Embodiment of social inequality. American Journal of Physical Anthropology 139(1):47-57.
- Hawkey DE. 1988. Use of Upper Extremity Enthesopathies to Indicate Habitual Activity Patterns. Tempe, AZ: Arizona State University.
- Hawkey DE, and Merbs CF. 1995. Activity-induced Musculoskeletal Stress Markers (MSM) and Subsistence Strategy Changes among Ancient Hudson Bay Eskimos. International Journal of Osteoarchaeology 5:324-338.
- Hayden B. 1995. Pathways to power: Principles for creating socioeconomic inequalities. In:Price T, and Feinman G, editors. Foundations of Social Inequality. New York, NY:Plenum Press. p 15-86.
- Hegmon M. 2003. Setting theoretical egos aside: Issues and theory in North American archaeology. American Antiquity 68(2):213-243.
- Helms MW. 1993. Craft and the Kingly Ideal: Art, trade, and power. Austin, TX: University of Texas Press.
- Henderson CY. 2013. Subsistence strategy changes: the evidence of entheseal changes. Homo -Journal of Comparative Human Biology 64:491-508.
- Henderson CY, and Cardoso FA. 2013. Special Issue Entheseal Changes and Occupation: Technical and Theoretical Advances and Their Applications. International Journal of Osteoarchaeology 23:127-134.
- Henderson CY, Mariotti V, Pany-Kucera D, and Villotte S. 2016. The new 'Coimbra Method': A biologically appropriate method for recording specific features of fibrocartilaginous entheseal changes. International Journal of Osteoarchaeology 26:925-932.
- Henderson CY, and Nikita E. 2016. Accounting for multiple effects and the problem of small sample size in osteology: a case study focusing on enthesial changes. Archaeological and Anthropological Sciences 8:805-817.

- Henrich J, and Boyd R. 2008. Division of Labor, Economic Specialization, and the Evolution of Social Stratification. Current Anthropology 49(9):715-724.
- Hodder I. 1980. Social structure and cemeteries: a critical appraisal. In: Rahtz P, Dickinson T, and Watts L, editors. Anglo-Saxon Cemeteries, 1979. Oxford, UK: British Archaeological Reports. p 161-169.
- Hodder I, and Hutson S. 2003. Reading the Pasr: Current approaches to interpretation in archaeology.
- Howell DC. 2007. Statistical Methods for Psychology, Sixth Edition. Belmont, CA: Thomson Wadsworth. 739 p.
- Howell TL, and Kintigh KW. 1996. Archaeological Identification of Kin Groups Using Mortuary and Biological Data: An Example from the American Southwest. American Antiquity 61(3):537-554.
- Inomata T. 2001. The Power and Ideology of Artistic Creation: ELite Craft Specialists in Classic Maya Society. Current Anthropology 42(3):321 349.
- Inomata T. 2007. Knowledge and Belief om Artistic Production by Classic Maya Elites. Archaeological Papers of the American Anthropological Association 17(1):129-141.
- Isbell WH. 1995. Constructing the Andean Past or 'As You Like It'. Journal of the Steward Anthropological Society 23:1-12.

Joyce RA. 2005. Archaeology of the Body. Annual Review of Anthropology 34:139-158.

Jurmain R, and Kilgore L. 1999. Stories from the skeleton: Behavioral reconstruction in human osteology. Amsterdam: Gordon & Breach.

- Jurmain RD, Cardoso FA, Henderson C, and Villotte S. 2012. Bioarchaeology's Holy Grail: The Reconstruction of Activity. In: Grauer AL, editor. A Companion to Paleopathology. West Sussex, UK: Blackwell Publishing, Ltd. p 531-552.
- Kakaliouras AM. 2017. Theory for a Bioarcaheology of Community: Potentials, Practices, and Pitfalls. Archaeological Papers of the American Anthropological Association 28:13-23.
- Kennedy K. 1989. Skeletal markers of occupational stress. In: Iscan MY, and Kennedy K, editors. Reconstruction of life from the skeleton. New York, NY: ALan R. Liss. p 129-160.
- Kerbo H. 2009. Social Stratification and Inequality: Class conflict in historical, comparative, and golbal perspective. New York, NY: McGraw Hill Higher Education.
- Klaus HD. 2003. Life and death at Huaca Sialupe: The mortuary archaeology of a Middle Sicán community, North coast of Peru. Carbondale, IL: Southern Illinois University, Carbondale.
- Klaus HD. 2008. Out of Light Came Darkness: Bioarchaeology of Mortuary Ritual, Biological Stress, and Ethnogenesis in the Lambayeque Valley Complex, North Coast of Peru (AD 900-1750) [Dissertation]. Columbus, OH: Ohio State University.
- Klaus HD. 2014. La población mochica de la cultura Sicán Medio: Una primera aproximación a un sustrato cultural prehispánico tardío del valle de Lambayeque. In: Shimada I, editor.
  Cultura Sicán: Esplendor preincaico de la costa norte. Lima, PE: Fondo Editorial del Congreso del Perú. p 239-261.
- Klaus HD, and Chang MET. 2011. Surviving contact: Biological transformation, burial, and ethnogenesis in the colonial Lambayeque valley, Peru. In: Knudson KJ, and Stojanowski

CM, editors. Bioarchaeology and identity in the Americas. Gainsville, FL: University Press of Florida. p 126-152.

- Klaus HD, Larsen CS, and Tam ME. 2009. Economic intensification and degenerative joint disease: Life and labor on the postcontact north coast of Peru. American Journal of Physical Anthropology 139(2):204-221.
- Klaus HD, and Shimada I. 2016. Bodies and blood: Middle Sicán human sacrifice in the Lambayeque Valley Complex (AD 900 - 1100). In: Klaus HD, and Toyne JM, editors.
  Ritual Violence in the ancient Andes: Reconstructing sacrifice on the North coast of Peru.
  Austin, TX: University of Texas Press. p 120-149.
- Klaus HD, Shimada I, Shinoda K-i, and Muno S. 2017. Middle Sicán Mortuary Archaeology,
  Skeletal Biology, and Genetic Structures in late Pre-Hispanic South America. In: Klaus
  HD, Harvey AR, and Cohen MN, editors. Bones of Complexity: Bioarchaeological case
  studies of social organization and skeletal biology. Gainsville, FL: University Press of
  Florida. p 408-449.
- Knapp A, and van Dommelen P. 2008. Past Practices: Rethinking Individuals and Agents in Archaeology. Cambridge Archaeological Journal 18(1):15-34.
- Knudson K, and Stojanowski C. 2008. New Directions in Bioarchaeology: Recent Contributions to the Study of Human Social Identities. Journal of Archaeological Research 16(4):397-432.
- Knüsel CJ. 2010. Bioarchaeology: A synthetic approach. Bull Mem Soc Anthropol Paris 22:62-73.

- Krieger N. 1999. Embodying Inequality: A Review of Concepts, Measures, and Methods for Studying Health Consequences of Discrimination. International Journal of Health Services 29(2):295-352.
- Krieger N. 2001. Historical roots of social epidemiology: socioeconomic gradients in health and contextual analysis. International Journal of Epidemiology 30(4):899-900.
- Larsen CS. 1997. Bioarchaeology: Interpertating behavior from the human skeleton. Cambridge, UK: Cambridge University Press. 461 p.
- Larsen CS. 2017. Foreward. In: Klaus HD, Harvey AR, and Cohen MN, editors. Bones ofComplexity: Bioarchaeological Case Studies of Social Organization and Skeletal Biology.Gainsville, FL: University Press of Florida. p xvii xix.
- Larsen CS. 2018. Bioarchaeology in perspective: From classifications of the dead to conditions of the living. American Journal of Physical Anthropology 165:865-878.
- Larsen CS, and Milner GR, editors. 1994. In the wake of contact: Biological responses to conquest. New York, NY: Wiley-Liss.
- Lass B. 1998. Crafts, chiefs, and commoners: Production and control in precontact Hawai'i. In: Costin C, and Wright R, editors. Craft and Social Identity. Arlington, VA: The American Anthropological Association. p 19-30.
- Lopez-Hurtado E. 2015. Far Away, So Close: Living with the ancestors in Panquilma, Peruvian Central Coast. In: Eeckhout P, and Owens LS, editors. Funerary Practices and Models in the Ancient Andes. New York, NY: Cambridge University Press. p 24-33.
- Lovell NC. 2008. Analysis and interpretation of skeletal trauma. In: Katzenberg AM, and Saunders SR, editors. Biological anthropology of the human skeleton. New York, NY: Wiley & Sons, Inc. p 341-.

- Marmot MG, Rose G, Shipley M, and Hamilton P. 1978. Employment grade and coronary haert disease in British civil servants. Journal of Epidemiology and Community Health 32(4):244-249.
- Marmot MG, and Smith GD. 1991. Health inequalities among British civil servants: The Whitehall II study. Lancet 337(8754):1387.
- Marsteller SJ. 2015. Community identity and social diversity on the central Peruvian coast: A bioarchaeological investigation of Ychsma diet, mobility, and mortuary practices [dissertation]. Tempe, AZ: Arizona State University.
- Marsteller SJ, Zolotova N, and Knudson K. 2017. Investigating economic specialization on the central Peruvian coast: A reconstruction of Late Intermediate Period Ychsma diet using stable isotopes. American Journal of Physical Anthropology 162:300-317.
- Martin DL. 1998. Owning the Sins of the Past: Historical Trends, Missed Opportunities, and new Directions in the Study of Human Remains. In: Goodman AH, and Leatherman TL, editors. Building a new Biocultural Synthesis: Political-Economic Perspectives on Human Biology. Ann Arbor, MI: THe University of Michigan Press. p 171-190.
- Martin DL, Harrod RP, and Pérez VR. 2013. Bioarchaeology: An Integrated Approach to Working with Human Remains. New York, NY: Springer Science+Business Media. 262 p.
- Matsumoto G. 2014. El culto a los ancestros: aproximacion y evidencia. In: Shimda I, editor. Cultura Sican : Esplendor preincaico de la costa norte. Lima, PE: Editorial del Congreso del Perú. p 195-215.
- McGuire R. 1992. A Marxist Archaeology. San Diego, CA: Academic Press, Inc.

- Meskell L. 1996. The Somatization of Archaeology: Institutions, Discourses, Corporeality. Norwegian Archaeological Review 29(1):1-16.
- Meskell L. 2001. Archaeologies of Identity. In: Hodder I, editor. Archaeological Theory Today. Oxford, UK: Polity Press. p 187-213.
- Meskell L. 2002. The Intersections of Identity and Politics in Archaeology. Annual Review of Anthropology(31):279-301.
- Metcalf P, and Huntington R. 1991. Celebrations of Death: The Anthropology of Mortuary Ritual. New York, NY: Cambridge University Press.
- Michopoulou E, Nikita E, and Valakos ED. 2015. Evaluating the efficiency of different recording protocols for entheseal changes in regards to expressing activity patterns using archival data and cross-sectional geometric properties. American Journal of Physical Anthropology 158(4):557-568.
- Milella M, Cardoso FA, Assis S, and Lopreno GP. 2015. Exploring the Relationship Between Entheseal Changes and Physical Activity: A Multivariate Study. American Journal of Physical Anthropology 156:215-223.
- Molnar P. 2006. Tracing prehistoric activities: Musculoskeletal stress marker analysis of a stoneage population on the Island of Gotland in the Baltic sea. American Journal of Physical Anthropology 129(1):12-23.
- Morgan GA, Leech NL, Gloeckner GW, and Barrett KC. 2007. SPSS for Introductory Statistics: Unse and Interpretation. Mahwah, NJ: Lawrence Erlbaum Associates.
- Mujica EB, Jordán RG, Mora C, Quilter J, Cruz AM, De la Cruz CG, Cisneros V, Alcalde SL, Verano J, and Merello MA. 2007. El Brujo: Huaca Cao, a Moche ceremonial center in the Chicama Valley. Lima, Peru: Fundación Wiese.

Netherly PJ. 1977. Local level lords on the North coast of Peru. Ithaca, NY: Cornell University.

- Netherly PJ. 1984. The management of late Andean irrigation systems in the north coast of Peru. American Antiquity 49:227-254.
- Netherly PJ. 1990. Out of many, one: The organization of rule in the north coast polities. In: Moseley M, and Cordy-Collins A, editors. The Northern Dynasties: Kingship and Statecraft in Chimor. Washington, DC: Dumbarton Oaks. p 461-487.
- Niinimäki S. 2011. What do muscle marker ruggedness scores actually tell us? International Journal of Osteoarchaeology 21(3):292-299.
- Nikita E. 2014. The Use of Generalized Linear Models and Generalized Estimating Equations in Bioarchaeological Studies. American Journal of Physical Anthropology 153:473-483.
- Owens LS, and Eeckhout P. 2015. To the god of death, disease and healing: Social bioarchaeology of Cemetery I at Pachacamac. In: Eeckhout P, and Owens LS, editors.
  Funerary practices and models in the ancient Andes. New York, NY: Cambridge University Press. p 158-185.
- Parfitt A. 2004. The attainment of peak bone mass: What is the relationship between muscle growth and bone growth? Bone 34:767-770.
- Partridge R, and Duthie J. 1968. Rheumatism in dockers and civil servants: A comparison of heavy manual and sedentary workers. Annals of the Rheumatic Diseases 27:559-568.
- Pearson MP. 2000. The Archaeology of Death and Burial. College Station, TX: Texas A & M University Press. 250 p.
- Pearson OM, and Buikstra JE. 2006. Behavior in the Bones. In: Buikstra JE, and Beck LA, editors. Bioarchaeology: The Contextual Analysis of Human Remains. Burlington, MA: Elsevier. p 207-225.

- Peng B, Wu W, Hou S, Shang W, Wang X, and Yang Y. 2003. The pathogenesis of Schmorl's nodes. J Bone Joint Surg Br 85-B(6):879-882.
- Peregrine P. 1991. Some Political Aspects of Craft Specialization. World Archaeology 23(1):1 11.
- Porčić M, and Stefanović S. 2009. Physical activity and social status in Early Bronze Age society: The Mokrin necropolis. Journal of Anthropological Archaeology 28:259-273.
- Quade D. 1966. On Analysis of Variance for the k-sample problem. Annals of Mathematical Statistics 37:1747-1758.
- Rakita GFM. 2014. Bioarchaeology as a process: An examination of bioarchaeological tribes in the USA. In: O'Donnabhain B, and Lozada MC, editors. Archaeological human remains: Global perspectives. New York, NY: Springer International Publishing. p 213-234.
- Ramírez S. 2007. It's all in a day's work: Occupational specialization on the Peruvian North Coast, revisited. In: Shimada I, editor. Craft production in complex societies: Multicraft and producer perspectives. Salt Lake City, UT: The University of Utah Press. p 262-280.
- Ramírez SE. 2008. Los curacas y las jerarquías del poder en la costa norte. In: Makowski K, editor. Señores de los reinos de la luna. Lima, PE: Banco de Crédito. p 223-245.
- Ramírez-Horton S. 1981. La organizacion economica de la costa norte: Un analisis preliminar del Periodo Prehispanico Tardio. In: Castelli A, Koth de Paredes m, and Mould de Pease M, editors. Etnohistoria y antropologia andina. Lima, Peru: Centro de Proyección Cristiana. p 281-297.
- Rapillard L, Charlebois M, and Zysset PK. 2006. Compressive fatigue behavior of human vertebral trabecular bone. Journal of Biomechanics 39(11):2133-2139.

- Reents-Budet D. 1998. Elite maya pottery and artisans as social indicators. In: Costin C, and Wright R, editors. Craft and Social Identity. Arlington, VA: The American Anthropological Association. p 71 - 92.
- Rengifo CC, and Castillo LJ. 2015. The construction of social identity: Tombs of specialists at San José de Moro, Jequetepeque Valley, Perú. In: Eeckhout P, and Owens LS, editors.
  Funerary practices and models in the ancient Andes. New York, NY: Cambridge University Press. p 117-136.
- Rheinheimer D, and Penfield D. 2001. The effects of Type 1 error rate and power of ANCOVA f-tests and selected alternatives under non-normality and variance heterogeneity. Journal of Experimental Education 69:373-391.
- Robb J. 2010. Beyond Agency. World Archaeology 4:493-520.
- Robb J, Bigazzi R, Lazzarini L, Scarsini C, and Sonego F. 2001. Social "status" and biological "status": A comparison of grave goods and skeletal indicators from Pontecagnano.
   American Journal of Physical Anthropology 115:213-222.
- Robb JE. 1998. The interpretation of skeletal muscle sites: a statistical approach. International Journal of Osteoarchaeology 8(5):363-377.
- Rogers J, and Waldron T. 1995. A Field Guide to Joint Disease in Archaeology. New York, NY: John Wiley.
- Rostworowski M. 1975. Pescadores, Artesanos y Mercaderes Costeños en El Perú Prehispánico. Revista del Museo Nacional XLI:311-349.
- Rothschild BM. 1997. Porosity: A curiosity without diagnostic significance. American Journal of Physical Anthropology 104(4):529-533.

- Ruff C, Holt B, and Trinkus E. 2006. Who's afriad of the big bad wolff?: "Wolff's law" and bone functional adaptation. American Journal of Physical Anthropology 129:484-498.
- Santos AL, Alves-Cardoso F, Assis S, and Villotte S. 2011. The Coimbra Workshop in Musculoskeletal Stress Markers (MSM): an annotated review. Antropologia Portuguesa 28:135-161.
- Saxe AA. 1970. Social Dimensions of Mortuary Practices: University of Michigan. 240 p.
- Schlecht S. 2012a. A histomorphic analysis of muscular insertion regions: Understanding enthesis etiology [Dissertation]. Columbus, Ohio: Ohio State University. 162 p.
- Schlecht S. 2012b. Understanding entheses: Bridging the gap between clinical and anthropological perspectives. The Anatomical Record 295:1239-1251.
- Schortman EM, and Urban P. 2004. Modeling the Roles of Craft Production in Ancient Political Economies. Journal of Archaeological Research 12(2):185 - 226.
- Schrader SA. 2014. Elucidating inequality in Nubia: An examination of entheseal changes at Kerma (Sudan). American Journal of Physical Anthropology 156(2):192-202.
- Segura R, and Shimada I. 2010. The Wari footprint in the central coast: A view from Cajamarquilla and Pachacamac. In: Jennings J, editor. Beyond Wari walls: Exploring the nature of Middle Horizon Peru away froom Wari centers. Albuquerque, NM: University of New Mexico Press. p 113-135.
- Segura R, and Shimada I. 2014. La interacción Sicán Medio Costa Central, hacia 1000 d.C. In:
  Shimada I, editor. Cultura Sicán: Esplendor Preincaico de la Costa Norte. Lima, PE:
  Fondo Editorial del Congreso del Perú. p 303-322.
- Service E. 1971. Primitive Social Organization: An Evolutionary Perspective. New York, NY: Random House.

- Shanks M, and Tilley C. 1987. Re-Constructing Archaeology: Theory and Practice. Cambridge: Press Syndicate of the University of Cambridge.
- Shaw H, and Benjamin M. 2007. Structure-function relationship of entheses in relation to mechanical loading and exercise. Scandinavian Journal of Medicine and Science in Sprots 17:303-315.
- Shimada I. 1991. Pachacamac Archaeology: Retrospect and Prospect. Pachacamac: A Reprint of the 1903 Edition by Max Uhle. Philadelphia, PA: University of Pensylvania Press. p 14-67.
- Shimada I. 1994. Pampa Grande and the Mochica Culture. Austin, TX: University of Texas Press.
- Shimada I. 2000. The late prehispanic coastal states. In: Minelli L, editor. The Inca world: The development of Pre-Columbian Peru, AD 1000-1534. Norman, OK: University of Oklahoma Press. p 49-110.
- Shimada I. 2001. Late Moche Urban Craft Production: A First Approximation. In: Pillsbury J, editor. Moche Art and Arcaheology. New Haven, NY: Yale University Press.
- Shimada I. 2006. Unraveling the Secrets of the Pre-Hispanic Royal Tomb of Sicán: Wealth, power, memory, and kinship. In: Hildebrand J, and Buckley K, editors. Ancient Peru Unearthed: Golden Treasures of a Lost Civilization. Calgary, AB: The Nickle Arts Museum. p 80-114.
- Shimada I. 2010. The Sicán culture: Its development, characteristics, and legacies. In: Shimada I, and Cervantes G, editors. Behind the golden mask: The Sicán culture of the north coast of Peru. Lima: Editorial del Congreso del Perú.

- Shimada I. 2014. Detrás de la Máscara de Oro: La Cultura Sicán. In: Shimada I, editor. Cutura Sicán: Esplendor Preincaico de la costa norte. Lima, PE: Fondo Editorial del Congreso del Perú. p 15-90.
- Shimada I. 2015. Obsessed with metal: Manufacture, use, and significance of metals in the thousand year old Sicán culture of South America. Second Beno Rothberg Memorial Lecture. Institute of Archaeo-Metallurgy, University College of London.
- Shimada I, Cervantes G, Elera C, Kato K, Matsumoto G, Mondragón E, and Watanabe H. 2007a.
   Organization and Variability among Middle Sicán Elite Burials in Peru. 72nd Annual
   Meeting of the Society for American Archaeology. Austin, TX.
- Shimada I, Cervantes G, Elera C, Segura R, and Matsumoto G. 2007b. Rethinking Funerary Analysis in Andean Archaeology: Perspective from Sicán and Pachacamac. 35th Midwest Conference on ANdean and Amazonian Archaeology and Ethnohistory. Carbondale, IL.
- Shimada I, and Craig A. 2013. The style, technology, and organization of Sicán mining and metallurgy, Northern Peru: Insights from a holistic study. Antropología Chilena 45(1):3-31.
- Shimada I, and Griffin J. 2005. Precious metal objects of the Middle Sicán. Scientific American 15(1s):80-89.
- Shimada I, Griffin J, and Gordus A. 2000. The technology, iconography, and social significance of metals: a multi-dimensional analysis of Middle Sicán objects. In: McEwan C, editor.
  Precolumbian gold: Technology, style, and iconography. Chicago, IL: Fitzroy Dearborn Publishers. p 28-61.

- Shimada I, Klaus HD, Segura R, and Matsumoto G. 2015. Living with the dead: Conception and treatment of the dead on the Peruvian coast. In: Shimada I, and Fitzsimmons JL, editors.Living with the dead in the Andes. Tuscon, AZ: University of Arizona Press. p 101-172.
- Shimada I, and Samillán C. 2014. Arte, religión, y comología de Sicán Medio: Nuevos enfoques.
  In: Shimada I, editor. Cultural Sicán: Esplendor preincaico de la costa norte. Lima, PE:
  Editorial del Congreso del Perú. p 169-194.
- Shimada I, Segura R, Goldstein D, Shimada M, Speakman R, Wagner U, and Watanabe H. 2006a. What did people do at Pachacamac?: Identity, form, timing, and meaning of offerings. 34th Midwest Conference on Andean and Amazonian Archaeology and Ethnohistory. Vanderbilt University, Nashville.
- Shimada I, Segura R, Goldstein DJ, Knudson KJ, Shimada MJ, Shinoda K-i, Takigami M, and
  Wagner U. 2010. Un Siglo Después De Uhle: Reflexiones Sobre La Arqueología de
  Pachacamac y Perú. In: Kaulicke P, Fischer M, Masson P, and Wolff G, editors. Max
  Uhle: Evaluaciones De Sus Investigaciones Y Obras. Lima, Peru: Fondo Editorial de la
  Pontificia Universidad Católica del Perú. p 109-150.
- Shimada I, Segura R, and Matsumoto G. 2006b. Living with the dead: Conception and treatment of the dead at Pachacamac. 34th Midwest COnference on Andean and Amazonian Archaeology and Ethnohistory. Vanderbilt University Nashville, TN.
- Shimada I, Shinoda Ki, Farnum J, Corruccini R, and Watanabe H. 2004. An Integrated Analysis of Pre-Hispanic Mortuary Practices: A Middle Sicán Case Study. Current Anthropology 45(3):369-402.

- Shimada I, and Wagner U. 2007. A holistic approach to Pre-Hispanic craft production. In: Skibo J, Graves, MW., Stark, MT., editor. Archaeological anthropology: perspectives on method and theory. Tuscon, AZ: University of Arizona Press. p 163-197.
- Shinoda K-i. 2014. Análisis de ADN de la sociedad sicán. In: Shimada I, editor. Cultura Sicán: Esplendor preincaico de la costa norte. Lima, PE: Fondo Editorial del Congreso del Perú. p 293-300.
- Sofaer JR. 2006. The Body as Material Culture: A Theoretical Osteoarchaeology. Oxford, UK: Oxford University Press. 188 p.
- Spector T, and MacGregor A. 2004. Risk factors for osteoarthritis: genetics. Osteoarthritis and Cartilage 12:s39-s44.
- Steen SL, and Lane RW. 1998. Evaluation of habitual activities among two Alaskan Eskimo populations based on musculoskeletal stress markers. International Journal of Osteoarchaeology 8(5):341-353.
- Stiger T, Kosinski A, and Barnhart H. 1998. ANOVA for repeated ordinal sata with small sample size? A comparison of ANOVA, MANOVA, WLS, and GEE methods by simulation. Communications in Statistics 27:357-375.
- Sulsky SI, Carlton L, Bochman F, Ellegast R, Glitsch U, hartmann B, Pallapies D, Seidel D, and Sun Y. 2012. Epidemiological evidence for workload as a risk factor for osteoarthritis of the hip: A systematic review. PLoSONE 7(2):e31521.
- Swartz EE, Floyd RT, and Cendoma M. 2005. Cervical spine functional anatomy and the biomechanics of injury due to compressive loading. Journal of Athletic Training 40(3):155-162.

- Tainter JA. 1978. Mortuary Practices and the Study of Prehistoric Social Systems. Advances in Archaeological Method and Theory 1:105-141.
- Takigami M, Shimada I, Segura R, Muno S, Matsuzaki H, Tokanai F, Kato K, Mukai H, Takayuki O, and Yoneda M. 2014. Assessing the chronology and rewrapping of funerary bundles at the prehispanic religious center of Pachacamac, Peru. Latin American Antiquity 25(3):322-343.
- Tarlow S. 1999. Bereavement and Commemoration: An Archaeology of Mortality. Oxford, UK: Blackwell Publishers.
- Tilley L. 2015. Theory and Practice in the Bioarchaeology of Care. New York, NY: Springer.
- Toyne JM. 2002. Tales Woven in Their Bones: Osteological examiniation of the human skeletal remains from the stone temple at Túcume, Peru. London, ON: University of Western Ontario.
- Toyne JM, White CD, Verano JW, Uceda SC, Millaire JF, and Longstaffe FJ. 2014. Residential histories of elites and sacrificial victims at Huacas de Moche, Peru, as reconstructed from oxygen isotopes. Journal of Archaeological Science 42: 15-28.
- Trigger BG. 1991. Post-processual developments in Anglo-American archaeology. Norwegian Archaeological Review 24(2):65-76.
- Uceda S, and Rengifo CE. 2006. La especialización del trabajo: El caso de los orfebres Mochicas. Bull Inst fr études andines 35(2):149-185.
- Ucko PJ. 1969. Ethnography and interpretation of funerary remains. World Archaeology 1(2):262-280.
- Uhle M, and Shimada I. 1991. Pachacamac: A Reprint of the 1903 Edition by Max Uhle. Philadelphia, PA: University of Pensylvania. 103 p.

- Vallejo F. 2004. El estilo Ychsma: Características generales, secuencia y distribución geográfica. Bull Inst fr études andines 33:595-642.
- Vaughn KJ. 2006. Craft Production, Exchange, and Political Power in the Pre-Incaic Andes. Joournal of Archaeological Research 14:313-344.
- Verano J. 2007. Biodistance analysis of the Moche sacrificial victims from Huaca de la Luna plaza 3C: Matrix method test of their origins. American Journal of Physical Anthropology 132(2): 193-206.
- Villotte S. 2006. Connaissances médicales actuelles, cotation des enthésopathies: Nouvelle méthode. Bull Mém Soc Anthropol Paris 18:65-85.
- Villotte S, Castex D, Couallier V, Dutour O, Knüsel CJ, and Henry-Gambier D. 2010.
   Enthesopathies as occupational stress markers: Evidence from the upper limb. American Journal Of Physical Anthropology 142(2):224-234.
- Villotte S, and Knüsel CJ. 2012. Comment on "Musculoskeletal stress marker (MSM) differences in the modern American upper limb and pectoral girdle in relation to activity level and body mass index (BMI)" by K. Godde and R. Wilson Taylor. Forensic Science International 217(1):e31.
- Villotte S, and Knüsel CJ. 2013. Understanding Entheseal Changes: Definition and Life Course Changes. International Journal of Osteoarchaeology 23:135-146.
- Walker PL. 2001. A Bioarchaeological Perspective on the History of Violence. Annual Review of Anthropology 30:573-596.
- Weiss E. 2007. Muscle markers revisited: Activity pattern reconstruction with controls in a central California Amerind population. American Journal of Physical Anthropology 133(3):931-940.

- Weiss E. 2015. Examining Activity Patterns and Biological Confounding Factors: Differences between Fibrocartilaginous and Fibrous Musculoskeletal Stress Markers. International Journal of Osteoarchaeology 25(3):281-288.
- Weiss E, and Jurmain R. 2007. Osteoarthritis revisited: a contemporary review of aetiology. International Journal of Osteoarchaeology 17(5):437-450.
- Wesp JK. 2014. Bodies of Work: Organization of Everyday Life Activities in Urban New Spain [dissertation]. Berkeley, CA: University of California, Berkeley.
- White TD, and Folkens PA. 2000. Human Osteology, Second Edition. San Diego, CA: Academic Press, Inc. 563 p.
- Wilczak CA. 1998. Consideration of sexual dimorphism, age, and asymmetry in quantitative measurements of muscle insertion sites. International Journal of Osteoarchaeology 8(5):311-325.
- Wood JW, Milner GR, Harpending HC, Weiss KM, Cohen MN, Eisenberg LE, Hutchinson DL, Jankauskas R, ÄŒesnys G, Katzenberg MA et al. 1992. The Osteological Paradox:
  Problems of Inferring Prehistoric Health from Skeletal Samples [and Comments and Reply]. Current Anthropology 33(4):343-370.
- Wright LE, and Yoder CJ. 2003. Recent Progress in Bioarchaeology: Approaches to the Osteological Paradox. Journal of Archaeological Research 11(1):43-70.
- Zumwalt A. 2006. The effect of endurance exercise on the morphology of muscle attachment sites. J Exp Biol 209(3):444-454.

APPENDICES

## APPENDIX A

# Non-Elite Skeletal Sample Inventory

		Year		
Burial ID	Site	excavated	Storage ID	Location
				Coordinates: S 52.70 – 53.30; E 93.40
El Brujo 2	Cao Viejo, El Brujo	1992	Box 8	- 94.30
El Brujo 4	Cao Viejo, EL Brujo	1992	Box 46	Coordinates: S 59; E 85
El Brujo 5	Cao Viejo, EL Brujo	1992	Box 43	Coordinates: S 58; E 84
El Brujo 6	Cao Viejo, El Brujo	1992	Box 113	Coordinates: S 58 – 59; E 85 – 86
El Brujo 7	Cao Viejo, El Brujo	1992	Box 4	Coordinates: S 53.54; E 94.74
El Brujo 8	Cao Viejo, El Brujo	1992	Box 54	Coordinates: S 54.98; 90.82
El Brujo 9	Cao Viejo, El Brujo	1992	Box 5	Coordinates: S 53.70; E 92.15
El Brujo 12	Cao Viejo, El Brujo	1992	Box 21	Coordinates: S 62.10; E 87.60
El Brujo 14	Cao Viejo, El Brujo	1992	Box 131	Coordinates: Y 53.10 R 88.10
El Brujo 15	Cao Viejo, El Brujo	1992	Box 14	Coordinates: S 58.59; E 94.74x
El Brujo 20	Cao Viejo, El Brujo	1992	Box 4	Coordinates: S 62.25; E 89.30

		Year		
Burial ID	Site	excavated	Storage ID	Location
El Brujo 28	Cao Viejo, El Brujo	1992	Box 128	Coordinates: S 50.48; E 89 – 86
El Brujo 29	Cao Viejo, El Brujo	1992	Box 135	Coordinates: Y 52.20; R 89.35
El Bruio 31	Cao Viejo El Brujo	1002	Box 130	Coordinates: V 53 10: P 80 40
El Diujo 51		1772	D0X 150	Coordinates: 1 33.10, K 87.40
El Brujo 32	Cao Viejo, El Brujo	1992	Box 35	Coordinates: S 60 – 95; E 85.90
	5 / 5			·
El Brujo 33	Cao Viejo, El Brujo	1992	Box 46	Coordinates S 64.85; E 79.90
		1000	<b>D</b> (0)	
El Brujo 34	Cao Viejo, El Brujo	1992	Box 60	Coordinates: S 62; E 89
El Bruio 36	Cao Vieio El Bruio	1992	Box 11	Coordinates: \$ 51 30: E 91 70
LI DIUJO 50		1772	DOX 11	Coordinates: 5 51.50; E 71.70
El Brujo 38	Cao Viejo, El Brujo	1992	Box 52	Coordinates: S 60.75; E 86.40
El Brujo 41	Cao Viejo, El Brujo	1992	Box 3	Coordinates: S 50.92; E 93.20
El Davia 42	Cas Visia El Druis	1001	Day (2	Coordinates: 5 (1.95; E.99.05
El Brujo 42	Cao viejo, El Brujo	1991	B0X 62	Coordinates: 5 61.85; E 88.05
El Brujo 44	Cao Vieio. El Bruio	1992	Box 51	Coordinates: S 61.62: E 89.20
				·····, _ ·····, _ ·····
				Coordinates: S 50 – 55; E 105 –
El Brujo 51	Cao Viejo, El Brujo	1990	Box 3	110 98
٨F	Painted Temple,	2006	Box 175	Tomb 1.2 upper lavel
AI'	Pachacamac	2000	DUX 1/J	10110 1-2, upper level
	Painted Temple,			
BA	Pachacamac	2006	Box 176	Tomb 1-2, upper level

		Year				
Burial ID	Site	excavated	Storage ID	Location / Feature		
Б	Painted Temple,	2006	Day 160	Tamb 1.2 Januar Javal		
E	Pachacamac	2006	B0X 169	1 omb 1-2, lower level		
~	Painted Temple,					
G	Pachacamac	2006	Box 174	Tomb 1-2, lower level		
	Painted Temple,					
0	Pachacamac	2006	Box 172	Tomb 1-2, lower level		
_	Painted Temple,	• • • • •				
R	Pachacamac	2006	Box 170	Tomb 1-2, lower level		
	Painted Temple,					
Т	Pachacmac	2006	Box 179	Tomb 3, intrusion		
	Painted Temple,					
U	Pachacamac	2006	Box 171	Tomb 1-2, lower level		
	Painted Temple,					
V	Pachacamac	2006	Box 164	Tomb 1-2, lower level		
	Painted Temple,					
Z	Pachacamac	2006	Box 166	Tomb 1-2, lower level		
	Huaca Loro, Sicán					
HL NP Sac	Precinct	2008	Box 19	North platform; H-160		
	Huaca Rodillona, Sicán					
HR C10 1	Precinct	_	_	Column 10		

		Year					
Burial ID	Site	excavated	Storage ID	Location / Feature			
HR C12 2	Huaca Rodillona, Sicán Precinct	_	_	Column 12			
HS C 8	Huaca Sialupe, Sicán Precinct	2001	_	Central sector			
	Huaca Sialupe, Sicán	2001		Execution area 5 and automaion			
пзе з	Precinct	2001	_	Excavation area 5, east extension			
	Huaca Sialupe, Sicán	1000		M 11			
HS MI I	Precinct	1999	_	Mound 1			
	Huaca Sialupe, Sicán	2001					
HS M1 C6 I	Precinct	2001	—	Mound 1, cemetery excavation 6			

## APPENDIX B

# Elite Skeletal Sample Inventory

	0	Year		
Burial ID	Site	excavated	Storage ID	Location / Feature
El Brujo 13	Cao Viejo, El Brujo	1992	Box 112	Coordinates: S 58.59; E 85.86
El Brujo 39	Cao Viejo, El Brujo	1995	Box 45	Coordinates: S 68.56; E 77.98x
BB	Painted Temple, Pachacamac	2006	_	Tomb 1-2, lower level
Н	Painted Temple, Pachacamac	2006	Box 168	Tomb 1-2,
N	Painted Temple, Pachacamac	2006	Box 165	Tomb 1-2
Q	Painted Temple, Pachacamac	2006	Box 173	Tomb 1-2
W	Painted Temple, Pachacamac	2006	Item 14446	Tomb 1-2, lower level
	Huaca Las Ventanas,			
HLV SS 5	Sicán Precinct	_	_	South sector, excavation area 1, offering 3
	Huaca Las Ventanas,			
HLV SS 7	Sicán Precinct	_	_	South sector, excavation area 1, offering 5
	Huaca Las Ventanas,			
HLV SS 8	Sicán Precinct	_	_	South sector, excavation area 1, offering 3

		Year		
Burial ID	Site	excavated	Storage ID	Location / Feature`
HL 06 T1 1	Huaca Loro, Sicán Precinct	2006	_	Trench 1, Tomb 1
HL 06 T1 2	Huaca Loro, Sicán Precinct	2006	_	Trench 1, Tomb 2
HL 06 Tr1 T4	Huaca Loro, Sicán Precinct	2006	_	Trench 1
HL E 5	Huaca Loro, Sicán Precinct	1991	_	East Tomb, Trench 1
HL WT 2	Huaca Loro, Sicán Precinct	1995	_	West Tomb
HL WT 3	Huaca Loro, Sicán Precinct	1995	_	West Tomb
HL WT 7	Huaca Loro, Sican Precinct	1995	_	West Tomb, Pit 3
HL WT 9	Huaca Loro, Sican Precinct	1995	_	West Tomb, Pit 4
HL WT 10	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 4
HL WT 11	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 5
HL WT 12	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 5
HL WT 14	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 7
HL WT 17	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 10
HL WT 18	Huaca Loro, Sicán Precint	1995	_	West Tomb, Pit 11
				West Tomb, Central chamber, Excavation
HL WT 20	Huaca Loro, Sicán Precinct	1995	_	area 1, Niche 11

		Year		
Burial ID	Site	excavated	Storage ID	Location / Feature`
HL WT 21	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Pit 9
HL WT 24	Huaca Loro, Sicán Precinct	1995	_	West Tomb, Niche 12

#### APPENDIX C

### Non-Elite Sample Fibrous Entheses Data

		Av			$LD^1$	LD	TM <sup>2</sup>	TM	PM <sup>3</sup>	PM	$D^4$	D	AG
Site	Burial ID	Age	Sex	Size	(R)	(L)	(R)	(L)	(R)	(L)	(R)	(L)	$EC^5$
Sicán Precinct	HS C 8	17.5	F	-2.971	0	0	1	3	0	0	0	3	7
Sicán Precinct	HR C10 1	22	М	-0.365	0	0	0	0	0	0	0	0	0
Sicán Precinct	HS M1 1	22.5	М	-0.404	0	0	0	0	0	0	0	0	0
Sicán Precinct	HR C12 2	23.5	F	-0.249	0	0	0	0	0	0	0	0	0
Sicán Precinct	HS E 5	40	М	-0.365	0	0	0	0	0	0	0	0	0
Sicán Precinct	HS M1 C6 1	45	М	3.042	0	0	0	0	3	3	3	3	12
Sicán Precinct	HL NP Sac	45	М	0.534	0	0	1	1	3	3	0	2	10

<sup>1</sup> LD = Lattisiums dorsi insertion
 <sup>2</sup> TM = Teres major insertion
 <sup>3</sup> PM = Pectoralis major insertion

 $^{4}$  D = Deltoideus insertion

<sup>5</sup> AG EC = Aggregate fibrous EC score

Site	Burial ID	Av Age	Sex	Size	LD <sup>1</sup> (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D <sup>4</sup> (R)	D (L)	AG EC <sup>5</sup>
El Brujo	El Brujo 9	20.5	М	3.631	1	1	1	3	3	3	2	3	17
El Brujo	El Brujo 32	21	F	-4.038	0	0	1	1	2	1	2	2	9
El Brujo	EL Brujo 12	22	F	-0.338	1	1	1	1	3	3	1	2	13
El Brujo	El Brujo 15	23	М	3.186	1	1	2	1	3	3	2	2	15
El Brujo	El Brujo 44	25	М	0.771	1	1	1	2	2	2	2	2	13
El Brujo	EL Brujo 36	27.5	М	1.26	1	1	1	2	3	3	2	3	16
El Brujo	El Brujo 38	27.5	М	0.92	1	1	2	1	2	3	2	3	15
El Brujo	El Brujo 6	30	F	-4.645	1	2	1	2	2	1	2	2	13
El Brujo	EL Brujo 31	30	F	-2.539	1	1	1	1	2	2	2	2	12

<sup>1</sup> LD = Lattisiums dorsi insertion
 <sup>2</sup> TM = Teres major insertion
 <sup>3</sup> PM = Pectoralis major insertion

 $^4$  D = Deltoideus insertion

<sup>5</sup> AG EC = Aggregate fibrous EC score

Site	Burial ID	Av Age	Sex	Size	LD <sup>1</sup> (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D <sup>4</sup> (R)	D (L)	AG EC <sup>5</sup>
El Brujo	EL Brujo 14	30	М	2.458	1	1	2	1	3	2	2	2	14
El Brujo	El Brujo 20	30	М	4.428	1	1	2	2	3	3	2	2	16
El Brujo	El Brujo 66	32.5	М	3.806	1	1	1	1	3	3	2	3	15
El Brujo	El Brujo 33	35	F	-0.636	1	1	2	1	3	3	3	2	16
El Brujo	El Brujo 2	35	М	2.785	1	1	2	2	3	2	3	2	16
El Brujo	El Brujo 5	35	М	1.3	1	1	2	1	3	3	2	2	15
El Brujo	El Brujo 42	37.5	F	-1.391	1	1	3	2	3	3	3	3	19
El Brujo	El Brujo 34	38.5	F	-3.077	1	1	2	1	3	3	2	3	16
El Brujo	EL Brujo 41	40	F	-1.362	1	1	2	2	3	3	2	3	17
El Brujo	El Brujo 7	40	М	4.058	1	1	2	2	3	3	2	3	17

<sup>1</sup> LD = Lattisiums dorsi insertion

<sup>2</sup> TM = Teres major insertion <sup>3</sup> PM = Pectoralis major insertion

 $^4$  D = Deltoideus insertion

 $^{5}$  AG EC = Aggregate fibrous EC score

<u> </u>		Av	a	<i>a</i> .	$LD^1$	LD	$TM^2$	TM	$PM^3$	PM	$D^4$	D	AG
Site	Burial ID	Age	Sex	Size	(R)	(L)	(R)	(L)	(R)	(L)	(R)	(L)	EC <sup>3</sup>
El Brujo	EL Brujo 28	45	М	1.949	1	1	1	2	3	2	2	2	14
El Brujo	El Brujo 8	50	М	3.524	1	1	2	2	3	3	2	2	16
El Brujo	El Brujo 4	55	F	-2.245	0	2	0	1	3	3	3	3	15
Pachacamac	Т	19.5	F	-4.431	0	0	0	0	1	0	1	0	2
Pachacamac	AF	19.5	F	-2.748	0	0	0	1	1	1	1	0	4
Pachacamac	U	27.5	М	5.125	0	0	2	0	3	1	2	1	9
Pachacamac	G	29.5	М	4.333	0	0	2	1	3	3	3	3	15
Pachacamac	Е	30	М	-2.033	0	0	1	1.5	2	1.5	3	1.5	10.5
Pachacamac	BA	32.5	М	3.171	0	0	2	2	2	2	2	2	12
Pachacamac	Х	34.5	0	-0.209	0	0	0	0	1	1	2	2	6
Pachacamac	0	42	1	5.143	0	0	0	0	3	2	2	1	8

- <sup>1</sup> LD = Lattisiums dorsi insertion

- <sup>2</sup> TM = Teres major insertion
  <sup>3</sup> PM = Pectoralis major insertion
  <sup>4</sup> D = Deltoideus insertion
  <sup>5</sup> AG EC = Aggregate fibrous EC score

		Av			$LD^1$	LD	TM <sup>2</sup>	ТМ	PM <sup>3</sup>	PM	$D^4$	D	AG
Site	Burial ID	Age	Sex	Size	(R)	(L)	(R)	(L)	(R)	(L)	(R)	(L)	$EC^5$
Pachacamac	V	44.5	1	-0.615	0	0	1	1	3	2	2	1	10
Pachacamac	Ζ	44.5	1	4.454	0	0	2	2	2	2	2	1	11
Pachacamac	R	47	0	-2.165	0	0	1	0	2	2	1	1	7

<sup>&</sup>lt;sup>1</sup> LD = Lattisiums dorsi insertion

<sup>&</sup>lt;sup>2</sup> TM = Teres major insertion
<sup>3</sup> PM = Pectoralis major insertion
<sup>4</sup> D = Deltoideus insertion
<sup>5</sup> AG EC = Aggregate fibrous EC score

#### APPENDIX D

### Non-Elite Fibrocartilaginous Entheses Data

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Sicán Precinct	HS C 8	17.5	F	-2.971	0	0	0	0	0	3	3
Sicán Precinct	HR C10 1	22	М	-0.365	0	0	0	0	0	0	0
Sicán Precinct	HS M1 1	22.5	М	-0.404	3	3	0	0	0	0	6
Sicán Precinct	HR C12 2	23.5	F	-0.249	0	0	0	0	0	0	0
Sicán Precinct	HSE5	40	М	-0.365	0	0	0	0	0	0	0
Sicán Precinct	HS M1 C6 1	45	М	3.042	0	0	0	0	0	0	0

- <sup>1</sup> BB = Biceps brachii insertion
  <sup>2</sup> TB = Triceps brachii insertion
  <sup>3</sup> B = Brachialis insertion
  <sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Sicán Precinct	HL NP Sac	45	М	0.534	0	2	1	1	1	2	7
El Brujo	El Brujo 9	20.5	М	3.631	1	1	1	1	1	1	6
El Brujo	El Brujo 32	21	F	-4.038	1	1	0	0	2	2	6
El Brujo	El Brujo 12	22	F	-0.338	1	1	1	1	3	3	10
El Brujo	El Brujo 15	23	М	3.186	1	1	1	1	2	2	8
El Brujo	El Brujo 44	25	М	0.771	2	2	1	1	2	2	10
El Brujo	El Brujo 36	27.5	М	1.26	3	3	1	1	3	3	14
El Brujo	El Brujo 38	27.5	М	0.92	1	1	1	1	1	2	7
El Brujo	El Brujo 31	30	F	-2.539	1	1	1	1	1	2	7
El Brujo	El Brujo 6	30	F	-4.645	1	1	1	1	2	2	8

<sup>1</sup> BB = Biceps brachii insertion
<sup>2</sup> TB = Triceps brachii insertion
<sup>3</sup> B = Brachialis insertion
<sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
El Brujo	El Brujo 14	30	М	2.458	3	3	1	1	3	3	14
El Brujo	El Brujo 20	30	М	4.428	1	2	1	1	2	2	9
El Brujo	El Brujo 66	32.5	М	3.806	1	1	1	1	3	3	10
El Brujo	El Brujo 33	35	F	-0.636	1	1	1	2	3	3	11
El Brujo	El Brujo 2	35	М	2.785	3	3	1	1	2	1	11
El Brujo	El Brujo 5	35	М	1.3	1	1	1	1	3	3	10
El Brujo	El Brujo 42	37.5	F	-1.391	3	2	1	2	3	3	14
El Brujo	El Brujo 34	38.5	F	-3.077	1	1	1	1	2	2	8
El Brujo	EL Brujo 41	40	F	-1.362	3	3	1	1	2	3	13
El Brujo	El Brujo 7	40	М	4.058	1	2	1	1	1	2	8

<sup>1</sup> BB = Biceps brachii insertion
 <sup>2</sup> TB = Triceps brachii insertion
 <sup>3</sup> B = Brachialis insertion
 <sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
El Brujo	EL Brujo 28	45	М	1.949	1	1	1	1	2	2	8
El Brujo	El Brujo 8	50	М	3.524	2	2	1	1	2	2	10
El Brujo	El Brujo 4	55	F	-2.245	1	1	1	1	2	3	9
Pachacamac	Т	19.5	F	-4.431	0	0	0	0	0	0	0
Pachacamac	AF	19.5	F	-2.748	0	0	0	0	3	0	3
Pachacamac	U	27.5	М	5.125	1	0	2	2	1	1	7
Pachacamac	G	29.5	М	4.333	1	0	1	1	1	1	5
Pachacamac	Е	30	М	-2.033	1.5	0	1.5	0	1	1	5
Pachacamac	BA	32.5	М	3.171	0	0	0	2	0	2	4
Pachacamac	Х	34.5	F	-0.209	1	2	0	0	2	1	6

<sup>1</sup> BB = Biceps brachii insertion
 <sup>2</sup> TB = Triceps brachii insertion
 <sup>3</sup> B = Brachialis insertion
 <sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Pachacamac	0	42	М	5.143	1	2	2	0	1	1	7
Pachacamac	V	44.5	М	-0.615	2	1	1	1	2	2	9
Pachacamac	Z	44.5	М	4.454	2	2	0	0	3	3	10
Pachacamac	R	47	F	-2.165	0	2	2	2	2	2	10

- <sup>1</sup> BB = Biceps brachii insertion
  <sup>2</sup> TB = Triceps brachii insertion
  <sup>3</sup> B = Brachialis insertion
  <sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score
#### APPENDIX E

Elite Sample Fibrous Entheses Data

Site	Burial ID	Av Age	Sex	Size	$LD^1$ (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D <sup>4</sup> (R)	D (L)	AG EC <sup>5</sup>
Sicán Precinct	HL WT 18	18.5	F	0.282	1	1	1	2	1	1	2	2	11
Sicán Precinct	HL WT 12	19	F	-4.391	1	1	1	1	2	3	1	0	10
Sicán Precinct	HL WT 9	20	F	-3.583	0	0	0	2	0	2	2	2	8
Sicán Precinct	HLV SS 8	20	F	-1.997	0	0	0	0	0	2	3	3	8
Sicán Precinct	HL WT 11	20	F	-3.092	1	1	1	0	3	3	2	0	11

<sup>1</sup> LD = Lattisiums dorsi insertion
 <sup>2</sup> TM = Teres major insertion
 <sup>3</sup> PM = Pectoralis major insertion

 $^{4}$  D = Deltoideus insertion

<sup>5</sup> AG EC = Aggregate fibrous EC score

Site	Burial ID	Av Age	Sex	Size	LD <sup>1</sup> (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D <sup>4</sup> (R)	D (L)	AG EC <sup>5</sup>
Sicán Precinct	HL WT 14	20.5	F	-1.505	1	1	1	1	2	2	2	2	12
Sicán Precinct	HL 06 T1 2	22	F	-0.603	0	0	0	0	0	0	0	0	0
Sicán Precinct	HL WT 3	22	F	-4.586	0	0	1	0	2	2	0	0	5
Sicán Precinct	HL WT 24	22.5	F	-5.086	0	0	1	1	3	3	2	3	13
Sicán Precinct	HL WT 7	22.5	F	-1.329	1	0	2	2	2	2	2	2	13
Sicán Precinct	HL ET 5	22.5	F	-1.337	0	1	0	2	0	3	0	3	9
Sicán Precinct	HL WT 17	27	F	-1.721	1	1	1	2	2	2	1	1	11
Sicán Precinct	HLV SS 5	23.5	F	-0.603	1	1	1	1	2	2	3	2	13
Sicán Precinct	HL WT 10	23.5	F	-0.603	0	0	1	0	3	0	2	0	6

- $^{1}$  LD = Lattisimus dorsi insertion
- <sup>2</sup> TM = Teres major insertion <sup>3</sup> PM = Pectoralis major insertion <sup>4</sup> D = Deltoideus insertion
- <sup>5</sup> AG EC = Aggregate fibrous EC score

Site	Burial ID	Av Age	Sex	Size	LD <sup>1</sup> (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D <sup>4</sup> (R)	D (L)	AG EC <sup>5</sup>
Sicán Precinct	HL WT 21	30	F	-3.558	0	0	1	1	3	1	2	2	10
Sicán Precinct	HL WT 20	30	F	-1.51	0	0	2	2	2	2	1	2	11
Sicán Precinct	HL 06 Tr1 T4	30	М	-0.603	0	0	0	0	0	0	0	1	1
Sicán Precinct	HL 06 T1 1	40	F	-0.603	0	0	0	0	0	0	0	0	0
Sicán Precinct	HL WT 2	42.5	М	2.743	1	1	2	1	3	3	2	2	15
El Brujo	El Brujo 13	30	М	2.448	1	1	1	1	2	1	2	2	11
El Brujo	El Brujo 39	35	F	-0.676	1	1	1	1	2	2	2	2	12
Pachacamac	Н	27	М	2.52	0	0	1	1	2	2	2	2	10
Pachacamac	Q	40	F	-0.29	0	0	0	0	3	1	3	3	10

<sup>1</sup> LD = Lattissimus dorsi insertion

<sup>2</sup> TM = Teres major insertion
<sup>3</sup> PM = Pectoralis major insertion
<sup>4</sup> D = Deltoideus insertion
<sup>5</sup> AG EC = Aggregate fibrous EC score

Site	Burial ID	Av Age	Sex	Size	LD <sup>1</sup> (R)	LD (L)	TM <sup>2</sup> (R)	TM (L)	PM <sup>3</sup> (R)	PM (L)	D 4(R)	D (L)	AG EC <sup>5</sup>
Pachacamac	BB	47	М	3.662	0	0	3	3	3	3	2	2	16
Pachacamac	W	57	F	0.694	0	0	0	0	2	2	2	2	8

- <sup>2</sup> TM = Teres major insertion
  <sup>3</sup> PM = Pectoralis major insertion
  <sup>4</sup> D = Deltoideus insertion
  <sup>5</sup> AG EC = Aggregate fibrous EC score

<sup>&</sup>lt;sup>1</sup> LD = Lattissimus dorsi insertion

### APPENDIX F

# Elite Sample Fibrocartilaginous Entheses Data

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Sicán Precinct	HL WT 18	18.5	F	0.282	0	0	0	0	0	0	0
Sicán Precinct	HL WT 12	19	F	-4.391	2	1	0	1	1	2	7
Sicán Precinct	HL WT 9	20	F	-3.583	1	2	0	0	2	2	7
Sicán Precinct	HLV SS 8	20	F	-1.997	2	0	0	0	1	0	3
Sicán Precinct	HL WT 11	20	F	-3.092	1	1	1	2	2	3	10
Sicán Precinct	HL WT 14	20.5	F	-1.505	1	2	0	1	2	2	8

<sup>1</sup> BB = Biceps brachii insertion
<sup>2</sup> TB = Triceps brachii insertion
<sup>3</sup> B = Brachialis insertion
<sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Sicán Precinct	HL 06 T1 2	22	F	-0.603	0	0	0	0	0	0	0
Sicán Precinct	HL WT 3	22	F	-4.586	0	0	0	0	1	1	2
Sicán Precinct	HL WT 24	22.5	F	-5.086	2	1	1	1	3	2	10
Sicán Precinct	HL WT 7	22.5	F	-1.329	2	2	1	1	3	2	11
Sicán Precinct	HL ET 5	22.5	F	-1.337	1	2	1	1	2	2	9
Sicán Precinct	HL WT 10	23.5	F	-0.603	1	0	0	0	2	0	3
Sicán Precinct	HLV SS 5	23.5	F	-0.603	1	0	1	1	2	2	7
Sicán Precinct	HL WT 17	27	F	-1.721	1	1	1	1	2	3	9
Sicán Precinct	HLV SS 7	25	М	-1.634	0	3	0	0	0	0	3
Sicán Precinct	HL WT 21	30	F	-3.558	1	1	1	1	2	1	7

<sup>1</sup> BB = Biceps brachii insertion
<sup>2</sup> TB = Triceps brachii insertion
<sup>3</sup> B = Brachialis insertion
<sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Sicán Precinct	HL WT 20	30	F	-1.51	1	1	1	1	2	1	7
Sicán Precinct	HL 06 Tr1 T4	30	М	-0.603	0	0	0	0	0	0	0
Sicán Precinct	HL 06 T1 1	40	F	-0.603	0	0	0	0	0	0	0
Sicán Precinct	HL WT 2	42.5	М	2.743	0	0	1	1	1	1	4
El Brujo	El Brujo 13	30	М	2.448	3	2	1	2	2	2	12
El Brujo	El Brujo 39	35	F	-0.676	1	3	0	0	1	2	7
Pachacamac	Н	27	М	2.52	1	1	0	0	1	1	4
Pachacamac	Ν	32.5	М	3.914	2	3	0	0	2	2	9
Pachacamac	Q	40	F	-0.29	0	0	2	2	2	2	8
Pachacamac	W	57	F	0.694	0	0	0	0	0	2	2

<sup>1</sup> BB = Biceps brachii insertion
<sup>2</sup> TB = Triceps brachii insertion
<sup>3</sup> B = Brachialis insertion
<sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

Site	Burial ID	Av Age	Sex	Size	BB <sup>1</sup> (R)	BB (L)	TB <sup>2</sup> (R)	TB (L)	B <sup>3</sup> (R)	B (L)	AG EC <sup>4</sup>
Pachacamac	BB	47	М	3.662	2	2	0	0	0	0	4

<sup>&</sup>lt;sup>1</sup> BB = Biceps brachii insertion
<sup>2</sup> TB = Triceps brachii insertion
<sup>3</sup> B = Brachialis insertion
<sup>4</sup> AG EC = Aggregate fibrocartilaginous EC score

## APPENDIX G

# Permission Letter for Use of Copyrighted Photograph

Dear Sarah Muno,

Your query below has been forwarded to me. Cambridge University Press manages rights and permissions for LAQ, on behalf of the SAA which holds the copyright.

Cambridge University Press grants permission freely for the reproduction in another work of a short prose extract (less than 400 words), a single figure or a single table in which it holds rights (see the important caveat in the Notes below). In such cases a request for permission need not be submitted, **but the reproduced material must be accompanied by a full citation of the original source.** 

Keep in mind that Cambridge University Press may not be able to grant permission for the use of all the material it publishes. In particular, illustrations, tables, poetry or other extensive quotations included in Cambridge publications, may have been reproduced by Cambridge with the permission of another rights holder. Always check the caption or citation to ensure that Cambridge is the rights holder before proceeding.

This information can be found at: http://www.cambridge.org/about-us/rights-permissions/permissions-requests/

Requests for permissions beyond that above (which aren't needed in this case) can always be found at the article's homepage. For example, if you go to the article online, link below, you'll see just below the abstracts a "Request permission" button, which will lead to a series of questions.

https://www.cambridge.org/core/journals/latin-american-antiquity/article/assessing-the-chronologyand-rewrapping-of-funerary-bundles-at-the-prehispanic-religious-center-of-pachacamacperu/C73036F5B720BECC8B24AF15C7A65B99

Much good luck with your work.

Best, Mark Zadrozny Executive Publisher Cambridge University Press One Liberty Plaza New York NY 10006

#### VITA

#### Graduate School

## Southern Illinois University

Sarah K. Muno

smuno@pasadena.edu

San Francisco State University

Bachelor of Arts, Anthropology, April 2003

Western Michigan University

Master of Arts in Anthropology, December 2006

Dissertation Title:

Labor and Social Identity in Ancient Peru: A Bioarchaeological Perspective

Major Professor: Dr. Izumi Shimada

#### **Publications:**

Klaus, H.D., Shimada, I., Shinoda, K., & Muno, S. 2017 Middle Sicán mortuary archaeology, skeletal biology, and genetic structures in Late Pre-Hispanic South America. In H.D. Klaus, A.R. Harvey, & M.D. Cohen (Eds.), *Bones of complexity: Bioarchaeological case studies of social organization and skeletal biology* (pp. 408-449). Gainsville, FL: University Press of Florida.

Takigami, M., Shimada, I., Segura, R., Muno, S., Matsuzaki, H., Tokanai, F., Kato, K., Mukai, H., Takayuki, & O., Yoneda, M. 2014. Assessing the chronology and rewrapping of funerary bundles at the prehispanic religious center of Pachacamac, Peru. *Latin American Antiquity*, 25, 322-343.

Muno, S. 2014. La sociedad Sicán: Una vision desde los esqueletos. In I. Shimada (ed), *La cultura Sicán: Una vision global* (pp. 265-281). Lima, Peru: Fondo Editorial Congreso del Peru.

Muno, S. 2009 Life in the Bones: What skeletal remains reveal about Sicán society. In I.Shimada & M. Ono (Eds.), Precursors of the Inka empire: The golden capital of Sicán. (pp. 298-310). Tokyo, Japan: Tokyo Broadcasting System.